

University of Dundee

DOCTOR OF PHILOSOPHY

Towards the full inclusion of people with severe speech and physical impairments in the design of Augmentative and Alternative Communication software

Prior, Suzanne

Award date: 2011

Awarding institution: University of Dundee

Link to publication

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

• Users may download and print one copy of any publication from the public portal for the purpose of private study or research.

You may not further distribute the material or use it for any profit-making activity or commercial gain
You may freely distribute the URL identifying the publication in the public portal

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim. DOCTOR OF PHILOSOPHY

Towards the full inclusion of people with severe speech and physical impairments in the design of Augmentative and Alternative Communication software

Suzanne Prior

2011

University of Dundee

Conditions for Use and Duplication

Copyright of this work belongs to the author unless otherwise identified in the body of the thesis. It is permitted to use and duplicate this work only for personal and non-commercial research, study or criticism/review. You must obtain prior written consent from the author for any other use. Any quotation from this thesis must be acknowledged using the normal academic conventions. It is not permitted to supply the whole or part of this thesis to any other person or to post the same on any website or other online location without the prior written consent of the author. Contact the Discovery team (discovery@dundee.ac.uk) with any queries about the use or acknowledgement of this work.

Towards the Full Inclusion of People with Severe Speech and Physical Impairments in the Design of Augmentative and Alternative Communication Software

By

Suzanne Prior

Doctor of Philosophy

University of Dundee

Scotland

April 2011

Dedication

This work is dedicated to Alan for keeping me sane, making me laugh, being my best friend and doing the dishes every day for the final months of this work!

Acknowledgements

This research would not have been possible without the support and assistance of my friends and colleagues in the School of Computing and particularly the AAC research team for being willing to listen to my ideas and share their insights. Thank you to you all for getting me through the past three years.

I would like to acknowledge my funders, Capability Scotland for allowing me the opportunity to do this work, and to all of the Capability Scotland staff for their invaluable assistance.

Thank you to my family for all their support and encouragement.

I would like to extend my special thanks to the following people:

To Dr. Annalu Waller for her belief in this work and for her invaluable advice and support.

To Dr. Thilo Kroll for his careful editing in this work and for offering a fresh perspective on computing and social science.

And, most importantly my thanks to all my fantastic participants and in particular the "three troublemakers", without their effort and dedication this work would not have been possible. This is their story, I am proud to be able to tell it.

Declaration of the Candidate

I declare that I am the author of this thesis; that all references cited have been consulted by me; that the work of which this thesis is a record has been done by myself; and that this thesis has not been previously accepted for a higher degree.

Suzanne Prior

Declaration of the Supervisors

We declare that Suzanne Prior has satisfied all the terms and conditions of the regulations under Ordinances 12 and 39, and has completed the required terms of research to qualify in submitting this thesis in application for the degree of Doctor of Philosophy.

Dr. Annalu Waller

Dr. Thilo Kroll

Photographs

Permission from research participants has been given for the use of their photographs in this thesis.

Table of Contents

Abstract	t	1
Chapter	1. Introduction	3
Chapter	2. The Abandonment of Communication Aids	7
2.1	Introduction	7
2.2	Severe Speech and Physical Impairments	7
2.3	Augmentative and Alternative Communication	.10
2.4	Abandonment Rate	.13
2.5	Reasons for Abandonment	.14
2.6	AAC Abandonment, Human Computer Interaction and Usability	.14
2.7	Categorisation of AAC Abandonment Using Usability Components	.15
2.8	Abandonment and Usability	.20
2.9	Summary	.21
Chapter	3. User Centred Design	.23
3.1	Introduction	.23
3.2	User Centred design	.23
3.3	International Standards in User Centred Design	.24
3.4	Example of Problems when User Centred Design is not conducted	.28
3.5	Summary	.29
Chapter	4. Literature Review on User Centred Design and Adults with SSPI	.31
4.1	Introduction	.31

4.2	Current Amount of User Centred Design in Development of Assistive
Techr	nology
4.3	Rationale for a Systematic Review
4.4	User Centred Design Literature Review
4.5	Problems Cited in Conducting User Centred Design with Participants with
SSPI	43
4.6	User Centred Design and SSPI47
4.7	The Research Question
4.8	Summary
Chapter	5. Methods in Software Development with Adults with SSPI
5.1	Introduction
5.2	User Centred Design Methods
5.3	Choice of Methods
5.4	Summary
Chapter	6. Creation of CHAMPION Software with Adults with SSPI60
6.1	Introduction
6.2	Choice of Assistive Software
6.3	Planning the HCI Process for Adults with SSPI63
6.4	Identifying the Context of Use Through Observation67
6.5	Focus Groups with Adults with SSPI72
6.6	Review of Establishing Context of Use
6.7	Gathering Requirements Through Forum Theatre
6.8	Storyboards with Adults with SSPI

6.9	Review of Requirements Gathering Stage	94
6.10	Produce Design Solutions with Adults with SSPI	94
6.11	Low Fidelity Prototyping with Adults with SSPI	95
6.12	Mid Fidelity Prototyping with Adults with SSPI	99
6.13	High Fidelity Prototypes First Iteration with Adults with SSPI	104
6.14	High Fidelity Prototyping Second Iteration with Adults with SSPI.	107
6.15	Software Evaluation with Adults with SSPI	110
6.16	Evaluation of Participant's Experience of Process	114
6.17	Credibility of study	
6.18	Limitations to Study	121
6.19	Summary	122
Chapter	7. Development of Guidelines	
7.1	Introduction	
7.2	Development of Guidelines	
7.3	Preparation of Guidelines	124
7.4	Guidelines for Working With Adults with SSPI in User Centred Dev	velopment
	125	
7.5	Summary	139
Chapter	8. Evaluation of Guidelines and Second Study	140
8.1	Introduction	140
8.2	Second Study	140
8.3	Evolution of Guidelines	

8.4 Evaluation of Guidelines		
8.5 Conclusion of Second Study14		
8.6 Summary14		
Chapter 9. Discussion		
9.1 Introduction		
9.2 Summary of Research		
9.3 "Can the problems cited in the literature be overcome to allow User Centred		
Design to be conducted with participants with SSPI?"		
9.4 Can methods currently used in Human Computer Interaction be combined and		
adapted to allow people with SSPI to contribute to the design of technology?157		
9.5 Summary16		
Chapter 10. Implications and Conclusion		
10.1 Introduction		
10.2 Implications for Policy162		
10.3 Implications for Practice163		
10.4 Implications for Research		
10.5 Summary		
References		
Appendix A. Overview of User Centred Process,		
Appendix B. Rationale for CHAMPION		
Appendix C. Focus Groups as a Requirements Gathering Method with Adults with		
Severe Speech and Physical Impairments		
Appendix D. Forum Production and Script		

Appendix E.	Participatory Design of a Talking Photo Album	233
Appendix F.	Dissertation on Development of Talking PhotoAlbum	234
Appendix G.	Search Strings Used	328
Appendix H.	Description of Studies in Literature Review	330
Appendix I.	Requirements	331
Appendix J.	Evaluations	333
Appendix K.	Straight Talking User Centre	340
Appendix L.	Guidelines	345

Table of Figures

Figure 1 - Adult with SSPI Using Paper Word Board1	1
Figure 2 - Selection of AAC Devices and Access Methods1	2
Figure 3 – Comparison of ISO 13407 and ISO 92412	25
Figure 4 - Use of User Centred Design in ASSETS Proceedings Publications	52
Figure 5 - Ethical Procedure in CHAMPION Development	55
Figure 6 - Methods Used in Development Process	i8
Figure 7 - Forms of Observation	<u>i9</u>
Figure 8 - Participants Demonstrating Current Use of Computers	'1
Figure 9 - Layout of Focus Group7	'5
Figure 10 - Participants In Focus Group Discussing Bad Hospital Experience7	6
Figure 11 – Participants During Debriefing7	'7
Figure 12 - Responses of Participants (Prior et al., Accepted)7	'9
Figure 13 - Clip from First Play in Forum Theatre	\$4
Figure 14 - Clip from Second Play in Forum Theatre	\$4
Figure 15 - Scene with Requirements Written On	8
Figure 16 – Storyboard Featuring Scenes in Using System	12
Figure 17 - Magnetic Paper Prototyping9	16
Figure 18 - Participants Using Magnetic Prototypes9	16
Figure 19 - Peripherals10)1
Figure 20 - Participant Using Mid Fidelity Prototype10)2
Figure 21 - Poster For CHAMPION Evaluation11	0
Figure 22 - Graphic Review of Participant Debrief	6
Figure 23 - Feelings when Taking Part11	.7
Figure 24 - Summary of Project11	.8

Table of Tables

Table 1 - Reasons for Exclusion	38
Table 2 - Methods Used in Stages of User Centred Design Process	40
Table 3 - Methods Available for Sections	58
Table 4 - Participants in Focus Group (Pseudonyms Used)	74
Table 5 - Demographics of Participants (Pseudonyms used)	85
Table 6 – Alterations to Design from Paper Prototyping	99
Table 7 - Alterations to Design from Mid Fidelity Session	103
Table 8 - Alterations to Design from High Fidelity Sessions First Iteration	106
Table 9 - Demographics For Gavin	108
Table 10 - Alterations to Design from High Fidelity Sessions Second Iteration	109
Table 11 - Participants in Software Evaluation	112
Table 12 - Participant Demographics in HB's Study	142
Table 13- Themes from HB's Work	144
Table 14 - Degree of Participant Involvement	160

Abstract

User Centred Design is accepted as being essential to good software design, only by involving the users throughout the development process can the developers understand what the end users really want (Sharp et al., 2007). One area which has in the past had little experience of User Centred Design is Assistive Technology, and in particular Augmentative and Alternative Communication software (Waller et al., 2005a). Augmentative and Alternative Communication (AAC) offer the possibility of being able to access communication for people with Severe Speech and Physical Impairments.

Currently Augmentative and Alternative Communication aids have a high rate of abandonment due in part to poor usability (Prior et al., Accepted). The term abandonment has been used for a variety of different definitions in AAC and assistive technology fields (Johnson et al., 2006). For this thesis, Johnson's definition of abandonment will be used; abandonment refers to inappropriate discontinuation of an AAC device. Discontinuation of AAC devices due to a move to a more appropriate device or due to the user no longer having a need for the device are not considered to be an abandonment in this thesis.

This abandonment causes frustration and upset to the user, their family and friends and their support staff (Johnson et al., 2006).

It has been suggested that by carrying out User Centred Design in the development of Augmentative and Alternative Communication aids that their usability could be improved (Waller et al., 2005a). However, there are many challenges identified in the literature as to how to carry this out with adults with Severe Speech and Physical Impairments (Prior, 2010).

1

This thesis describes a software development study which investigated methods currently used in software development and how they could be adapted for use with this population. The way difficulties cited in the literature when working with this population were tackled are also discussed. The study involved four adults with Severe Speech and Physical Impairments in the User Centred Development of a piece of assistive software.

The study found that with careful planning it was possible to conduct User Centred Design with participants with Severe Speech and Physical Impairments, the lessons from this study were translated into recommendations and provided to a second developer who wished to work with adults with Severe Speech and Physical Impairments. The second study found similar levels of contribution to the features of the software were found in the pilot study.

This work has demonstrated the potential for adults with Severe Speech and Physical Impairments to be actively involved (i.e. contribute a high proportion of the features) in the development of Augmentative and Alternative Communication software. A number of areas for further investigation have been identified including the differences found in usability of devices developed using User Centred Design compared to traditional methods, and also how adults with Severe Speech and Physical Impairments can be more actively included in a range of research fields.

Chapter 1. Introduction

"This is my biggest complaint about new technology. It's almost like technology manufacturers think people with disabilities don't go outdoors or anywhere that's not flat. The newer AAC devices all have LCDs nobody can use outdoors and my regular power chair can't even handle a couple inches of thick grass. These things are cramping my style."

(Dattilo et al., 2008 p. 22)

The above quote belongs to a 27 year old woman with cerebral palsy. Cerebral Palsy affects how a person's brain is able to control their movements, including their ability to speak verbally. She has a college degree and is fiercely independent but is restricted in her desire to be independent by the poor usability of Assistive Technology, in particular her Augmentative and Alternative Communication (AAC) device which she uses to communicate (Dattilo et al., 2008).

It is not merely the users of AAC devices who become annoyed by the poor usability of the devices. Their friends and family also become frustrated when the devices do not operate as expected, as this quote by a mother of a 20 year old who uses AAC shows:

"The scanning was accomplished by a head switch, and was about as slow as chiselling on stone with a toothpick. This problem was further complicated by the computer's hard drive crashing about once a week. Needless to say, desire to communicate plummeted."

(McNaughton et al., 2008 p. 49)

These quotes illustrate the frustration felt by end users of AAC devices and their families when the devices do not perform as they wish. AAC devices have the potential to greatly enhance the lives of people who are unable to communicate verbally, but

currently this potential is not being fully realised due to the poor usability of the devices (Waller et al., 2005a). The devices are often complicated, do not respond in the way a user would expect and are often organised in a way that is confusing to the end user. In both industry and academia it is accepted that the best way to improve the usability of technology is through User Centred Design (ISO, 2010). User Centred Design is a methodology which demands the early and continual involvement of end users (ISO, 2010, Sharp et al., 2007). To date little work has been done into how the end users of AAC devices can be involved in User Centred Design.

The original motivation for this work came out of a need to look at the difficulties faced by people with Severe Speech and Physical Impairments¹ in hospital. A software development was commissioned by Capability Scotland which would allow medical staff to find out who the patient with SSPI was. However as the author spent time in a care centre with adults with SSPI with the aim of discovering how they currently used technology it became evident that current assistive technology was not being used effectively. The author investigated the issues of poor usability and abandonment of Assistive Technology and in particular Augmentative and Alternative Communication. A review of the literature showed that to date little work had been done in User Centred Design (one of the main techniques used in improving usability) with this group of users. This led to a wish to explore this issue and the original focus (the software for use in hospital) became the vehicle for the research.

The author used a variety of techniques to work with the adults with SSPI in the software development process. The hypothesis was that despite the published view that

¹ The term SSPI is used to indicate that the people concerned have severe impairments and will have significant problems in communication. They may also have some degree of cognitive impairment which can range from slight to profound (Redmond & Johnston, 2001).

the end users of AAC could not actively participate in the software development process, they could.

By working with the participants with SSPI on the development of the software process the author produced evidence that be adapting techniques in User Centred Design it is possible to involve them in design activities.

This work resulted in recommendations for developers wishing to include adults with Severe Speech and Physical Impairments in the development of software. These recommendations were then successfully piloted by a Master of Science student.

This thesis is divided into eight chapters. Following this introduction, Chapter Two introduces Severe Speech and Physical Impairments and AAC in more detail. The chapter then discusses the issue of abandonment of AAC devices and how this may be linked to problems in the usability of devices.

Chapter Three provides a short overview of User Centred Design.

Chapter Four contains a review of the literature regarding User Centred Design and adults with SSPI. The areas of missing research are identified. The chapter ends by setting out the aims of this research, identifying the specific research questions the research activities sought to answer.

Chapter Five presents an examination of the current methods in User Centred Design that are available and their suitability for adults with SSPI.

Chapter Six reports on a case study looking at the involvement of participants with SSPI in the User Centred Development of an item of Assistive Technology.

Chapter Seven discusses the development of a series of recommendations. These were produced from lessons learnt in Chapter Six.

Chapter Eight reports on a second case study in which a new software developer carried out a User Centred Design lifecycle with adults with SSPI using the guidelines. The results of the second study using these recommendations are also reported.

Chapter Nine contains a review of what was learnt in the study and the answers to the research questions.

Chapter Ten discusses the implications of this work and areas for future research.

Chapter 2. The Abandonment of Communication Aids

"Because I don't speak they think 'ah he doesn't have a mind of his own"

Doug, a participant in the study with SSPI

2.1 Introduction

This chapter presents an overview of Severe Speech and Physical Impairments (SSPI), with particular regard to its effect on communication. The use of Augmentative and Alternative Communication (AAC) to assist people with SSPI in communication and the different forms of AAC available are presented. The chapter concludes with an examination of problems that despite advances in technology are still prevalent in high tech AAC devices and the impact this has on the abandonment rate of AAC. It is proposed that an improvement in the usability of AAC devices could assist in reducing the abandonment rate.

2.2 Severe Speech and Physical Impairments

Severe Speech and Physical Impairment (SSPI) is an umbrella term used to describe a set of impairments that affect the motor and communication abilities of an individual (Redmond and Johnston, 2001).

2.2.1. Prevalence of SSPI

A number of different disabilities can contribute to someone having SSPI. Congenital disabilities are the major reason for children having SSPI and originate either before, during or shortly after birth (Munson and Munson, 2000). The most common congenital cause of SSPI is cerebral palsy (Larsson et al., 2009) which is thought to occur in 1 in 500 births (MacDonald et al., 2000).

Another cause of SSPI is an acquired impairment, often as a result of a brain or spinal injury from a car crash or stroke (Roy et al., 1993). The third reason is as a result of developmental delay which can result in problems with communication and mobility (e.g. Down's syndrome or severe autism) (Mundy et al., 1995, Aldred et al., 2004).

All of these causes may also result in a cognitive impairment (Wing, 1981, Hamm et al., 1993). This can range from mild learning disabilities to a profound impairment which renders the individual unable to make any decision on their own (Wing, 1981).

The number of people with SSPI is difficult to ascertain, there has been no recent empirical work done to gather data on the prevalence of SSPI (Holmes et al., 2010) and details on SSPI do not appear in the United Kingdom census (Office for National Statistics, 2011). However, the Bercow review in 2008 showed that 1% of children entering school in England have SSPI and require aids for communication (Bercow Report, 2008). This number correlates closely with figures from 1986 which put the prevalence of SSPI in the population at 1.4% (Enderby and Philipp, 1986).

2.2.2. Impact of SSPI

In order to better understand the population it is helpful to examine each of the impairments individually along with the Assistive Technology devices which are designed to minimise the impact of these impairments on a person's daily life (Robitaille, 2010).

2.2.3. Physical Impairment

The physical impairments in SSPI include problems in gross and/or fine motor control. Gross motor control refers to large general movements of limbs (e.g. shaking a leg or waving an arm) and requires the proper control of muscles, bones and nerves which must all be coordinated (Barreto et al., 1999). Fine motor control refers to small finely controlled movements such as eating, swallowing or writing. Both fine and gross motor control rely on the brain controlling the actions (Barreto et al., 1999). When the brain is damaged this can impact on the ability of a person to carry out these tasks.

There are a wide range of motor devices available on the market to help an individual move more independently (Enders and Hall, 1990). These devices include wheelchairs, walking frames, adapted cars and scooters.

2.2.4. Communication Impairment

Individuals may experience a physical inability to speak or be unable formulate the words needed for communication (Glennen and Descoste, 1997).

Additional impairments such as sensory (e.g. hearing) or cognitive impairment can affect the receptive processing of communication (Glennen and Descoste, 1997). Receptive processing refers to how a person interprets or understands what is being said to them (Demchak et al., 2002).

2.2.5. Effect of Communication Impairment on Quality of Life

Without access to communication it is incredibly difficult for a person to be able to share their needs and preferences:

"I know what it is like to be fed potatoes all my life. I hate potatoes! I know what it is like to be dress in reds and blues when my favourite colors are mint green, lemon yellow, and pinks. But then, who knew that but me" (Paul-Brown and Diggs, 1993 p. 140)

Given how difficult it can be to express even basic needs it is perhaps not so surprising that adults with SSPI often experience social isolation. Without access to communication it is almost impossible to share personal narratives and experiences, vital for building social networks (Blum et al., 1991).

People with SSPI report a greater external locus of control (this is the extent to which an individual believes they are able to control what happens in their life) (Edyburn, 2006)

and depression (Wressle and Samuelsson, 2004). Those with the greatest level of communication impairment report the highest levels of depression (Edyburn, 2006). Depression and the locus of control in the general population are also known to be related to unemployment (Osberger et al., 1993).

Only 10% of people with SSPI are believed to be in employment in the United States of America (McNaughton et al., 2002a). While figures are not available for the level of employment in the United Kingdom, Eatson (1992) reported that those with SSPI are one of the least likely groups to be employed and those who are employed will earn significantly less on average than adults who are not disabled.

It has been suggested that access to employment would be easier if the communication impairment could be reduced or its impact lessened through assistive devices (McNaughton et al., 2002a).

Much work has been done into improving physical mobility, and both the mobility Assistive Technology research field and commercial market are large (Enders and Hall, 1990). By contrast, aids or communication is a small niche market. Given the effect of non access to communication can have on an individual it is an area worthy of attention.

2.3 Augmentative and Alternative Communication

Augmentative and Alternative Communication (AAC) is the general term for the methods used to aid communication by those for whom the usual form of communication through speech is not sufficient and is an area of practice which attempts to compensate for loss of verbal communication through a variety of techniques (Glennen and Descoste, 1997).

The aim of AAC is to provide access to communication for people with SSPI (ASHA, 2011).

Augmentative and Alternative Communication (AAC) can be categorised into aided and unaided communication (Glennen and Descoste, 1997).

2.3.1. Unaided AAC Communication

Unaided communication refers to methods of communication that do not rely on external devices. For those who have adequate motor control this could include Sign Language or Makaton. For people with SSPI whose motor control is not sufficient for this, there is the possibility of using facial or hand gestures (Millikin, 1997).

The advantage of unaided AAC is that it is portable and is always available to the user. However, use of unaided AAC is dependent upon the Communication Partner² understanding what the gestures or signs mean.

2.3.2. Aided AAC Communication

Aided AAC relies on an aid which is external to the user. This category of AAC can be divided up into low-tech and high-tech.



Figure 1 - Adult with SSPI Using Paper Word Board

² The term Communication Partner is used in this thesis to refer to the person who is communicating with the individual with SSPI.(Calculator, 1997)

Low Tech

Low-tech devices can be defined as "*anything that doesn't involve electricity*" (Scott, 1998 p. 13). Low-tech devices are not expensive to produce but are often difficult for a Communication Partner not familiar with this type of aid to understand. They are also not useful in attracting a Communication Partner's attention (Scott, 1998). An example of low tech AAC would be a paper word board (see Figure 1).

High Tech

High tech AAC devices use electricity and enable users to store and retrieve messages or compose messages using symbol sets or letters. High tech AAC will usually allow users to output their messages as synthesised voice (Glennen and Descoste, 1997).



Figure 2 - Selection of AAC Devices and Access Methods (Taken from Misericordia University, 2008)

High-tech AAC aids (see Figure 2) are electronic devices that permit the storage and retrieval of messages, with most allowing the user to communicate with others using speech output (Glennen and Descoste, 1997). Modern high-tech AAC devices may also incorporate features such as: SMS text messaging, Internet browsing, E-Book readers, wheelchair controls, environmental controls and can even act as a remote control for televisions (Mayer-Johnson, 2010).

2.4 Abandonment Rate

Despite these exciting features in the devices, high tech AAC devices still have a high rate of abandonment (Johnson et al., 2006).

Previously little work has been done to investigate the level of abandonment of AAC, particularly in the United Kingdom (Holmes et al., 2010) but work is now underway to classify this figure. Various studies in other countries have put the average figure of abandonment of AAC devices at 30% (Johnson et al., 2006, Sutherland et al., 2005, Culp et al., 1986, Rory, 2009). The time between the first of these studies and the most recent would indicate that the abandonment rate of AAC has not altered significantly in the past 20 years. This means that almost one third of all devices prescribed and bought are not being used.

Each abandonment of a device represents wasted time and effort in training, a lost opportunity and a large amount of money spent (Johnson et al., 2006).

These figures relate to both high and low tech AAC devices but it is the abandonment of high tech AAC devices that causes the greatest disillusionment for users and professionals (Waller et al., 2005a) and given that they cost thousands of pounds (before even considering support costs) which must be met by the National Health Service (NHS)³, local authority or charity (Kientz et al., 2006), there is good motivation to focus on the area of high tech AAC when investigating the issue of abandonment further.

³ The National Health Service provides health care for all citizens of the United Kingdom based upon medical need as opposed to the ability to pay.

2.5 Reasons for Abandonment

There are many reasons for the abandonment of high tech AAC and not of all of these reasons are directly related to the device itself, e.g. negative family attitudes (Sutherland et al., 2005) or a lack of support for the end user (Johnson et al., 2006).

Other reasons are more closely related to the device itself and in particular the software that is running on the AAC device, e.g. the software is not flexible (Culp et al., 1986, Rackensperger et al., 2005, De Jonge and Rodger, 2006, Hoppestad, 2007) and cannot be used easily in different situations (Rackensperger et al., 2005).

It is through the software on the AAC device that the users can access communication (van de Sandt-Koenderman, 2004) and so it is vital that it works correctly for the device to be a success.

The study of the interaction between a user and software and how this interaction can be improved is known as Human Computer Interaction (HCI). HCI engineers can be seen as the bridge between the technical software engineers and the end users (Miesenberger et al., 2002). If the problems with the software are to be communicated effectively to the end software engineers they must be first understood by HCI experts.

It is proposed that by examining the reasons for abandonment in terms that apply to HCI it may become clearer to researchers in this field why the high tech AAC devices are not as successful as one would hope.

2.6 AAC Abandonment, Human Computer Interaction and Usability

Usability is a term used in HCI to describe how easily a piece of software can be used by its intended users to complete their intended tasks (ISO, 2010).

2.6.1. Means of Describing Software Usability

There are two main schools of thought when it comes to describing the usability of a piece of software (Sharp et al., 2007): Shneiderman's five usability measures and Nielsen's five usability components. Shneiderman's measures are: time to learn, speed of performance, rate of errors by users, retention of steps over time and subjective satisfaction (Shneiderman, 1998). Nielsen's components are: learnability, efficiency, memorability, errors and satisfaction (Nielsen, 1992).

Although these two different sets of components use different terms for the components they essentially are related to the same categories: how easy it is to learn to use the system, how quickly a user can carry out a specified task, how many errors the system generates and how easy it is to deal with these errors, how easy the system makes it for users to remember steps and the satisfaction the user gets from using the system.

Both of these categories are understood by HCI experts (Sharp et al., 2007). Nielsen's components were chosen in this thesis to categorise the reasons for abandonment of AAC which are related to the software, because according to Google Scholar, these components have been used more frequently in the literature than Shneiderman's measures (7363 times compared to 4663 times).

2.7 Categorisation of AAC Abandonment Using Usability Components

2.7.1. Learnability

Learning to use AAC devices can be a difficult and long process and a lack of training at this stage and throughout the life of a device has been cited as a reason for abandonment (Culp et al., 1986, Johnson et al., 2006, Light et al., 1996, Murphy et al., 1996). There is also the impact that long periods of training have on the friends and family of the AAC user which can contribute to abandonment (Parette and Angelo, 1996, Parette et al., 2000). These reasons suggest that not only is there a lack of resources when it comes to providing training but that the training itself is too long. A shorter training period would mean that there was less of an impact on a family's daily routine. This shorter training period would also allow the trainers to work with more end users and would also allow more time for follow up training. This follow up training is very important; a user often needs time to test and experiment with their device in the user's home environment for problems to emerge (McNaughton and Bryen, 2007).

The best way to reduce the amount of training needed is to make the systems easier to learn (Grossman et al., 2009). A system which is designed to help users achieve tasks more quickly and without a great deal of prior training will not require such an extensive training period (Grossman et al., 2009).

2.7.2. Efficiency

Once a user has learnt how to use the system the next measure of usability is to look at how quickly they can perform the tasks, this is known as the efficiency measure (Nielsen, 1992).

One reason for abandonment is that the user prefers a simpler method of communication which is quicker for them (Johnson et al., 2006, Smith-Lewis and Ford, 1987). The end user may also be understood by some communication partners without an AAC device and they feel that the time taken on the device to communication is too long for the reward of being able to communicate with people who are not their usual communication partner (Johnson et al., 2006, Murphy et al., 1996). These reasons would suggest that the AAC device's benefits are not enough to outweigh the difficulties in using it. The devices are too slow to be useful for a user who has any other means of communicating.

The physical effort of using the device is cited by McCall (1997) as a reason for abandonment. While this does not relate directly to the device's efficiency it does relate to efficient uses of the end user's resources. A device should be designed to require as small an amount of a user's physical capability as possible, while still maintaining its own efficiency.

Parette (2000, 1996) in work done with families of AAC users gives a lack of time for family to help with AAC devices as one reason for abandonment. This means that the devices are taking too long to do tasks, and that devices should be designed with the time that may be available in mind. A device should allow tasks to be completed as quickly as possible to help families with busy schedules. This is linked to another reason for abandonment, that there is no time to program materials (Parette and Angelo, 1996). Again if this task was made simpler and quicker to perform the demand it makes on time would be reduced.

The other problem with efficiency that a user may face is the vocabulary not meeting their needs (Johnson et al., 2006). When the vocabulary is not available a user has the option of attempting to type the word or use gestures or vocalisation to help the communication partner guess what they want to say (Culp et al., 1986). Typing can be problematic for AAC users who may not have literacy skills (Millar and Kerr, 1998). Not having access to the appropriate vocabulary can slow or even stop communication. This means that users cannot complete their desired tasks on the device efficiently (Light et al., 1996).

2.7.3. Memorability

While it is important for a user to be able to learn to use the software quickly, it is also important that they are able to easily remember how to perform a task again the next time they use the system (Nielsen, 1992). Several papers (Johnson et al., 2006, McCall et al., 1997, Smith-Lewis and Ford, 1987) state that the system being too complex or

difficult is a reason for abandonment. A system that has many different components and steps to be completed in order to achieve a task will be difficult for users to remember (Norman, 2010). This is particularly important for users with SSPI who may also have cognitive impairments. This cognitive impairment could mean that a person with SSPI has problems with short term memory problems or problems in processing multiple tasks at once (Redmond and Johnston, 2001).

2.7.4. Errors

While it is almost impossible to ensure that any computer system will not generate an error (Daran and Thévenod-Fosse, 1996), it is important to ensure that when an error does occur that it is as quick and as simple as possible for the user to recover from it.

In AAC there are frequent references in literature on abandonment to the unreliability⁴ of systems (Culp et al., 1986, Light et al., 1996, McCall et al., 1997, Murphy et al., 1996, Parette et al., 2000).

Many systems are unreliable and users of all forms of technology will be familiar with it occasionally breaking down (Parette et al., 1996). But the fact that it is being reported in the literature suggests that it is a large enough problem as to be a factor for the complete abandonment of an AAC device. This means that it is likely that the system does not make it easy for the end users to recover from errors and is too complex for errors to be corrected easily by the user and their communication partner.

2.7.5. Satisfaction

Satisfaction looks at how pleasant is it to use the system and if the user enjoys using the system (Nielsen, 1992). From an examination of the literature on AAC abandonment it

⁴ Unreliability refers to a system that cannot be depended upon to act as expected without errors or system failure. (Sharp et al., 2007).

would seem this category can be subcategorised in four areas: customisation, environment, family and communication partners.

Customisation

Ratcliffe (1997) states that lack of customisation of devices is one reason for abandonment. AAC devices are designed for a large heterogeneous group, like many pieces of technology (Enders and Hall, 1990). Most mainstream software programs can now be customised to a user's particular needs or wishes (Dourish et al., 1999). AAC devices are difficult to customise. Without this customisation, a user's enjoyment of the device is likely to be reduced (Ratcliffe, et al., 1997).

Environment

AAC devices are designed in technology labs, but it is important to remember that they will be used in a user's own environment which itself may be varied in terms of temperature, light conditions, background noise and access to electricity (Newell and Gregor, 1999). A user's environment should also look at the people in the environment and what the accepted standard of behaviour is in the environment. Devices that do not fit into the environment or which do not give due consideration to the culture of the environment are more likely to be abandoned (McCall et al., 1997, Murphy et al., 1996, Parette et al., 2000, Parette and Angelo, 1996, Ratcliffe et al., 1997, Sigafoos and Iacono, 1993). If a device is not suited to what is going on then a user may struggle to use it or may feel embarrassed by using the device. Scherer's (2001) work in the USA has shown the importance of the environment being taken in account when matching people to all forms of technology, and especially assistive technology (Chandrashekar et al., 2006).

Family

Families are often consulted during the prescribing of AAC technology and even in some cases during the development of the devices. However as stated before, this designing will be done in a technology lab (Deruyter et al., 2007). Designers often do not consider how a device will fit into current family dynamics and the effect on a family's quality of life (Parette et al., 2000, Parette and Angelo, 1996). No matter how attractive a device looks if it doesn't make life better for the family as a whole it is likely to be abandoned (Parette & Angelo, 1996).

Communication Partners

The attitude of the communication partner and their motivation is crucial for a user's successful experience with an AAC device. A negative attitude from a communication partner (Light et al., 1996) or a lack of family support can result in abandonment (Culp et al., 1986, Johnson et al., 2006, Parette and Angelo, 1996). Some of these problems may be because of negative attitudes to AAC in general but many of them come from experiences of AAC being difficult to use (Johnson et al., 2006) and difficulties in communicating with people who use AAC (Johnson et al., 2006).

If the software on AAC devices was more usable then it is possible that communication between AAC users and communication partners could become more natural and therefore the communication partners would become more motivated to provide communication opportunities with the AAC user (Johnson et al., 2007).

2.8 Abandonment and Usability

It has been suggested in this chapter that AAC devices have an unacceptably high level of abandonment. As discussed in Section 2.6 many of the reasons for this abandonment can be related to poor usability. It would seem sensible therefore to investigate ways in which the usability of AAC devices can be improved. In industry and academia it is accepted that while usability can be improved by following guidelines in design, the best way to improve the usability of software is when a developer involves the end users through a process known as User Centred Design⁵ (Jokela et al., 2003).

Surveys have been conducted which show that the majority of HCI experts believe User Centred Design to be the best method of improving usability (Mao et al., 2005) and while there is little empirical evidence to prove its effectiveness, there are many cases where User Centred Design has not been conducted and the end software has failed due to poor usability (for a larger discussion see Chapter 3). In light of the lack of evidence to the contrary User Centred Design is currently accepted as the best means of improving software usability (Lazar et al., 2010).

The next stage of this research is to investigate the extent to which User Centred Design has already been conducted with this group of users.

2.9 Summary

This chapter has shown the effect on a person when they do not have access to communication. For those whose own means of communication are not sufficient AAC can offer the means of communicating. There are many different forms of AAC available, but the form that can perhaps offer the most potential is high technology AAC.

There is a rejection rate of almost 30% of AAC devices; this means that all the effort that is required on the part of the end user, their family and their speech therapy team to select and become familiar with the device is wasted.

⁵ User Centred Design is a process for the continual involvement of end users throughout the development of a product. It is discussed in-depth in Chapter Three.
There is good motivation to investigate the extent to which User Centred Design has already been conducted with participants with Severe Speech and Physical Impairments, however before this can be carried out, Chapter Three will provide an overview of the User Centred Design lifecycle.

Chapter 3. User Centred Design

"Talk to the organ grinder, not the monkey"

Robert, a participant in the study with SSPI

3.1 Introduction

As discussed in Chapter Two the usability of the software on high tech Augmentative and Alternative Communication (AAC) devices is poor. The best way recognised by industry and academia to improve usability is by conducting User Centred Design (Vredenburg et al., 2002). This chapter takes a step away from AAC to provide a short overview of the User Centred Design process, before returning to examine the issue of User Centred Design in the development of AAC in Chapter Four.

This chapter concludes with a short example of the problems that can occur in general software development when User Centred Design is not followed.

3.2 User Centred design

User Centre Design has its origins in Gould and Lewis's 1983 paper which first put forward the three principles of design needed to create software that is easy to use (Brown-Sica, 2010). These principles were: early and continual focus on users; empirical measures using simulations; and prototypes and iterative designs (Gould and Lewis, 1983). These principles are found throughout the process that is now known as User Centred Design, but when they were first proposed in the 1980s they were not accepted by most designers (Gregor et al., 2002). Gould and Lewis continued to pursue the idea of focus on users during design and over time it gradually became more accepted. By 1999 it was given its own standard by the International Standards Organisation (1999). It is now considered the best way to improve the usability of software, above guidelines and design frameworks (Vredenburg et al., 2002).

3.3 International Standards in User Centred Design

ISO 13407:1999 (known commonly as ISO 13407) built on Gould and Lewis's principles of design (1983) to put forward four updated principles of design: the active involvement of users, an appropriate allocation of function between users and technology, iterations of solutions and multi-disciplinary design teams (ISO, 1999). There are four stages of software development: understand and specify the context of use, specify the user and organisational requirements, produce design solutions and evaluate design against requirements (ISO, 1999). The new standard for User Centred Design is ISO 9241-210 (2010) which has replaced ISO 13407. The new ISO takes the existing recommendations and provides more detail on them. Many of the recommendations from ISO 13407 are now essential criteria which must be fulfilled for a product/process to be classed as User Centred Design (ISO, 2010).

In ISO 13407 (1999) the User Centred Design process was seen as a separate process, which could run concurrently with the actual software development. ISO 9241-210 gives a process which is to be incorporated into the whole software development, not as a process on its own (ISO, 2010). The design of a piece of software should be driven and defined by the users' feedback (ISO, 2010).

Unlike ISO 13407 (1999) when iteration was a suggestion concentrated in the evaluation phase, it is now a requirement and should be carried out in every phase of the process (ISO, 2010). The differences between the two ISO processes can be seen in Figure 3.

The four processes from ISO 13407 (1999) remain in the new standard: identify the context of use, gathering requirements, producing design solutions and evaluating the solution against requirements (ISO, 2010) but are now iterated. This means that the software will be developed through several lifecycles, and the end users taking part in the development will have a greater impact on the end result as their requirements and ideas for design are being continually returned to (ISO, 2010).



Figure 3 – Comparison of ISO 13407 and ISO 9241 (Adapted from ISO, 1999 p. 6, ISO, 2010 p. 10)

3.3.1. Determining the context of use

The context of use looks at who will be using the software, the environment it will be used and what the motivation for using it is (Maguire, 2001). By the end of this section the development team must be able to provide details on the different stakeholders in the product, including the end users and their goals, the tasks the system should support and the environment in which the software will be used (ISO, 2010). Even at this early stage, the users should be involved in helping to provide the information which is used to inform the requirements and should be referred back to in forming the requirements (Lazar et al., 2010).

3.3.2. Requirements

The needs of the users and stakeholders have to be gathered while considering the context of use. The requirements should be focussing on <u>what</u> the users want to achieve and not <u>how</u> the system will support this (ISO, 2010). The requirements should also include how any constraints (e.g. "the software will be used outside") imposed by the context of use will be addressed. When there are conflicts or potential conflicts between requirements these should be resolved at this stage (ISO, 2010). How these conflicts are resolved (e.g. rationales, factors and weightings used in tradeoffs) should be noted so that they can be referred back to later in development or when an update is being created (Boehm and In, 1996)

3.3.3. Design

When conducting User Centred Design one of the major goals for the software should be to provide a good user experience. A good user experience is one where the user finds the software easy to use and enjoys using the software (Bach et al., 2009). One of the areas which has a major impact on the user experience is the design of the product (Redmond and Johnston, 2001).

There are various information sources which should be considered and used when developing design solutions. These information sources include the context of use, baseline evaluations, what is currently happening in the field, guidelines available and the research team's own experience (Jeffries et al., 1981). The users should be involved in the design process or shown the proposed solutions as this may encourage further requirements. As the process progresses the designs should become more concrete (e.g. using scenarios, simulations and mock ups) (ISO, 2010).

The interaction between the users and the systems is a crucial feature of the design process and developers should give careful consideration as to how the user will interact and not just what they will do as a result of the interaction (Patton, 2002). The new ISO takes the principle of user/researcher interaction in design further saying that it "should" be done, rather than saying it is "best" if it is done (ISO, 2010).

3.3.4. Evaluation

The evaluation of software in User Centred design is now a <u>required</u> activity (ISO 13407 stated that it is <u>essential</u>) (ISO, 1999, ISO, 2010). Evaluation should take place throughout the development process, even at the beginning when design concepts can be evaluated to better understand user's needs (ISO, 2010). The two main approaches in User Centred evaluation are user-based testing and inspection based evaluation (Redmond and Johnston, 2001).

In user-based evaluation when prototypes are used the user should be asked to carry out tasks rather than simply be shown the prototypes (ISO, 2010). The user will then report on their thoughts about the software and any problems they encountered, quantitative objective measures may be taken as well such as time taken to complete a task, or errors made (Boehm et al., 1976).

When inspection based evaluation takes place, experts use usability and accessibility guidelines or requirements to evaluate the prototype. There are a variety of guidelines available, some provide strict criteria which the software should meet, (e.g. the International Standards Office ISO/TS, 2003) while others provide more general guidance on important aspects of the software. This more general guidance is available from a variety of sources including usability websites and papers on

usability (e.g. Bevan and Spinhof, 2007). Heuristic evaluations provide a means of inspection based evaluation. Nielsen and Molich (1990) specify eleven heuristics which software should be evaluated against (e.g. is there a match between the system and the real world).

The software will also be checked against the requirements given by users in the earlier stages of the software lifecycle to ensure it meets all of them (Sreemani and Atlee, 1996). Inspection based evaluations should not be the sole method of evaluation but can be used to eliminate major issues before user testing. In User Centred design there should be long term monitoring of the product (Woods et al., 1996).

To achieve conformance with ISO 9241:210 a process must meet all of the applicable requirements and detail why any recommendations or requirements have not been followed (ISO, 2010).

3.4 Example of Problems when User Centred Design is not conducted

There are numerous examples of the problems that can occur when user centred design has not been done in the general Software Development field. One extreme case of this is the Therac-25 case. The Therac-25 was a machine designed to provide radiotherapy to cancer patients. Eleven Therac-25 machines were installed in Northern America in the 1980s (Taylor et al., 1983) and six accidents are known to have occurred in which large overdoes of radiation were accidentally administered to patients between 1985 and 1987 following a software update (Leveson and Turner, 1993).

The cause of this problem was not due to a fault in the software operation, instead it was due to an error on the user interface. The developers had attempted to simplify the user interface which had originally required the end user to input the all the settings through a computer terminal and a control panel (Edmond and William, 2004). The developers felt this was redundant and changed the software so that the settings only had to be entered by hitting the return key and then confirmed on the terminal by again hitting the return key. Little user input had been sought during the development and evaluation of the system, thus developers did not have a full understanding of how the user would operate the system.

When the system was deployed "live" the users began to hit the return key on the confirmation screen as a matter of habit, similar to a double clock on the mouse. This meant that on the six occasions detailed in the Therac-25 case report, crucial overdoses were not noticed on the confirmation screen. A review of the software in 2003 suggested that had the end users been consulted regularly from the beginning of the development process as opposed to only once the system was complete, this fatal flaw could have been avoided (Leveson and Turner, 1993).

3.5 Summary

This chapter has discussed the history of User Centred Design and the different stages in each process. An example has been provided of the problems that can occur when User Centred Design is not followed.

The methods that can be carried out in each stage of the User Centred Design process are discussed in Chapter Five.

29

Now that the process of User Centred Design has been discussed it is possible to look at how much work is currently happening in User Centred Design in AAC and to carry out a systematic literature review on the ways in which User Centred Design has been conducted with adults with SSPI.

Chapter 4. Literature Review on User Centred Design and Adults with SSPI

"Sometimes people talk to you but they don't seem to take it in, they can hear you but they don't seem to listen"

Pam, a participant in the study with SSPI

4.1 Introduction

This chapter discusses the involvement of end users in Augmentative and Alternative Communication (AAC) software development and then provides a systematic literature review on how adults with Severe Speech and Physical Impairments (SSPI) have been involved in User Centred Design.

Before the review is introduced, a short overview of the use of Systematic Reviews in software engineering is provided.

This chapter concludes by evaluating the extent to which User Centred Design has been conducted with people with SSPI and discusses the rationale behind the research question.

4.2 Current Amount of User Centred Design in Development of Assistive Technology

The first examination of the literature was to see how much User Centred Design was currently being integrated into the development of AAC and how this compared to development of other assistive technologies. Assistive Technology conferences and journals such as RESNA and sister conferences have a focus on rehabilitation engineering as a whole and do not have a focus on software engineering (RESNA, 2011). While clinical organisations such as ISAAC (Larraz and Escoin (eds), 2010) have a focus on AAC they again do not focus on software engineering. It was therefore concluded that a review of ACM ASSETS would provide a representative sample of current research projects focusing on Assistive Technology within a software engineering context. A review of the conference proceedings from the years 2006-2010 shows that on average 15% of papers on development of technology employed User Centred Design (see Figure 4). The papers from the conference proceedings were first examined by reading the abstracts, where it became clear during the abstract that the paper employed User Centred Design (e.g. mentioned using methods with end users in the abstract) or clearly did not employ User Centred Design (e.g. was a purely technical paper), these were included in the total and not examined further. Papers which were not clearly in either category from the abstract were read in full. The author examined the papers for description of how development had taken place and who the participants (if any) had been in the development of the software.



Figure 4 - Use of User Centred Design in ASSETS Proceedings Publications

While this figure is still low compared to general technology development (a review of ACM CHI (SIGCHI, 2010) the premier conference in Human Computer Interaction had 34% of papers involving User Centred Design) the figure is higher than for AAC

development. A review of papers in ASSETS shows that no papers which dealt with AAC development used User Centred Design⁶.

Due to this, an investigation was conducted into what work has been done in User Centred Design in AAC development, what methods were used and what alternatives to user involvement were used in different stages.

4.3 Rationale for a Systematic Review

There was little empirical evidence within the existing literature regarding the current state of User Centred Design and participants with Severe Speech and Physical Impairments (SSPI).

It was therefore necessary for the author to conduct a review in this area which looked at publications which had involved adults with SSPI at some stage in development of technology.

As this was a multi-disciplinary area it was important to analyse the literature from different areas consistently. In other areas of research, particularly clinical research, when multi-disciplinary literature or large volumes of literature are being reviewed, the Systematic Literature Review process have been found to be invaluable (Mulrow, 1994).

4.3.1. Systematic Review Process

Systematic reviews have a strict process to follow before one can be carried out and it is important to have a full understanding of the ten stages to this process (Mulrow, 1994). Systematic reviews begin by developing a research question (Stage One) and then by creating a review protocol (Stage Two) (Brereton et al., 2007). This review protocol specifies what electronic databases will be searched for literature, how the databases

⁶ A summary of the main reasons given by papers at ASSETS and other studies for not using User Centred Design in AAC development is provided in Section 4.5.

will be searched and how the retrieved literature will be analysed for inclusion in the review (Glanville, 2006). This review protocol should then be validated by an independent researcher (Stage Three) (Glanville, 2006). Search strings are then developed and entered into the specified online electronic databases of literature (Stage Four) (Mulrow, 1994). Once these publications are retrieved, the abstracts are first analysed to check they meet the inclusion criteria specified in the protocol before full publications are retrieved and analysed (Stage Five) (Mulrow, 1994). The remaining publications are assessed for quality and given different weightings based on this quality (Stage Six) (Glanville, 2006). It is not until Stage Seven that the data is extracted from the retrieved publications, and synthesised in Stage Eight into a form that is suitable for answering the review question(s) (Glanville, 2006). The report on the findings is then written (Stage Nine) and validated (Stage Ten) (Mulrow, 1994).

Systematic reviews are now being used in new areas, and one of these areas is software engineering (Brereton et al., 2007). The benefits that Systematic Reviews bring strongly appealed to many software engineering researchers, however very quickly it was discovered that the well tried and tested protocol for use in clinical research did not translate fully into software engineering (Brereton et al., 2007).

The main areas which are not directly translatable are sections four, five and six. Brereton et al., (2007) discuss these difficulties and offer suggestions for how these problems can be overcome:

• Stage Four – Identify Relevant Research

In systematic reviews in clinical research, search strings would be created and used in electronic literature databases, all of these databases are organised in a similar way and one search string will work in all of these databases (Brereton et al., 2007). In software engineering the main electronic databases⁷ are not all compatible in this way. Different search strategies will be needed for individual databases (Brereton et al., 2007)

• Stage Five – Select Primary Studies

In clinical research, the abstracts of the publications retrieved from the databases would be assessed to establish if it meets the criteria specified in the protocol, full publications of abstracts passing this stage would then be used to confirm that the study meets the criteria. In software engineering abstracts are often too poor to rely on when selecting studies and often the full publication, or at the very least the conclusion will need to be read as well (Brereton et al., 2007).

• Stage Six

Stage six involves assessing the quality of the studies included in the review. This often means ensuring that there is a limitation of bias in the study and that there has been both internal and external validation of the results (Brereton et al., 2007). Studies are given a weighting based on their quality and this is used when investigating the contribution the study makes to the overall results of the review. In software engineering this stage is often not required or even possible; it depends on the type of empirical study included in the review and what information the researcher is wishing to gain from the studies (Brereton et al., 2007).

Despite these significant modifications Brereton's study (2007), which involved three systematic reviews in software engineering, showed that systematic literature reviews

⁷ The main electronic databases in Software Engineering are: IEEExplore, ACM Digital Library, Google Scholar, Citeseer library, Keele University's electronic database, Inspec, and ScienceDirect (Breeton et al., 2007)

do have benefits for the field in guiding literature searches and assisting researchers who wish to gather statistical evidence on reviews.

An example of the increasing awareness of systematic reviewing in software engineering is that, since its publication in 2007, eighty-nine studies have cited Brereton's study (2007). Of these eighty-nine, seventy-seven of the studies were systematic reviews. Systematic reviews have been particularly beneficial when reviewing large areas of data from primary studies (Bailey et al., 2007) or multi-disciplinary fields (Armitage et al., 2009).

Taking this into account, the author made the decision that, due to the multidisciplinary nature of this research, it would be useful to conduct a Systematic Literature Review in answering the question of "How have studies involved adults with SSPI in User Centred Design?".

4.4 User Centred Design Literature Review

4.4.1. Rationale

There is little information in the existing literature as to the extent to which User Centred Design has been conducted with participants with SSPI in the development of software. It has been stated that this is an under researched area (Waller et al., 2005a), but no studies have investigated what the extent of this is. This review aims to answer the research question of "How have end users with SSPI been involved in the User Centred Design of software and at what stages of the process was this involvement?"

4.4.2. Method

A systematic approach was taken to searching the electronic databases, the six databases searched were: The ACM Digital Library, IEEExplore, INSPEC, hcibib, Web of Science and ScienceDirect. This search did not include medical databases (except for Web of Science and ScienceDirect which cover medicine as well as engineering and science research), the aim of the review was to investigate User Centred Design from the perspective of software engineering and it was felt that the majority, if not all, of these papers would be found within these databases (Brereton et al., 2007).

The ACM Digital Library does not support search strings and so keyword combinations were searched for individually. The other databases were searched using search strings using thesaurus terms of User Centred Design and Communication Impairment. Examples of the search strings are detailed in Appendix G. Following this an author search on recovered studies was conducted.

Inclusion criteria for this structure review were (a) published in a peer reviewed journal or conference; (b) published in English; (c) discussed the inclusion of participants with SSPI in the software development process; and (d) were empirical studies. Excluded from the review were articles which stated a product was designed as a result of user centred design but did not elaborate on this.

There are difficulties in establishing study quality in software engineering (Brereton et al., 2007) and so for this review the quality appraisal was embedded within the inclusion rules (in criteria a). All of the studies in this review had been peer reviewed; this was deemed sufficient in this instance. No time limits were applied to studies, the author was interested in all studies which had included participants with SSPI in User Centred Design and as such, did not feel a time limit was appropriate. The databases themselves covered the years 1872 (IEEExplore) to the present date.

A total of twenty-three studies were recovered, of which eight did not meet the inclusion criteria (Barreto et al., 1999, Clayton, 2006, Foreman and Crews, 1998, Small et al., 2005, Lundalv et al., 1998, Lundalv et al., 1999, Newell and Gregor, 2000, Panek et al., 1999). The criteria upon which each of these studies are excluded is given in Table 1.

Paper	Inclusion Rule	Inclusion	Inclusion Rule C	Inclusion Rule
	Α	Rule B	discussed the	D
	published in a	published in	inclusion of	were empirical
	peer reviewed	English	participants with	studies
	journal or		SSPI in the	
	conference		software	
			development	
			process	
(Barreto et al.,	Passed	Passed	Failed – Discussed	Passed
1999)			people with physical	
			impairment only	
(Clayton, 2006)	Passed	Passed	Failed – no User	Failed -
			Centred Design	commentary piece
			discussed	
(Foreman and	Passed	Passed	Failed – no User	Passed
Crews, 1998)			Centred Design	
			discussed	
(Small et al.,	Passed	Passed	Failed – no User	Passed
2005)			Centred Design	
			discussed	
(Lundalv et al.,	Passed	Passed	Failed – no User	Passed
1998)			Centred Design in	
			software. The	
			description of	
			participants does not	
			make it clear if they	
(Lundaly at al	Failed	Dessed	Failed no User	Dassad
	Conference is not	rasseu	Cantrad Dasign	rasseu
1777)	conference is not		Centred Design	
(Newell &	Passed	Passed	Failed – no	Failed –
Gregor 2000)	1 0000	1 0000	discussion of User	commentary paper
2000)			Centred Design	commentary puper
(Panek et al	Failed –	Passed	Failed – no	Passed
1999)	Conference is not		discussion of User	
	peer reviewed		Centred Design	

Table 1 - Reasons for Exclusion

The fifteen studies which met the criteria for inclusion were examined for themes relating to the four stages of user centred design as described by ISO 9241-210 (ISO, 2010): understand and specify the context of use, specify the user and organisational requirements, produce design solutions and evaluate designs against requirements (See Chapter Three). A full description of the studies is provided in Appendix H. These four stages create the User Centred Design process but are clearly separated with their own

outcomes, and by analysing the results in this fashion it was possible to see which areas of the process are currently neglected in the development with participants with SSPI.

4.4.3. Examination of Results

The results are categorised by the four phases of the User Centred Design lifecycle (see Chapter Three). These results are summarised in Table 2.

Understand and Specify the Context of Use

Understanding and Specifying the Context of Use involves gaining knowledge of the users, their characteristics, other stakeholders (e.g. assistants, families) and the environment the software will be used in. The developers should also understand the user's goals and tasks by the end of this stage (ISO, 2010).

All of the studies found in this literature review employed some technique to understand and specify the context of use. In this review seven of the studies used only a literature review of existing software and/or work done in the field (Visser et al., 2008, Allen et al., 2007, McCoy et al., 1997, Dunlop et al., 2002, Steriadis and Constantinou, 2003, Allen, 2005, Hornof, 2008, Waller et al., 2009). The two studies which did not use a literature review as part of their understanding and specifying the context of use, used observations of the users and how they currently performed tasks (Davies et al., 2004, O'Connor et al., 2006).

Other methods were used in addition to a literature review in seven of the studies such as a discussion with experts on the issues (Light and Drager, 2007, Boyd-Graber et al., 2006, Hengeveld et al., 2008a).

Only one publication worked with end users and interviewed them on where and how they would like to use the new technology (Davies et al., 2004).

Paper	Understand and Specify Context of Use	Specify the user and organisational requirements	Produce Design Solutions	Evaluate Design Against User Requirements
(Waller, et al., 2009)	Conducted Literature Review	Expert Discussion	Life Like Proxy Users	End Users Longitudinal
(Dunlop, et al., 2002)	Lit Review		Life Like End Users	End Users Short Term
(O'Connor, et al., 2006)	Distance observation of End User	Prototypes with End Users	Lo fidelity End Users	End Users Short Term
(Steriadis & Constantinou, 2003)	Conducted Literature Review			End Users Longitudinal
(McGrenere et al., 2003)	Conducted Literature Review	Interviews with end users	Range Lo-Hi fidelity End Users	
(Tee et al., 2005)	Conducted Literature Review	Examine existing systems	High Fidelity End and Proxy users	End Users Short Term
(Allen, et al., 2007)	Conducted Literature Review	Participatory Design with experts	High Fidelity End Users	End Users Short Term
(Visser, et al., 2008)	Conducted Literature Review		Lo Fidelity Proxy Users	
(Light, Page, Curran, & Pitkin, 2007)	Conducted Literature Review Expert Users Discussions		Lo fidelity Proxy Users	
(Boyd-Graber et al., 2006)	Conducted Literature Review Expert Users Discussions	Participatory Design with experts	Range Lo-Hi fidelity Proxy users	End users longitudinal
(Allen, 2005)	Conducted Literature Review	-Experts Discussion -Prototypes with End Users -Interview End Users -Examine Existing Systems -Participatory Design with Experts	Range Lo-Hi fidelity End Users	
(McCoy, et al., 1997)			Lo fidelity End Users	
(Hornof, 2008)	Conducted Literature Review			
(Davies, et al., 2004)	-Observation of users -Interviews	Multiple Prototypes with end users		
(Hengeveld, Voort, Hummels, Overbeeke, et al., 2008)	Expert Users Discussion Conducted Literature Review		Range Lo-Hi Fidelity with end users	

Table 2 - Methods Used in Stages of User Centred Design Process

4.4.4. Specify the User and Organisational Requirements

The requirements section of software development was not detailed in all of the studies in this review; eight of the studies did not provide details on how the requirements were gathered (Dunlop et al., 2002, Steriadis and Constantinou, 2003, Davies et al., 2004, Light and Drager, 2007, Hengeveld et al., 2008a, Hornof, 2008, Visser et al., 2008). Methods employed in gathering requirements included holding discussions with experts

(Allen, 2005, Waller et al., 2009) and conducting participatory design with proxy users⁸. (Allen, 2005, Boyd-Graber et al., 2006, Allen et al., 2007). Tee et al., (2005) and McCoy et al., (1997) used examination of existing systems as their only method in requirements gathering while Allen (2005) used it in addition to other methods. The involvement of end users at this stage was conducted in four studies in the requirements gathering (Davies et al., 2004, Allen, 2005, O'Connor et al., 2006, McGrenere et al., 2003). Multiple prototypes were provided for the end users in order for them to judge and provide additional requirements in three of these studies (Davies et al., 2004, Allen, 2005, O'Connor et al., 2006) and two studies used interviews with the end users (McGrenere et al., 2003, Allen, 2005).

4.4.5. Produce Design Solutions

There were no details on the design solutions for two of the studies (Steriadis and Constantinou, 2003, Hornof, 2008). The remaining fourteen studies were split between using end user participants and using proxy participants. They also varied in whether they used low or high prototypes. High level prototypes with proxy users were used in three studies (Tee et al., 2005, Hengeveld et al., 2008a, Waller et al., 2009). Tee et al (2005) also used end users with high level prototypes as did Dunlop (2002), Davies (2004) and Allen (2007). Only low level prototypes were discussed in four studies, two showed these designs to the end users (McCoy et al., 1997, O'Connor et al., 2006) with

⁸ Proxy User is a term for the use of a participant who is not an end user but who simulates being an end user when the researchers do not believe it is possible to use the actual end user (Ibrahim et al., 2007)

the remaining two showing the low level prototypes to proxy users (Light et al., 2007, Visser et al., 2008). Boyd-Graber et al (2006) discussed the involvement of proxy participants at all stages of the process.

4.4.6. Evaluate Designs against User Requirements

There were eight studies which discussed the methods by which the system was evaluated. All of these studies used the end users to evaluate the system, either in a short term study (Dunlop et al., 2002, Tee et al., 2005, O'Connor et al., 2006, Allen et al., 2007, Hengeveld et al., 2008a) or in a longitudinal study (Waller et al., 2009, Steriadis and Constantinou, 2003, Boyd-Graber et al., 2006). Of those studies which did not include details of the evaluations, six were still in development and had not reached the evaluation stage yet (McCoy et al., 1997, McGrenere et al., 2003, Davies et al., 2004, Allen, 2005, Hengeveld et al., 2008a, Hornof, 2008). The remaining two studies were only looking at a particular aspect of software design (Light et al., 2007, Visser et al., 2008).

4.4.7. Discussion

No publication in this review included direct interaction with participants in all four stages of the user centred design process as defined by ISO 9241-210 (ISO, 2010).

None of the studies included end users actively in the understanding and specifying the context of use, although three used experts in the field in discussion (Boyd-Graber et al., 2006, Light et al., 2007, Hengeveld et al., 2008a). O'Connor did not directly interact with the end users at this stage and instead used indirect observation of the end users (O'Connor et al., 2006).

Innovative methods were used to gather requirements from end users in four studies in this review (Davies et al., 2004, Allen, 2005, Allen et al., 2007, O'Connor et al., 2006). These methods included presenting participants with a variety of prototypes in order to spark discussion on what the user would and would not like to see in the final

system. While this is not the first time these methods have been used for requirements gathering, it is a change from traditional focus groups and interviews. Expert opinions were relied on in three studies (Boyd-Graber et al., 2006, Allen et al., 2007, Waller et al., 2009), although Allen (2007) also worked with end users.

The fourth step in ISO 9241-210 is the evaluation of the solution against user requirements. Within this literature review it is difficult to examine how successful this evaluation is as so few of the studies took the requirements (on which evaluations are measured against) directly from the end users. Only eight of the studies reported on the evaluations, all of these did include end users in the evaluation either in a longitudinal or short term study. It is perhaps not surprising that all of the evaluations reported were able to use end users as evaluators, as these will be people using the system once it is finished. Some considerations might need to be taken into account when planning evaluations with these users such as whether traditional evaluation techniques are appropriate with this user group. A citation search and an author search was conducted following this literature review to investigate if any papers had been published which detailed evaluations for the papers which did not report on evaluations. However, no papers were found.

It can be seen from the literature review that there has been limited work done in User Centred Design with people with SSPI, and what work has been done has been limited and has not encompassed the entire software development process.

4.5 Problems Cited in Conducting User Centred Design with Participants with SSPI

Studies on the topic of involving end users with complex disabilities, in particular those with SSPI, in User Centred Design have discussed the problems that can occur when attempting this.

4.5.1. Representative Sample

When conducting User Centred Design the participants are likely to only be a small sample of the entire user group; it is therefore important to make sure that the sample is representative of the user group. For example, if the software was being designed for use in a school one would expect that the participants would include teachers and pupils with a range of ages and a mixture of both genders (Preece, 1993).

No two people with SSPI will have an identical set of impairments or needs, and there is a great variety of user characteristics, so there are problems in finding representative users (Glennen and Descoste, 1997).

One important task in User Centred Design is to specify the characteristics of the user group and their abilities and needs. This variation of impairments and characteristics of the population with SSPI can make the task of specifying the characteristics of the user group difficult (Newell and Gregor, 2000).

4.5.2. Achieving Consensus

While ensuring that a purposive sample of participants are involved is one way to improve the validity of decisions, another challenge to achieve agreement between all the participants on a decision (Boehm and In, 1996). One risk when working with adults with SSPI is that those participants who are most adept at using their AAC device will be able to provide the most information on their preferences and requirements. This means there can be a tendency to pay too much attention to "*the articulated needs of one user*" (Newell and Gregor, 2000 p. 41).

4.5.3. Communicating Thoughts

There can be the risk of paying the most attention to participants who can communicate the most effectively, but even with these participants it can be very challenging to hold a conversation with them which goes into more detail than simple phrases, and delves into opinions and feelings (Black et al., 2010a). It is important particularly when gathering requirements or conducting design that the participants can provide good feedback and explain why they hold an opinion (Lazar et al., 2010).

Newell suggests that participants may struggle to communicate their thoughts (Newell and Gregor, 2000). The author understood this concern to mean that a participant may not be able to offer suggestions for how features of a design or discuss their views on the way a design is progressing.

4.5.4. New Technology

Enabling participants to communicate their thoughts can be especially challenging when creating new technology (Carmichael et al., 2005). AAC research is frequently at the forefront of technology and often focuses on technology that has not previously been envisaged (O'Keefe et al., 2007). Careful consideration must be given to methods centring around new technology to ensure that they are suitable and ethical for use with adults with SSPI.

4.5.5. Ethical Methods

There are a wide variety of methods used in the development of software (see Chapter Five); however when working with participants with SSPI, some of the traditional methods used to uncover thoughts may be unethical (Braun et al., 2010).

For example, one of the traditional methods for evaluating the usability of software is to watch a participant attempt to use it and monitor the mistakes they made. It could be deemed unethical to do this with a user with SSPI as they may not fully understand and become distressed when feeling that they are making mistakes (Newell and Gregor, 2000).

Another issue is that often one of the measures of success is the level of disappointment shown by participants when the software is taken away (Newell and Gregor, 2000),

again this may cause undue distress to participants with SSPI, particularly if the software has improved their ability to communicate.

4.5.6. Legal Issues

Another ethical issue to consider in User Centred Design is how informed consent can be given by participants. One reason often given by researchers for not working with adults with SSPI and using proxy participants instead, is a difficulty in gaining informed consent (Braun et al., 2010). Many developers believe that participants with SSPI could be deemed legally 'incompetent' and this would mean they are unable to take informed consent (Newell and Gregor, 2000). It may also be difficult for a researcher to be sure that they are obtaining full informed consent.

An additional legal issue is that often in research, participants are recruited with the offer of payment, however payment may conflict with rules on benefit payments (Newell and Gregor, 2000).

Research projects frequently offer participants remuneration in some form in return for their assistance in their work. This appeals to students who will volunteer regularly and ensures a researcher can have enough participants to carry out their work. This can cause problems when working with adults with SSPI as it may interfere with benefits rules (Newell and Gregor, 2000). Some participants with SSPI may be employed and not receiving incapacity benefit, but will probably still receive some form of benefit for assistance or care support, any additional significant payment may affect how this is calculated. For the majority of adults with SSPI who do not work any form of payment can affect their benefits. Their benefits package is likely to be complicated and encompassing many different forms of benefits. It can detrimental to a participants financial situation to receive money (Newell and Gregor, 2000).

4.5.7. Purchase of Product

As participants with SSPI frequently rely upon benefits, they are unlikely to have much, if any, disposable income (Enders and Hall, 1990). The end user is the participant in the research but is often not the end purchaser (Newell and Gregor, 2000). This alters the usual business model. In a traditional technology route to market the product will be built around the market (that is the end user) problem and the channel (that is the way it is sold) will be designed in the way that the market wants to buy (Quesenbery, 2000).

There are examples however of other projects which have included people with other disabilities in user centred design (Ma et al., 2007, Adamo-Villani, 2006, Matthews et al., 2005, Petrie et al., 2006). Guidelines have been created for those wishing to develop technology for people with disabilities in general (ISO/TS, 2003, ISO, 2008) but these guidelines focus more on the technical specifications rather than usability.

At present there are no projects which include adults with SSPI in all stages of the User Centred Design software development process which could act as an example to software developers wishing to conduct similar work, nor are there any guidelines available on how to conduct user centred design with participants with SSPI.

4.6 User Centred Design and SSPI

The literature review has shown that to date limited work has been conducted in User Centred Design with adults with SSPI in the development of software. Before researchers can investigate if end user involvement in the design process reduces the abandonment of AAC devices, first an investigation into the feasibility of adults with SSPI being involved in User Centred Design must be carried out.

While it is important to include a wide variety of stakeholders in the development of assistive technology, the contribution that the end users themselves also bring cannot be overstated (Vredenburg et al., 2002). The end users cannot be expected to design a

system entirely on their own, but their views and opinions should be valued and taken into account throughout the design process (Olsson, 2004).

The remainder of this thesis concentrates on the involvement of the end users in the User Centred Design of assistive technology, while it is anticipated that in a full industrial project a variety of stakeholders would be involved, for the purpose of this work the intention was to establish if the end users could be involved fully.

4.7 The Research Question

The literature shows that communication is vital for humans and that for those who cannot communicate verbally AAC devices have the potential to greatly improve the quality of their lives (Glennen and Descoste, 1997, Beukelman and Mirenda, 1992). In Chapter Two it was shown that at present this potential is not being realised and that many AAC devices are being abandoned. In other areas of software development it is accepted that when end users are involved as co-designers through User Centred Design the usability of technology increases. Currently little work has been done in User Centred Design with participants with SSPI and investigation into what difference this makes to the usability of devices is required.

The literature cites several problems with involving participants with SSPI in User Centred Design and at present it is not possible to establish the difference that conducting User Centred Design in development makes to the usability of software for people with SSPI. Before this investigation can take place first it must be established if adaptations to methods can enable adults with SSPI to take part in User Centred Design.

The aim of this study is to determine the degree to which adults with SSPI can be involved in the design of software to facilitate communication processes, by adopting a User Centred Design paradigm. The research questions in this study are:

- 1. Can the problems cited in the literature be overcome to allow User Centred Design to be conducted with participants with SSPI?
- 2. Can methods currently used in Human Computer Interaction be combined and adapted to allow people with SSPI to contribute to the design of technology?

The first research question focuses on how the problems cited in the literature are dealt with throughout the user centred development of software while the second investigates the degree of involvement of the users. The degree of involvement will include a measure of the different contributions in each stage of the process.

4.8 Summary

This chapter built on the discussions in Chapter Two and Chapter Three on SSPI, AAC and User Centred Design, it introduced a review of the literature which shows that to date no study has included adults with SSPI actively in all stages of the User Centred Design process. It has also highlighted the problems which are discussed in the literature on conducting User Centred Design with adults with SSPI. This chapter concluded with the rationale for the research question which will be investigated in the remainder of this thesis. The first step will be to evaluate the suitability of existing User Centred Design methods for this population and the ways in which they will be adapted (see Chapter Five). Following this a pilot study involving adults with SSPI will be conducted (Chapter Six).

Chapter 5. Methods in Software Development with Adults with SSPI

"What we're saying is, what we want the most, is inclusion"

Robert, a participant in the study with SSPI

5.1 Introduction

In order to answer the research questions a software development project with User Centred Design was planned. This chapter provides an overview of the methods in User Centred Design along with a discussion on their suitability for participants with Severe Speech and Physical Impairments (SSPI) will be discussed. Possible adaptations to these methods are also suggested.

Many of these methods can be used in multiple stages of the User Centred Design lifecycle (see Chapter Three) and so are categorised by theme rather than grouped according to the lifecycle stages.

5.2 User Centred Design Methods

User Centred Design methods differ from other software development methods (e.g. performance measurements which investigate the speed software can automate a task (Sharp et al., 2007)) in that they include participants.

5.2.1. Ethnographic Methods

Most software developers have limited experience of SSPI or how the environment that people with SSPI live in (e.g. care centres) operate (Waller et al., 2005a). It would therefore seem logical to meet people in the environment, observe the day to day workings of a centre and perhaps become involved in some of the centre's activities. When these methods are used the process becomes known as ethnography (Hughes et al., 1994).

Ethnography is now becoming an accepted process in HCI with one of the most famous examples coming from the 1970s (Suchman, 1987) when a study was conducted into the use of the electronic help system of a photocopier. By the use of ethnographic methods, HCI researchers can understand the context surrounding the information's location and users.

The two main methods in the ethnographic process are observations and focus groups:

The use of observation for the user centred design phase of 'context of use' with adults with SSPI is not new (O'Connor et al., 2006), however previously it has been in the form of non-participant observations. Previous studies have shown the value in conducting ethnographic research with people with disabilities (e.g. Davidson, Howe, Worrall, Hickson, & Togher, 2008; Gordon, Ellis-Hill, & Ashburn, 2009).

Focus groups are now used as a means of evaluating the experiences of people with health and social services and in action research projects (Kitzinger, 1994; Willson et al., 2005). Focus groups are also commonly used in requirements gathering to identify the requirements from different groups of people and discuss any conflicting wishes in the design, but can be used in any stage (Nuseibeh and Easterbrook, 2000).

Focus groups are useful in gathering many different views on a topic and insights into people's shared and different opinions (Gibbs, 1997). In a focus group the discussion between participants and sharing of views may lead to a greater disclosure and volunteering of information; it also allows a researcher (or designer) to discover

why a participant holds a certain view (Gibbs, 1997). Participants may be encouraged or reminded of something they wish to share as they piggyback on each other's statements (Rennekamp and Nall, 2000).

Focus groups can be difficult to analyse and it can be difficult to get all of the participants to participate (Krueger and Casey, 2009), often one or two members can dominate the discussion.

Focus groups are now becoming more common in HCI as can be seen by recent panels and workshops at conferences such as ACM CHI (Rosenbaum, Cockton, Coyne, Muller, & Rauch, 2002).

There is evidence from the social and health fields of adults with SSPI being involved in focus groups (Hemsley et al., 2008b, Dattilo et al., 2008) and it would appear that with careful planning it should be possible for this to translate into the HCI area.

5.2.2. Interactive Methods

Interactive methods involve the participant and the researcher discussing topics and engaging with one another. They are useful because they allow the researcher the opportunity to probe further into why a participant thinks a certain way, but there is the risk that if a good rapport cannot be established between the participant and researcher that little data will be collected (Cartwright and Cartwright, 1974)

Forum Theatre

A forum theatre performance will generally consist of a short drama designed to spark conversation and feedback from the audience on what they have just seen (Boal, 2000). It will typically have some form of controversy and stop at the critical point to encourage audience members to participate. Its use as a requirements gathering method in computer design is increasingly being recognised (Newell et al., 2006a).

Forum theatre is particularly suited when the technology being designed does not yet exist as it allows participants to imagine how it might be used (Rice et al., 2007). The disadvantage of forum theatre is that participants might become totally focussed on the characters in the film and will not be able to generalise (Carmichael et al., 2005).

Forum theatre would seem useful for participants with SSPI to help them visualise the situation and to feel more comfortable sharing their views (Newell et al., 2006b).

Storyboarding

Storyboards were first used in the film industry to represent the major sections of a scene, when arranged in order it helped the film crew to visualise the film (Kinchin et al., 2000).

In software development the storyboards depict visually the interaction between the user and the system (Leveson, 1995). Storyboards can be sketched quickly as stick people or can be crafted more elaborately. Storyboard can be created quickly but they can also become out of date quickly.

Storyboarding allows requirements to be captured in graphical form rather than in text. Storyboards explore alternatives to what already exists or to test the feasibility of an approach (Kinchin et al., 2000).

Storyboarding would appear to be a good fit for use with participants with SSPI as they can help the participants visualise the new technology being developed and help to prompt for new requirements.

Interviews

There are three major forms of interviewing: unstructured, structured and semistructured (Britten, 1995). Unstructured interviews take the form of a general conversation around a particular topic and tend to be exploratory in nature whereas structured interviews are composed of predetermined questions and do not deviate from these. Semi structured interviews will have a combination of open and closed questions (Lazar et al., 2010).

When working with adults with SSPI interviews can be useful in providing them with the time they need to formulate an answer, but the researcher must ensure that the participant knows them and is comfortable with them so that they do not become overly anxious which can affect communication (Beukelman and Mirenda, 1992). If interviews are being used early in the process there is the difficulty of helping participants to imagine new technology and pre-prepared tools for prompts may assist with this.

5.2.3. Testing Methods

Testing methods will typically be used towards the end of a development to judge how effective the system is. They may however, also be used at the beginning to investigate how well existing technology can perform tasks.

Usability Testing

Usability testing is usually used in the latter half of the product lifecycle when evaluating prototypes and designs (Battleson et al., 2001). There are a number of measures that can contribute to this testing including: time taken to complete a task, number of errors made etc⁹. Usability testing can provide quantitative figures to produce statistical results on improvements in task performance with the new system (Nørgaard and Hornbæk, 2006). Usability testing can be expensive and time consuming and there is the risk that users may feel they are being tested (Virzi, 1992).

It is likely that several modifications would be required to allow this method of evaluation to be used with participants with SSPI.

Think Aloud

Think Aloud is usually used in evaluations when participants are using the system (Sharp et al., 2007). In think aloud a participant is asked to speak aloud their thoughts as they complete tasks on a system.

Think aloud can feel awkward to participants and many complain that it feels unnatural (Nielsen et al., 2002). It also has a high cognitive load (Lazar et al., 2010).

For participants with SSPI there is first the practical problem of how they would communicate on a device while using the system they are evaluating, there is also the significant concern on the extra cognitive load it requires.

5.2.4. Prototyping

There are three levels to prototyping: low fidelity, mid fidelity and high fidelity (Blum et al., 1991). Co-design prototyping where participants are involved in shaping the design is confined to the designing stage of the User Centred Design lifecycle (Westerlund et al., 2003). Prototypes which are created by the designers and used in requirements gathering have been used successfully in studies in the past

⁹ For this thesis the term error is used to indicate a situation in which a user is unable to complete a task. A situation where a user takes a path through a task which is unexpected but still arrives at the specified end point and completes the task is not considered an error.

(Rudd et al., 1996) but this comes with the risk that participants will be led by the prototypes.

Low-fidelity prototypes are often made from paper screen mock ups, when a user selects an option on the paper, the researcher or designer will then place another piece of paper with the results of the action displayed on it (Blum et al., 1991). Low fidelity prototypes are most useful at the beginning of the design process to understand better how a user will interact with the system and to identify any major design problems early on (Blum et al., 1991). Low fidelity have the advantage of being created quickly and cheaply, but are less useful for user evaluation as they do not simulate user interaction well. High-Fidelity prototypes will look more like a final device and are very useful for looking at interaction, however as with evolutionary prototypes there is a danger that users will be unwilling to give criticism as the prototypes will look finished(Rudd et al., 1996).

When using prototypes for adults with SSPI it may be best to start with lower fidelity prototypes so that participants are more likely to give criticism from the beginning of the project. If participants are shown high fidelity from the start there may be a reluctance to give criticism for fear of offending the developer of a finished product.

5.2.5. Indirect Methods

Indirect methods will often involve asking the participant to provide information without ever having met the researcher or with them only present for the taking of informed consent and explaining the method. The advantage is that the participant may feel more comfortable sharing their opinions when they are not in direct contact with the researcher (Sharp et al., 2007). These methods do not allow the researcher to probe further into a participant's view.

Diaries

When using diaries, participants are asked to record their thoughts or experiences on a particular subject, this method can be used before design commences to evaluate an existing experience or later during the system evaluation to see how well it worked in situ (Lichtner et al., 2009). Diaries can be useful as they allow self-reflection by the participants, but they are often forgotten or neglected and are not a rich source of information (Dickinson et al., 2007).

Diaries could be problematic for participants with SSPI who are likely to have problems with motor control for writing and may also have literacy problems (Glennen and Descoste, 1997). One possible adaptation is that the diaries could be produced on a computer which may reduce some of these difficulties.

Questionnaires

Questionnaires are similar to interviews in that they can have open and closed questions, they can be used at any stage of the development lifecycle (Griffith et al., 1999). Questionnaires have the advantage that participants may feel more comfortable answering personal questions when they are not directly communicating with the researcher but the researcher is not usually available to clarify queries which may lead to confusion over what the question is asking (Goodman, 1997). Participants who are not clear on what a question is asking may omit to answer it, this can cause problems when looking for statistical evidence from answers.

When working with participants with SSPI, researchers must ensure the questions are easy to understand and may need to develop new ways of providing the questions and getting answers so that participants who are not literate can answer them. Having
someone else read the questions and write the answers may reduce some of the benefits of questionnaire in allowing participants to respond privately.

5.3 Choice of Methods

The methods and the sections of the User Centred Design lifecycle that they are suitable for are detailed in Table 3.

Method	Identify Context of	Gather Requirements	Produce Designs	Evaluate Design Against
	Use			Requirements
Diaries	*	*		*
Interviews	*	*	*	*
Usability Testing				*
Questionnaires	*	*	*	*
Think Aloud	*		*	*
Forum Theatre		*		
Storyboarding		*		
Prototyping		*	*	
Focus Groups	*	*	*	*
Observation	*	*		*

Table 3 - Methods Available for Sections

Having looked at all these methods it is necessary to see if the reasons for not using them with participants with Severe Speech and Physical Impairments are valid. In order to do this, a User Centred Design software development project was undertaken to develop a multimedia patient profile.

5.4 Summary

This chapter introduced the major methods used in User Centred Design were presented and their suitability for use with participants with SSPI was discussed.

It became apparent that not all of the methods commonly used would be entirely suitable for this user group and the choice of which methods to use was taken seriously and given careful consideration.

Chapter Six will discuss how the methods were implemented and the effect they had on shaping the software being developed. Chapter Six will also discuss the means by which ethical approval for the project was granted and how informed consent was given by participants with SSPI. Work from this chapter contributed to a paper presented at the ACM CHI 2010 Conference (see Appendix A)

Chapter 6. Creation of CHAMPION Software with Adults with SSPI

"What we do matters".

Doug, a participant in the study with SSPI

6.1 Introduction

This chapter looks at the development of an Assistive Software program using User Centred Design with adults with Severe Speech and Physical Impairments (SSPI). Background information on the software developed and the rationale for choosing it is outlined at the beginning. This is followed by a discussion of the ethical problems in working with adults with SSPI that had to be overcome before work could commence.

For each method used in the different stages of the User Centred Design lifecycle, a description of the method used and the adaptations made are described. Following this, two separate sets of results are provided, these results related to the two different research questions.

6.1.1. Research Question One

The first research question investigated if the problems cited in the literature could be overcome to allow User Centred Design to be conducted with participants with SSPI. The results for this question report on the way participants responded to methods, their ability to work as co-designers and the input they had.

6.1.2. Research Question Two

The second research questions investigated if methods currently used in Human Computer Interaction (HCI) (see Chapter Five) could be combined and adapted to allow people with SSPI to contribute to the design of technology. The results to this question are grounded in HCI theory and report on the requirements given or the design alterations requested.

Following each stage in this chapter a review is presented of what worked and what did not work in the stage.

This chapter concludes with details of the debrief of the participants on their reflections of participation in the project.

Following this chapter a discussion is presented which looks at the lessons learnt in the development and how they can be used by other developers.

6.2 Choice of Assistive Software

As was established in Chapter Two, the poor usability of software in Augmentative and Alternative Communication (AAC) devices is linked to the high rate of abandonment and there are suggestions that if User Centred Design could be conducted in the development of this software then the usability could be improved. In order to investigate how User Centred Design could be conducted with end users of AAC (adults with SSPI) the author decided to produce a piece of software that would allow users of AAC to share information. It was not the aim of this research to produce software that was fully functioning and capable of communicating and there is evidence in the literature that a 'Communication Partner'¹⁰ can influence the satisfaction a user gains from the AAC device (Kent-Walsh and McNaughton, 2005). Instead it was decided to develop a tool by which users with SSPI could store and share information about themselves without requiring a communication partner.

¹⁰ A communication partner is someone who sends and receives messages with the person in question (Kent-Walsh and McNaughton, 2005)

The software chosen for development was an Electronic Patient Communication Profile, this met the criteria of allowing a person with SSPI to share information without a communication partner and was also a piece of software which was reported as being needed by this population.

6.2.1. Background to Electronic Patient Communication Profiles

This section provides a brief rationale for the choice of software. A fuller explanation is provided in .

A visit to hospital is traumatic for both a disabled patient and family members, especially when the patient has no or limited functional speech (Bartlett et al., 2008).

Adults with SSPI, in particular those with cerebral palsy, are 2.2 times more likely to attend an outpatient clinic than their peers with no disability and have a 10.6 times higher risk of being admitted into hospital as an inpatient (Young et al., 2007). Recent research with adults who are unable to speak in hospital has indicated that hospitalisation presents special challenges to both the families and the staff who care for the patient with moderate to profound communication impairment (Hemsley et al., 2001, Hemsley et al., 2008a, Hemsley and Balandin, 2004). Adults with SSPI are at higher risk of suffering from a preventable accident or medical error while in hospital (Zinn, 1995).

Currently the most commonly used method for assisting medical staff is a paper care book compiled by the care attendants of the adult with SSPI, providing information on their needs and habits (Millar, 1997). However, these have generally been overlooked by nurses, and patients themselves feel they are of little, if any use (Hemsley et al., 2001). The use of multimedia as a form of advocacy is gaining popularity amongst adults with SSPI or learning disabilities, and has been well received by professionals using it to make decisions on care for the adult involved (Grove, 2003).

There have been a suggestion in the literature that Electronic Patient Communication Profiles could help with this problem (Prior et al., 2009). This software fitted the requirements in section 6.2 in that it allowed a user of AAC to share information about their life without requiring a communication partner, and was chosen by the author for this pilot.

6.3 Planning the HCI Process for Adults with SSPI

When planning the HCI process, the first step is to consider the ethical issues in the study and to gain ethical approval (Molich et al., 2001). When working with participants with SSPI there is a high chance that ethical approval will have to be granted by a NHS ethics committee as many of the day and residential centres for adults with SSPI are run by the NHS or a local authority (Department of Health, 2003). Any time that NHS or Social Care resources (e.g. staff, buildings or service users) are used in a project it must be given approval by a NHS Ethics Committee (IRAS, 2011). When work is being carried out outside of the NHS, approval should come from an internal ethics board.

The Communication Hospital and Multimedia Patient Information Organisational Network (CHAMPION) project had ethical approval from both the internal University Ethics and the NHS Tayside Ethics Committees. In total there were three main alterations to a traditional ethics procedure (see Figure 5).

6.3.1. Capacity to Consent

Gaining informed consent from participants involved in research is seen as one of the most important aspects of ethical research (Mauthner et al., 2002).

When working with adults with SSPI in research there are two main issues to consider in relation to this. Firstly, gaining consent must provide participants with the opportunity to ask questions and when working with participants with SSPI, researchers must devote considerable extra time to allow for this and to ensure participants feel confident in asking all of their questions. The other issue is, given that participants may have some level of cognitive impairment how can researchers be sure that they are fully aware of what they are agreeing to (Balandin et al., 2006).

Adults with SSPI are at a higher risk than the general population in having a literacy impairment (Millar and Kerr, 1998, Pagulayan et al., 2003). Terms in consent forms and participant information sheets may contain technical and/or experimental terms which are unfamiliar to those with no previous experience of participating in research projects (Dickinson, Arnott, & Prior, 2007), and confusing to those with a lower rate of literacy than those who traditionally participate in research projects, such as students.

The modified consent process by Balandin et al., (2006) uses a series of multiple choice questions which the person responsible for taking the consent should ask a participant. All of the questions should be answered correctly for it to be classed as informed consent (Balandin, Berg, & Waller, 2006). This was used by the author throughout the study.

6.3.1. Stress on Participants

The NHS Tayside Committee was concerned that being asked to work on a piece of technology that had not yet been created would put too much of a strain on participants.

The author made it clear to the committee that they would attempt to minimise this as much as possible and would reassure participants frequently that there was no wrong answer and would be careful to watch for signs of fatigue.



Figure 5 - Ethical Procedure in CHAMPION Development

6.3.2. Recruitment of Participants

Traditionally participants would be recruited through adverts in the local press or University or by being contacted directly from researchers who have been given their contact details by organisations (Krueger and Casey, 2009). These methods are not best suited to working with adults with SSPI. Firstly many people with SSPI have literacy problems (see Section 6.5) and secondly the author did not want to distress people by inviting them to take part and then discover that the participant was not capable of providing informed consent, and have to tell them that they could not actually participate. Contact was made with a local centre for adults with a variety complex disabilities including SSPI. The centre management was provided with details of the study. The management then provided a site specific study co-ordinator to be the liaison on the project. The author met with the study co-ordinator and gave further study details and asked them to consider potential participants. The study co-ordinator met with potential participants and with their consent arranged a meeting between them and the author.

In this meeting the author went through the NHS Patient Information with the potential participants, discussed the activities they would be invited to take part in and what their rights in the study were. The participants were encouraged to ask questions and were then taken through the modified consent process.

The study co-ordinator attempted to select a representative sample of participants but the number of adults with SSPI who had the ability to consent was limited. There was a spread of ages, genders and means of communication but the sample could not be classed as a true representative sample.

Once ethical approval had been granted, the User Centred Design cycle could begin (see Chapter Three). The methods used in each stage of the User Centred Design cycle are shown in Figure 6.

6.3.3. Potential for User Centred Design with Adults with SSPI

The NHS ethics committee approved this process and this method of gaining participant consent. The study co-ordinator was able to consider the requirements for participants and selected as representative a sample of participants as possible. The staff co-ordinator would appear to be a good judge of the potential for participants to be able to consent as all of the potential participants were able to pass the informed consent questions and provide informed consent.

6.3.4. Potential for Adapted Methods to Allow Adults with SSPI to be Participants

The participants were able to complete the adapted consent forms. Initially participants were reluctant to ask questions on the information sheet and consent form, however thanks to the assistance of the staff co-ordinator they were encouraged to give serious consideration to their rights and responsibilities in the study. All of the consented participants asked at least one question regarding the study (e.g. "*What would you do with the work I have already done if I stop coming to meet you?*").

6.4 Identifying the Context of Use Through Observation

In this stage the aim is to identify the users' attributes, the organisation environment and the current tasks the user performs (ISO, 2010).

The author, in common with most software developers had limited experience of how care centres for adults with SSPI operated. The author decided to spend time at the centre to learn more about this environment. The primary aim of the observation was to gain an understanding how the environment operated and the characteristics of people in the environment.



Figure 6 - Methods Used in Development Process

6.4.1. Method

A major challenge for the ethnographic researcher is becoming immersed in the environment and being seen as a member of the environment. Researchers must decide on the degree of participation they will take in the environment (Hammersley and Atkinson, 1994). In some studies ethnographers hide their identity while in others they merely observe, although it could be argued that only observing is not strictly ethnography as the researcher is not participating in the environment (Hammersley and Atkinson, 1994).

In the care environment, concealing the author's identity would not be ethical or safe. The author declared to everyone her role but became immersed in the environment by providing assistance to staff and residents alike, for example helping with setting up devices or making coffee. This avoided the risk of missing out on vital information by becoming a complete observer (Gold, 1958). Using Gold's categorisation (see Figure 7) the current author took the role of "participant-asobserver" (Gold, 1958).

Complete Participant- Researcher's identity and purpose is not known to those being observed.

Participant-as-Observer – Identity is known by those being observed. Researcher takes part in community.

Observer-as-Participant - Identity is known, more formal, less direct interaction with those being observed.

Complete Observer – Identity is known. No direct interaction with participant.

Figure 7 - Forms of Observation

Field notes were taken throughout the observations and a digital camera was used to record incidents of note, such as a participant demonstrating how they currently carry out a task on the computer (see Figure 8).

To begin an ethnographic study the researcher must build a relationship with participants in the group. The author started with the study co-ordinator provided by the care centre and relied on her to make the initial introductions. Gold (1983) cautions that relying on one member of an environment to make introductions can open the researcher to a real risk of bias by only meeting people whom the initial member wishes them to (e.g. colleagues with the same perspective on issues being investigated). The author was careful to avoid the risk of bias as much as possible by asking to be introduced to a variety of people. A care centre will generally be used to a high staff turnover and used to new people arriving (Cumella and Martin, 2000). The staff at the care centre adapted very quickly to the current author's presence and after the initial few sessions they began to treat her as a member of staff.

The study coordinator offered to allow the author to join them in their usual work, look around the different facilities and observe activity sessions to see how these facilities were used and talk to service users and how they felt using technology.

6.4.2. Potential for User Centred Design with Adults with SSPI

The participants were very welcoming towards the author and were eager to engage in conversation, particularly when it was on a one to one basis. One participant informed the author that he did not often get an opportunity to engage with people from outside the centre and that he was always glad of the chance to be involved in conversations. The staff were more cautious initially but after the first two days would willingly talk to the author and shared their opinions on technology and how the centre members could use it.

The adults with SSPI were able to explain to the author how the centre operated and showed her around the centre explaining how things worked.



Figure 8 - Participants Demonstrating Current Use of Computers¹¹

6.4.3. Potential for Adapted Methods to Allow Adults with SSPI to be Participants

This method did not generate many concrete results. Instead it gave the author a better understanding of the participants' needs and the roles the different members of staff played.

6.4.4. Review of Observation

In order to further understand how service users felt about the idea of using software such as the patient profile and to enable them to discuss problems they had

¹¹ To protect the anonymity of participants, actors with SSPI have been used to recreate the original scenes in some photographs

experienced in which the software might have helped the methods employed by the end users to communicate. The focus group would also help the author to understand more about the characteristics of the end users.

6.5 Focus Groups with Adults with SSPI

Work from this stage contributed to a paper in The Journal of Behaviour and Information Technology (see Appendix C).

Focus groups have been used successfully with adults with SSPI in the fields of health care and social studies (e.g. Hemsley et al., 2008a), but to date have not been used with this user population in HCI (Prior et al., Accepted).

6.5.1. Method

After an examination of the current state of the field for focus groups with adults with SSPI (Prior et al., Accepted) and in light of the lack of guidelines available, the author looked at the general additional considerations when working with adults with SSPI.

Communication Considerations

Focus groups centre around the communication interaction between participants (Morgan and Krueger, 1993) but this can pose problems for adults with SSPI.

People who use a high tech AAC device for communication may produce words up to 25 times slower than those with normal verbal speech (Higginbotham, Bisantz, Sunm, Adams, & Yik, 2008). The time therefore for a participant to reply, for example "Yes, that has happened to me" could take up to two minutes for a participant relying on AAC (Augmentative and Alternative Communication at the University of Washington, 2009). For those participants using low tech AAC, a facilitator is likely to be required for the focus group. The facilitator would have to follow the participant's finger pointing and speak the message on their behalf (Prior et al., Accepted)

Literacy Levels

As discussed in Section 6.3.1 participants with SSPI may have low or limited literacy and in focus groups participants are often to take their own notes on certain discussions or be given 'prompts' to help with the discussion. These prompts will often be newspaper articles or written scenarios and discussions (Krueger and Casey, 2009). It may be necessary to consider what if any prompts can be used with participants with SSPI.

Concentration Levels

People with cerebral palsy or traumatic brain injury (two of the major causes of SSPI) are likely to become tired more quickly than the general population (Wood et al., 2008, Jahnsen et al., 2003b). The average length of a focus group is between ninety minutes to two hours (Kitzinger, 1995; Krueger & Casey, 2009); this will require an extended period of concentration and people with SSPI are likely to fatigue in this time.

Physical Considerations

Focus Groups have traditionally been conducted around a table (Krueger & Casey, 2009), with a moderator at one end of the table and possibly a moderator's assistant at the other, the assistant has the role of taking notes.

Participants with SSPI are likely to have a range of motor impairments, which may restrict their ability to sit at the table and/or may require them to use a wheelchair (Finnie & Bavin, 1997).

As a focus group is primarily a study of the interaction between participants (Morgan & Krueger, 1993), the layout of a group requires careful consideration to ensure that participants are positioned in a way that encourages conversation and debate.

Participant	Method of AAC	Other Means of	Established Group
		Communication	Membership
Danni (29)	Lightwriter AAC Device	Vocalisation	Lives at resource centre
Robert (58)	Gestures	Dysarthric Speech	Attends resource centre three days a week
Doug (53)	Dynavox AAC Device	Dysarthric Speech	Attends resource centre four days a week
Patricia (44)	Signing	Dysarthric Speech	Attends resource centre five days a week
Lisa (40)	Facial Gestures	Dysarthic Speech	Attends resource centre four days a week
Sam (30)	Hand gestures	Dysarthric Speech	Attends resource centre two days a week

Table 4 - Participants in Focus Group (Pseudonyms Used)

6.5.2. Participants

For this innovative focus group, six adults with SSPI who were capable of giving informed consent using the process described in Section 6.3.1 and who received services at the care centre either as day users or residents were recruited through the study coordinator at the centre (see Table 4). The study coordinator was asked to recruit as wide an age range as possible with an equal split of males and females. It was also hoped that participants would use a mix of communication methods. While again this could not be considered a true representative sample, it was hoped that the credibility of the convenience sample could be as high as possible.

The group of participants ranged in age from 29 to 58 and were composed of three female and three males. Five of the participants were day users of the centre and one was a resident user.



Figure 9 - Layout of Focus Group

6.5.3. Layout of Focus Group

The focus group was held at the centre in a room used for activity sessions. As all of the participants used motorized wheelchairs a semi-circle was formed with facilitators placed next to the participants who required the most help with their communication. One participant had a hearing difficulty in addition to his speech impairment so a facilitator was sat next to him at an angle to allow for better communication of the questions and to assist with signing of questions.

The moderator sat at the top of the semi-circle with the video recorder placed to the left of the moderator to ensure all the participants were in view (see Figure 9). The moderator's assistant sat opposite the moderator and took notes throughout the session.

6.5.4. Format of Session

Participants were asked for their experiences of sharing information with medical professionals (see Figure 10). This would give further information to complement the information in the literature on electronic personal care records.



Figure 10 - Participants In Focus Group Discussing Bad Hospital Experience

Due to the increased chances of fatigue the session only lasted sixty-two minutes with a further seven minutes for debriefing. Participants were told they could take a rest or leave at any time. During the session, one participant chose to leave for twenty minutes and returned before the end of the session.

The number of questions in the session was significantly less than the usual number of questions that would be set in a focus group (Krueger & Casey, 2009), only three questions were covered:

- 1) What good experiences have you had in hospital?
- 2) What bad experiences have you had in hospital?
- 3) What information would you like to share with medical staff.

This was for two reasons: to reduce the cognitive demands on participants and to allow participants longer to prepare their answers.

The moderator did not detect problems or impatience from the respondents and in general the participants did not interrupt one another. The moderator was only required once to stop a participant from talking until the other had given their reply.



Figure 11 – Participants During Debriefing

6.5.5. Debriefing of Participants

Frequently in focus groups, a written summary or even the full transcript will be provided to participants after the focus group session (Hemsley et al., 2007, 2008b). Given that the participants had varying levels of literacy (ranging from no functional literacy to reading age 16), the feedback was provided orally at the end of the session by the moderator's assistant (see Figure 11).

6.5.6. Results of "Potential for User Centred Design with Adults with SSPI" An examination of the transcript was conducted using coding (see Appendix C) and was verified by an independent researcher.

Participants were initially hesitant in answering questions and only gave short answers in reply to a direct question. After the first eighteen minutes participants began to directly engage with each other with only minimum input from the moderator to move onto the next question or to bring the conversation back on topic.

Author: can you think of anything else?

Danni types: to talk to me more

Doug: would you like them to have a chat with you

Danni types: yeah

Robert: talk to the organ grinder not to the monkey

Patricia: they always talk to someone else and never get time to talk to you.

Robert: I find it so frustrating when they don't talk to me on the ward and they ask someone else and I think for god's sake ask me. It's so frustrating

Lisa: I thought you were going to say something else Robert

Author: so that really annoys you when they don't speak to you?

Patricia: yeah

The time spent on each question increased during the session from an initial three minutes on the first question to a peak of twenty-six minutes on the third question.

This transcript shows that 47% (n = 43) of the utterances by participants were unprompted and that all participants responded unprompted at least once (see Figure 12). For two participants, the number of unprompted responses outnumbered their prompted responses.



Figure 12 - Responses of Participants (Prior et al., Accepted)

The other interesting result from the transcript was that the three participants who responded the most were the three with the most severe communication impairment. The feedback section of the focus group was the most animated of the hour with participants agreeing both vocally and with nodding of the head that the account was accurate. The summary section of the session finished with one participant saying:

"One last thing, thank you for listening".

Kitzinger (1994) suggests that one of the defining features of a focus group which differentiates it from a group interview is the examination of the interaction between participants. During the focus group session, the participants with SSPI were able to interact with each other and many of the ideas that they generated emerged as a result of discussion with and suggestions by other participants. Often one participant would share a story relating to an experience and this would prompt another to think of a similar story or offer suggestions on how a piece of technology could have helped them there:

Doug: They don't know how I drink

Robert: see I was the same see when I first came out of the coma, I asked for a drink so they gave me my liquids with a straw but I didn't have the muscles in my throat to suck so eventually I said "here" the doctors they got me a drink through a syringe cause I could not suck. But it took a long time to get a drink.

6.5.7. Results of "Adaptation to Methods in Enabling Adults with SSPI to be Participants"

The aim of the Focus Group was to understand the context of use, in particular the issues surrounding the need for the software. Participants were asked three questions related to their own experiences of sharing information with medical staff. The first question related to good experiences. Only one answer was provided, two of the participants felt that the food in the hospital was good. The next question asked

participants about negative experiences, participants related their personal experiences to this theme and shared stories about AAC devices being removed and lack of communication between the doctor and the patient. One example of this came when participants discussed the way they perceived the doctors thought about them:

Doug: The doctors speak to you as if you have a mental problem

Staff Liaison: do you mean an intellectual problem?

Doug nods

Robert: I can relate to that

Author: Has that happened to you to?

Robert: I think what Doug is trying to say is that they think you have aren't smart

Doug nods

Staff Liaison: Do you think people look at you Robert?

Danni makes noise to get attention.

Danni types.

Danni types: looked at step mum .

Robert: they think ah he doesn't have a mind of his own

The third and final question related to the current information sharing tasks that participants struggled with in hospital. Participants stated that they struggled to share information on anything in hospital but following discussion they identified six key areas: mobility, personal care, food and drink, getting staff's attention, having AAC devices removed and assumptions on their cognitive ability. In all of these areas the discussion went into detail of the problems and provided useful scenarios and vignettes which could then be used in the requirements gathering stage (see Section 6.7).

6.6 Review of Establishing Context of Use

During the establishment of the context of use the main aim was to learn more about adults with SSPI and the care environment. The ethnographic process was used for this and allowed the author to become embedded in the environment, not only allowing a better understanding of the way the environment operated but also preparing staff and centre users for her continued presence during the crucial requirements gathering and development phases.

The focus group was used to gather more information on the hospital experiences of adults with SSPI, to help the author understand the personal stories behind the rationale for the development of a multimedia patient profile and also allowed the author to spend a concentrated amount of time with the participants to get to know them better. The next stage was to establish the requirements for the software.

6.7 Gathering Requirements Through Forum Theatre

The aim of requirements gathering was to understand the needs of the end users, other stakeholders and the environment. The decision had been made to use Forum Theatre and Storyboarding techniques in the gathering of requirements. These techniques have been shown in the past to be of particular use when the technology being designed is completely new or when participants providing requirements have traditionally not been involved in research (Rice et al., 2007).

As discussed in Chapter Five, forum theatre was developed by Boal in Brazil and is described in this book "Theatre of the Oppressed" (Boal, 2000). It will typically have some form of controversy and stop at the critical point to encourage audience members to participate. It's use as a requirements gathering method in computer design is increasingly being recognised (Newell et al., 2006a).

Software designers who have experience in using forum theatre as a means of requirements gathering believe the advantages from forum theatre lie in its ability to facilitate discussion of sensitive topics with potential users and cover several potential uses for a system in one cohesive storyline (Carmichael et al., 2005). Forum theatre also allows audience participants to suspend disbelief and imagine how a piece of technology that may not be in existence at present could be used in the future (Carmichael et al., 2008). Forum theatre has been used in video format for software design in the UTOPIA video trilogy (Eisma, et al., 2004).

The use of forum theatre video with the participants with SSPI aimed to gather requirements on the types of information participants would want stored on the CHAMPION system. This information would be based around the problems adults with SSPI face when admitted to hospital.

6.7.1. Creation of Material for Forum Theatre

Stories from the Focus Group described in Section 6.5 were examined alongside reports in the literature on problems surrounding hospitalisation for adults with SSPI. The author created two possible scenarios (see Figure 13 and Figure 14). The scenarios were then given to a professional script writer and director with experience

in forum theatre. The video was produced using two actors with SSPI and two professional actors. Full details on the production were presented at the Biennial ISAAC Conference, the paper on this is provided in Appendix D.



Figure 13 - Clip from First Play in Forum Theatre

6.7.2. Method

Following the creation of the video, a session was held with three participants to use

the video in the gathering of requirements.



Figure 14 - Clip from Second Play in Forum Theatre

Three participants who had taken part in the focus group and were available to participate in requirements design decisions were invited to consider joining the group (see Table 5). The requirements gathering session took place in an activity room at the resource centre, with participants sitting around a table alongside the author and staff liaison from the centre.

Firstly the video was shown on a large television screen and paused at the tension point. The participants were invited to share their thoughts and at the end of the video a discussion was held thinking about what might have changed if the doctors had known more about the patient.

Participant	Method of	Other Means of	Established Group
	AAC	Communication	Membership
Danni (29)	Lightwriter AAC Device	Vocalisation	Lives at resource centre
Robert (58)	Gestures	Dysarthric Speech	Attends resource centre three days a week
Doug (53)	Dynavox AAC Device	Dysarthric Speech	Attends resource centre four days a week

 Table 5 - Demographics of Participants (Pseudonyms used)

6.7.3. Potential for User Centred Design with Adults with SSPI

The use of forum theatre in the development of innovative technology is not new, however the literature does not report any previous work using adults with SSPI as actors in forum theatre or as participants (Prior et al., 2010).

The participants were able to follow the scenarios on the video, following the screening a short discussion was held on what they had seen. Robert felt he could empathise with the characters in the video "*I can relate to the situation*". Participants

became engaged in discussion surrounding the video, one example being how doctors would be able to see the information.

Doug: See if they had the information when they started

Robert: They need to have it at the start

Danni: By my bed have it all the time

Doug: I like that idea. Have it next to me, big flashing light saying pay attention.

Participants were also able to complete one of the tasks in traditional forum theatre, imagining alternative endings. By imagining the alternative endings the participants were able to then work through the steps that would have had to occur to reach this ending. At the end of this method the main stages in the information sharing process were outlined by participants for use in the next method.

6.7.4. Potential for Adapted Methods to Allow Adults with SSPI to be Participants

As stated previously the aim of this method was to understand the way participants would want to share information with medical professionals. The three participants identified four requirements of significance related to the information on the CHAMPION system. The first emerged as a direct result from the forum theatre video, participants wondered at what stage during their admission to hospital the medical staff would be aware they had information on the CHAMPION system and how this information could be conveyed to the admission staff.

Participants were also clear that one of their major concerns was that the doctors could see how they were when they were well. Doug related a story of a doctor thinking his illness was affecting his arms as he was not able to carry out a request involving his left arm. Doug stated that he can never use his left arm and struggled to get a doctor to understand this.

The participants also felt that some information would be conveyed better by video than by text:

Danni: I like writing

Doug: sometimes it's quicker to watch a video

Robert: Yeah it'd be good to have a mix of writing and video. Means we could choose.

The participants agreed that for some information e.g. toileting, a written summary of their needs would be best, but for other situations, particularly descriptions of how they act normally a video might be more appropriate. These requirements regarding the information were used in the development stages (see Section 6.11).

6.8 Storyboards with Adults with SSPI

At the end of the forum theatre session the main stages in the patient profile software information sharing process were outlined, these stages were turned into seven cartoons.

6.8.1. Methods

The cartoons showed a character creating information, logging onto the system, saving their information and then viewing it on the system. At no stage did the cartoon show the actual interface or how the cartoon user interacted with the computer (see Figure 16).

The participants were shown each frame and asked to think about what they would do on the computer at that stage:

Author shows frame 3

Author: So this scene is thinking about how the user will work with the computer and what equipment they use to do this. What sorts of extra equipment do you think you and other folk in the centre would need when working with a computer?

After each scene was shown and introduced, the participants would discuss the help they would need to complete the task, how they would expect the computer to respond and what they would expect to do at this stage.



Figure 15 - Scene with Requirements Written On

Their ideas along with additional requirements were recorded on the scene paper. This allowed participants to have a visual representation of their ideas (see Figure 15).

6.8.2. Potential for User Centred Design with Adults with SSPI

The participants were initially reluctant in offering many suggestions, but after some encouragement where the author would address them directly or try to rephrase the question, the participants engaged in a lively discussion. Throughout the session participants showed the ability to compromise. An example of this can be seen when the participants were asked if they would prefer the information categories to be made for them or if they would prefer to make them themselves. Immediately Robert said he would prefer to make them himself

"a person like me, I know I'll take the easy option, but it's better to make me work it out".

Danni did not agree with this and felt she would need help with this. The question was asked again to Doug who could see the benefits and advantages of both but could not decide between them, Robert conceded that he could see that others like Danni "*need a lead to get you going*". Doug replied wondering if the computer could make some guesses on what a user would need and suggested that users could answer a series of questions which would then provide the set of topics. This was welcomed by all participants.Participants were also able to show the ability to be imaginative in providing requirements, this can often be a problem when creating new technology (Carmichael et al., 2008). A good example of this ability is shown when participants were asked how they would want the system to know that it was them using the software and not an imposter.

Danni moves her finger up and down

Author: Do you mean you want to use your finger Danni?

89

Danni nods

Author: Like a finger swipe?

Danni nods

Doug: An eye scanner

Staff Facilitator: Ah you see our Doug is a sci-fi fan!

Doug (laughing): In my ear

Author: I'm not quite sure I understand Doug, can you tell me more?

Doug: Scan through my ear, put a probe in and check my head!

Robert: Yeah, but Doug this ain't Star Trek!

After this very imaginative suggestion the group appeared confident in thinking of other original ideas such as a key or a pendant that could plug into the computer. The group eventually all agreed that a barcode card or swipe card would be the best plan.

The participants showed they were able to understand the idea of imagining what the software might do and good at giving a list of detailed requirements. Participants could think of imaginative and creative solutions and also consider what others might like to see in the system:

"I would be ok with that but other people might need help".

Differences of opinions were handled by the participants through discussions and compromises were reached.









Figure 16 – Storyboard Featuring Scenes in Using System

6.8.3. Potential for Adapted Methods to Allow Adults with SSPI to be Participants

As a result of the storyboard session seven requirements emerged, all of these requirements came directly from participants. The first requirement was to have a questionnaire which would select the items of information that a user would be prompted to provide. Participants stated that they would not find all of the categories of information useful and they would need some way of selecting the information items relevant to them. The questionnaire would appear when a user first registered with the system and users would be asked to repeat the questionnaire every six months to ensure the details were up to date. Participants were clear that the questionnaire should take as short a time as possible to complete.

The information was to be available in video, text or photograph form, a user would be able to use any combination of the information they wanted.

As stated previously one requirement which was the subject of a great deal of discussion was how a user would log onto the system. The requirement was that the user should be able to log in using an identity card with a barcode strip.

Participants requested the ability to look at, change and add information. It was also a requirement that no log should be made of the changes that had been made, participants were clear they did not want to have to look at the way their information had changed.

The software was required to be compatible with a variety of PC peripherals including mouse, switches and touch screen. The participants were clear that switch access was an essential requirement.

The final requirement emerging from this stage was that the user should be able to select which members of staff were allowed to see the information and that the staff should be sub divided into five categories: doctor, student doctor, nurse, student nurse, other staff.
6.9 **Review of Requirements Gathering Stage**

As stated, during the requirements gathering the main aim was to understand the needs of the end users, other stakeholders and the environment. As this form of software had not existed before, and the end users had traditionally been excluded from research, it was decided to use forum theatre as this has provide useful in similar situations. Storyboarding was seen to be a good complement to this method.

The forum theatre was based upon stories from the previous stage's focus group and depicted problems that can occur in hospitals when information is not shared appropriately. This then fed into the storyboard which presented an alternative ending to the theatre's story when the CHAMPION software was available. The storyboards showed each of the stages the end user would work through in creating their own patient profile. The participants provided a well structure list of requirements (see Appendix I) and this allowed the author to begin preparing for the next stage of producing design solutions to the problem.

6.10 Produce Design Solutions with Adults with SSPI

The production of a design solution will usually begin with a prototype (Hakim and Spitzer, 2000). Prototyping originated in mechanical engineering and design when a model of a new device would be created, often smaller than the final version and with poorer quality materials (Sharp et al., 2007). The aim of a prototype would often be to create a proof-of-concept and to evaluate the potential for the end device to work.

In the early 1990s there was an attempt to document how prototyping should be done in software engineering, and establish guidelines for this (Nielsen, 1992). Now prototyping is an accepted process in software design and is included in the ISO standards for user centred design (ISO, 2010).

A full discussion on the types of prototyping is provided in Chapter Five.

6.11 Low Fidelity Prototyping with Adults with SSPI

There was an awareness that users unused to critiquing software could be reluctant to provide critical feedback and that they would need to be 'coached' in how to do this (Eisma et al., 2004). In order to assist with this it was decided that the prototypes should start with low-fidelity prototypes. Low-fidelity prototypes have very limited functionality and are typically produced using paper sketches (Rudd et al., 1996) (see Chapter Five).

By beginning in this way it was hoped that participants would gain confidence and experience in providing constructive criticism of the software by the time it reached the later more challenging stages.

6.11.1. Method

Paper prototypes are usually sketches by the designer or participants with which a participant can 'press' a paper button and then be shown what would happen from that (Rudd et al., 1996). The problem anticipated with using this method with participants with SSPI was that the participants would not be able to easily draw a sketch or reposition small paper items. Due to participants' problems in fine motor control and visual impairments the author felt that the participants would need something more tactile and colourful to engage and to help with any sensory impairment.

The other requirement for the paper prototyping was that the objects in the process were large enough to be moved around on the prototype screen. Large pieces of foam were cut to shapes representing objects on the screen and magnets attached to the back of these in order to facilitate this (see Figure 17).



Figure 17 - Magnetic Paper Prototyping

This allowed the author to place objects on a screen sized magnetic white board and present the screen to participants at an angle in front of their wheelchair (see Figure 18).



Figure 18 - Participants Using Magnetic Prototypes

Participants were then asked questions such as "*How do you think you would let the computer know you wanted to save a picture for your information?*" or "*Can you show me how you would do that?*". These questions are used in traditional HCI and were aimed at uncovering how a user would be able to interact with a computer if the prototype was turned into a fully working piece of software.

6.11.2. Potential for User Centred Design with Adults with SSPI

Participants commented on what they liked and did not like about the screen and were encouraged to move the objects to positions that made more sense, ask for the text on the object to be reworded or to ask for the size of the object to be changed.

The participants reported enjoying this session and were able to offer negative and positive feedback, and suggest features that should be changed and features that should be added.

As in the previous session the participants showed the ability to compromise and think of creative solutions to problems. One new ability demonstrated by participants was that of helping each other. At one point Danni had stated that she did not like the text on the button (e.g. "Browse For File") and could not work out what it meant.

Robert: I think it means you need to pick where the file is stored on your computer.

Danni: I don't understand

Robert: You know when you put something on a computer? That means it wants to know where it is

Danni: Ok

Doug: It should say Find The File then!

Following the session in discussion with the author, the study co-ordinator highlighted this as the continuing personal development of participants. This move to help each other solve problems was new to all of the participants.

6.11.3. Potential for Adapted Methods to Allow Adults with SSPI to be Participants

The main issues were that although users might wish to use a touch screen, they were not all able to use a finger to point and used the side of their hand, this meant buttons, lists and text boxes all had to be bigger. It also showed that the text on buttons was often too technical in language, and that two of the participants would need the text read out to them. In this section, four changes were suggested (see

Table 6). These changes included two functional requirements (requirements related to how the software operated) and two design requirements (requirements related to how the software looked). All of the requirements came directly from participants.

The two functional requirements related to the accessibility of the software. When Doug was using the paper prototype he required the text to be read out to him, as although Doug had some literacy, his visual impairment meant he could not focus on the text. Doug realised this and asked the author if it was possible for a computer to read objects out to him in the same way. The requirement for voice synthesis was added to the list of requirements.

While the author was doing manual scanning by pointing to the objects, Robert commented that he had been trying eye gaze and he had enjoyed it. He put in a request for eye gaze in the system, stating that he thought it would help other users as well.

As stated previously Danni had struggled with the language used on one of the buttons, this was a common problem. Often technical terms such as "log on" or "browse for file" were used. Although the participants had used computers before, they were often assisted by a support worker when doing this. This meant that when more technical terms were used the support worker would carry out the task. The requirement for less technical terms was added.

The author had used pictures to aid the text on the buttons but it emerged that that participants would have preferred the addition of a symbolic language rather than general pictures. This requirement was to be implemented as soon as possible.

or

Table 6 – Alterations to Design from Paper Prototyping

6.12 Mid Fidelity Prototyping with Adults with SSPI

Mid fidelity allows software to have the appearance of a working program while still being easy to alter and rearrange without needing to code. It is used after the early initial design to look at detailed design and usability validation (Engelberg and Seffa, 2002).

6.12.1. Method

The main tool used in general design for mid fidelity prototyping is PowerPoint (Engelberg and Seffa, 2002)

With PowerPoint, designers can put in animation to give the impression of a task being conducted, this does not require programming skills (Hakim and Spitzer, 2000).

PowerPoint does have disadvantages; it is unable to support full interactivity (many tools are limited to hyper linking) and as these links are static the functionality is limited (Hakim and Spitzer, 2000).

When the author investigated the use of PowerPoint in this stage of the process for participants with SSPI, additional problems emerged. In PowerPoint including keyboard shortcuts is a very complex task, reducing the potential for incorporating switch access into the animation.

In PowerPoint the scanning would also have to be hardcoded in which requires considerable effort. It is also difficult to have voice synthesis for specific buttons.

In consultation with staff at the care centre and other assistive technology researchers it was suggested that the author could use Clicker5 (Cricksoft, 2011), a piece of software originally designed for use in education to help children with disabilities with reading and writing. It emerged that Clicker5 could be used for mid fidelity prototypes in a similar manner to PowerPoint. It has several accessibility advantages; it has a high contrast colour option (for those with visual impairment) and is fully switch accessible. It also has voice synthesis options and can read the text on objects in the Clicker5 screen when they become selected.



Figure 19 - Peripherals

The photographs from the lo-fidelity screen plans were used to guide and direct the design of the interactive mid-fidelity prototype.

For this design session the participants wanted to use their own preferred access methods (e.g. mouse, touch screen or switch). A large 37 inch touch screen monitor was set up with the options of keyboard and mouse, touch support or switches (see Figure 19). Due to these different access methods, three individual sessions were held when the participant had sole use of the system. A group session was held following these to discuss what they thought about designs.

6.12.2. Potential for User Centred Design with Adults with SSPI

In this method the participants demonstrated the ability to think more critically about designs and were able to describe in more detail why something did not work.

e.g. **Robert**: See this bit, it doesn't make sense

Author: Can you explain that a bit more?

Robert: Well see the button, when it just says quit I think that means quit this bit but the whole thing closes

As this example shows, participants were able to go into more detail about why they did not like certain aspects of the design, or what parts of the design were confusing them.

6.12.3. Potential for Adapted Methods to Allow Adults with SSPI to be Participants

This session revealed several issues simply by observing the participants and their use of the systems. Firstly, as Danni (the touch screen participant) had to use the side of her hand to select an object rather than her finger due to reduced dexterity, not only did the buttons need to be larger, they also needed to be placed away from the edges of the screen to prevent other windows being opened or closed (see Figure 20).



Figure 20 - Participant Using Mid Fidelity Prototype

The other issue observed was that it was necessary for the scanning to be switched on and off. The scanning was a distraction and was not required when the touch screen or mouse was being used.

In the discussions Doug, who used switch access requested that different "*types of scanning*" were used, previously only linear scanning had been available.

During the discussion participants confirmed the observation that they were able to cope with having the different types of information hidden behind category headings rather than having all information on the screen.

Name of Requirement	Functional	Design	Suggested by	Suggested
	Requirement	Requirement	Participant	By Author
Alter voice synthesis	*		*	
Get rid of back button in questionnaire	*			*
Just press yes/no to move through questions	*		*	
Use AAC device as keyboard	*		*	
Change wording on questions		*	*	
Stop it interfering with other buttons on screen		*		*
Make buttons large and centred		*		*
Have symbols on buttons		*	*	

Table 7 - Alterations to Design from Mid Fidelity Session

Following this session, a further eight changes were made before moving onto a high level prototype. These changes included four design requirements and four functional requirements. Three of the functional requirements came directly from participants and two of the design requirements came from participants. The other requirements arose after observations by the author and were approved by the participants during discussions. Details of the requirement changes and additions are contained in Table 7.

6.13 High Fidelity Prototypes First Iteration with Adults with SSPI

Following the "throwaway" mid-fidelity prototyping the next stage is usually the development of a evolutionary high level prototyping (Isensee and Rudd, 1996). High level prototypes will looked like a finished piece of software but will have less functionality (Rudd et al., 1996).

Marc Rettig (1994) argues that more projects should use the low-fidelity prototyping because high fidelity prototypes take too long to build and therefore developers are reluctant to change something they have worked so long on.

The CHAMPION project was careful to avoid the problems described by initially concentrating on the interface of the software. The software was run in Visual Studio¹² as this allowed for quick bug fixing. The software had limited functionality to begin with to help reduce the time required when changing features. Participants were encouraged to think about the ease of use as well as the appearance.

6.13.1. Method

The three participants from the prototyping sessions each had individual sessions lasting one hour using their preferred access method and then again came back together to discuss their views.

¹² Visual Studio is an Integrated Development Environment, a software application that allows software developers to program their own pieces of software.

6.13.2. Potential for User Centred Design with Adults with SSPI

This time it was clear that participants were more confident in giving their opinions in the feedback. Doug who previously had been the least forthcoming in his opinions was particularly notable as an example of improved confidence.

Doug was set up with switch access in front of the screen and asked to explore the software. Doug was very adept at scanning and quickly moved through the different areas of the software. After two minutes he hit the exit button. The author and support worker assumed that he had selected this button by mistake and offered him the option to start again.

Doug replied that he did not want to and when asked if he wanted to elaborate on this replied: "*it's easy to use, but it's for kids*".

The author probed this further and discovered that the reason Doug felt this way was due to the cartoon like graphics and "Comic Sans MS" font.

The author then worked with Doug to look at other possible fonts and images which were later approved at the group design sessions.

6.13.3. Potential for Adapted Methods to Allow Adults with SSPI to be Participants

Other issues raised by participants included a dislike of the topic titles either being too complex e.g. "Mobility Assistance" (later replaced with "How I move"), or too direct e.g. "Toileting" (later replaced with "Personal Care").

By using Visual Studio to run the software, the alterations could be implemented instantly and the participants could then discuss if it makes the software better or worse. Doing this often revealed new problems that could occur as a result of the change so that solutions for the new problem could then be discussed (see Table 8).

Name of Requirement	Functional Requirement	Design Requirement	Suggested by Participant	Suggested By Author
Swap between symbols	*		*	
Change scanning speed	*		*	
Use less "cartoon" like drawings		*	*	
Allow changes on staff who can see information	*		*	
Make all buttons the same size and improve consistency in the software		*		*

Table 8 - Alterations to Design from High Fidelity Sessions First Iteration

Review of the First Iteration

Two problems with high level prototyping emerged in this iteration. The first immediate problem was the time taken at the start of each session to set up the different access options for participants. Connecting the device, configuring it for use with the system and testing the device took time away from the session with the participant and forced them to wait while this was sorted. It would have been useful to have set aside 20 minutes at the beginning of the day to ensure this was all working.

During this session it also became apparent that if the participants were tired it was best to abandon the session. On one day, two out of the three participants were tired so it was decided to rearrange for another day and participants were reassured that this was not a problem. One participant was feeling up to a general chat and this was an opportunity for a more relaxed conversation about the technology. It is important to build extra time into a project plan to allow for this. It is already common to build time through risk management analysis into a project. This time should be increased when planning for work with people with SSPI. It is estimated that children with cerebral palsy (a major cause of SSPI (Larsson et al., 2009)) are 4.3 times more likely to be admitted to hospital as a result of illness while for adults the figure is 10.6 times than non-disabled peers (Young et al., 2007).

It also became clear that participants were not put off giving criticism of the designs despite a more "polished" appearance as shown by the previously described example with Doug.

6.14 High Fidelity Prototyping Second Iteration with Adults with SSPI

For the second iteration extra time was allocated to set up the different peripherals that the participants would use.

6.14.1. Method

Functionality was added to features in the software, which had already been approved in the first iteration. There was still limited functionality for the portions of the interface that had not been agreed on.

The participants made a few comments on the updated designs but were overall happy with the software. At this point it was decided to bring a fourth participant (Gavin) into the project (see Table 9).

Gavin had been aware of the project from the beginning but had been unable to participate due to conflicting commitments. When the original three participants felt they would like a second opinion on their choices it seemed an obvious choice to ask Gavin to join.

Participant	Method of	Computer Access	Established Group
	Communication	Method	Membership
Gavin (32)	AAC Device	Touch Screen Wobble Stick	Attends resource centre four days a week

Table 9 - Demographics For Gavin

6.14.2. Potential for User Centred Design with Adults with SSPI

During this stage the participants became aware of the fact that there was little they were able to suggest to improve the design of the software. It was the participants who recognised that this might be due to their experience with the system and they suggested recruiting a new participant to assist.

6.14.3. Potential for Adapted Methods to Allow Adults with SSPI to be Participants

Gavin was on the whole pleased with the designs but did request more instructions for signing up and changing information. The other participants had not noticed this lack of information as it related to features they had seen throughout the different iterations (see Table 10).

6.14.4. Review of High Fidelity Prototyping

Following this session participants discussed how they could involve more people from the care centre in the evaluation of the software. Robert suggested setting up the large computer in the social area of the centre and having the software running. He felt the other members of the centre were "*nosey so and so*" and would likely come over to see what was happening. Danni added that they could have a poster up so people knew what it was. The participants offered to organise this event with the agreement that the author would set up the technology on the agreed date.

Name of Requirement	Functional Requirement	Design Requirement	Suggested by Participant	Suggested By Author
	-		-	
Change voice synthesiser	*		*	
used				
Change wording on		*	*	
some buttons				
Allow user to choose	*		*	
whether instructions				
should be read out				
Provide Information on		*	*	
Registration				

Table	10 -	Alte	erations	to]	Design	from	High	Fidelity	Sessions	Second	Iteration
								/			

6.14.5. Review of Prototyping

The producing of design solutions aimed to involve the end users in developing a design solution from basic low fidelity prototypes into a high level functioning prototype. Many of the traditional methods used in prototyping were not suitable for this user group and so adaptations had to be created. The adapted methods included magnetic foam prototypes and Clicker5 interactive prototypes. While using Visual Studio for high level prototyping is not new (Rudd et al., 1996), additional care had to be taken to make sure the sessions were not too complex for the user group.

This stage showed that the participants were able to progress from low to high level prototypes and could provide detailed feedback and constructive criticism at all levels. Towards the end of this stage a new participant joined the group to provide a fresh outlook. The participants agreed to help organise the final evaluation stage with new participants.

6.15 Software Evaluation with Adults with SSPI

The aim of evaluation is to collect information about the user's experience when working with the software (Boehm et al., 1976).

The evaluations were to make use of semi structured interviews, observation and diaries. All of these methods would be used while the participants were carrying out tasks of their own choosing on the system.

6.15.1. Method

By this stage in the development the author was firmly ingrained into the environment of the centre and was able to work in it without being a major distraction to staff, this was evidenced in her participation in activity sessions. Participant-as-Observer observation would allow the author to watch the new participants using the software for an initial three days before leaving them with the diaries for a longer term evaluation (see Chapter Five).



Figure 21 - Poster For CHAMPION Evaluation

The staff at the centre arranged for posters (see Figure 21) to be placed in the main meeting room around the large touch screen monitor.

A wide range of different peripherals (including switches and joysticks) were set up with the computer so that participants could come up and use the device in their own time. A user guide on how to load the software, extra participant information and consent forms were left by the monitor as well.

Four new participants (see Table 11) were taken through the informed consent process and were given an introduction to the project and provided with their own log in cards. Each participant spent half an hour with the author setting up their account and exploring the software. The author gave minimal instructions (e.g. "Can you try and add some information about how you eat?)" and only gave assistance when requested. All requests for assistance were noted for inclusion in a user guide for the final version.

During these initial sessions, tasks carried out on the software were limited. Participants were provided with a set of tasks to carry out in the longitudinal study. Participants were also shown how to complete the diaries during the longitudinal phase (see Appendix J). After the initial session each participant was interviewed individually on their initial impressions of CHAMPION. This was aimed at discovering its initial usability and questions centred on how easily participants felt they would manage on their own and problems they anticipated in using the system.

6.15.2. Potential for User Centred Design with Adults with SSPI

Midway through the evaluation period the staff at the centre were having difficulty in accessing the software. They contacted the author to ask for advice and the author attempted to rectify this by taking remote control of the software. However, the

computer did not appear to be capable of connecting to the internet. The author then went to visit the centre to try and solve the problem. On arrival it was apparent that the internet was not working. None of the staff knew where the modem was situated and a search took place to locate it. Once it was located, the problem was identified easily as the modem had been unplugged. Once this was corrected, a second briefing session was held for staff.

Participant	Method of	Computer Access	Established Group
	Communication	Method	Membership
Paul (21)	Eye Gaze	Eye blink glasses	Attends resource centre
			four days a week
Patricia	Signing, Dysarthric	Switch	Attends resource centre
(44)	Speech		five days a week
Lisa (40)	Facial Gestures, Dysarthic	Switch	Attends resource centre
	Speech		four days a week
George (35)	Dysarthic Speech, Word	Switch, Keyboard	Attends resource centre
	Board		two days a week

Table 11 - Participants in Software Evaluation

At the end of the two months the author returned. Only limited sessions had been held as participants had experienced problems in finding uninterrupted space to use the system. Staff had also experienced difficulties in finding free time in their busy schedules to assist or encourage participants to use the software.

The diaries had only been used once and so the planned semi structured interviews were not possible. A short discussion was held with participants based on the limited sessions that had occurred. Participants stated that they enjoyed using the system and would like to use it again.

6.15.3. Potential for Adapted Methods to Allow Adults with SSPI to be Participants Four short term evaluations were conducted by the author during this two days. The results were very positive, participants pointed out three minor problems related to the layout of the program. A discussion was held related to these issues and changes were made.

Following the three recruitment days and with four new participants recruited the system was left at the centre for two months along with a diary for staff to write in with participants. This was less successful and suggests that having a researcher onsite at regular intervals during the evaluation period would be beneficial.

6.15.4. Review of Software Evaluation Process

The evaluation stage was divided into two sections, firstly a short term evaluation designed to recruit new participants and get their immediate feedback. This took place over three days at the care centre and recruited three new participants who conducted usability evaluations and were able to provide feedback, despite the fact that they had not been trained in evaluations as the original participants had. A longer evaluation session was then attempted but due to problems at the centre and the equipment there, limited sessions had been conducted. Despite this the short term evaluation demonstrated that participants with SSPI could quickly take part in usability evaluations with limited training. Four participants were able to complete tasks such as adding information, looking at information and changing information. These tasks were completed fully by participants who had only had thirty minutes initial training.

6.16 Evaluation of Participant's Experience of Process

6.16.1. Participants' Views of Process

Following the conclusion of the evaluation process Danni, Doug and Robert (the original three participants) and the staff liaison met with the author to look back over their experience of being involved in the project.

The participants were asked eight questions and the answers recorded by the staff member on an A1 sized flip chart using graphic note taking to help engage participants in the process (see Figure 22).

Author: What did you think about being involved in making CHAMPION?

Doug: It's the way you should do it

Robert nods energetically

Robert: "if we can see the goal of the project its worth doing",

Danni indicates she wishes to speak

Danni: I have enjoyed all of it

Author: Anything else you want to say about it? Ok so what was your favourite part of the project?

Danni: Designing stuff was really good

Doug: Drama was best

Author: Anything else?

Doug: being listened to

The questions had been read by other researchers ahead of the session to ensure they were not leading.

This discussion had provided some interesting answers. Danni who had previously only shown an interest in drama at the centre had enjoyed designing more than doing drama while Doug who usually did not enjoy being the focus of attention had enjoyed the drama most. Poignantly Doug also said that the thing he had enjoyed the most was being listened to.

This was a theme that had previously been seen in the focus group sessions (see Section 6.5).

Participants were encouraged to think about the negative parts of the process and were told that this information would be useful and help make the process better next time. The participants said that they could not think of any negative aspects. This could have been because they did not wish to upset the author but in the past the participants had been keen to provide negative feedback. The question was then rephrased to ask them if there was anything they would like to have done differently, Doug would have preferred a different room and Robert would have liked to have been able to work in an area with fewer interruptions and had more access to different types of equipment.



Figure 22 - Graphic Review of Participant Debrief

The discussion then turned to how participants felt when taking part in the research (see Figure 23).

Author: Now, thinking back over everything. I wonder how you guys have felt while doing the project?

Danni: Active

Author: Can you tell me more about that?

Danni: I liked being active

Robert: It's about inclusion. Inclusion is what we want.

Doug nods

Doug: It felt good. Being involved.



Figure 23 - Feelings when Taking Part

The author then asked the participants to sum up the project in three words or less (see Figure 24), they replied with "*inclusion*", "*possibility*" and "*it*'s *the best*". The discussion ended with participants being asked if there was anything else to note and Doug replied with "*what we do matters*".

The use of focus groups with these participants had already been shown to be effective in getting useful insights and new information. This focus group was of a different form to previous sessions, it was more relaxed and participants were being asked to undertake personal reflection as opposed to relating stories from past events (see Section 6.5) or evaluating technology (see Section 6.15).



Figure 24 - Summary of Project

There was a danger that participants may have forgotten aspects of the process or were only remembering the positive aspects and had forgotten negative aspects This was mediated to some extent by the author going through all the stages one by one to ask participants' opinions. By having a group interview as opposed to individual interviews, participants were able to remind each other of things they had forgotten. The group discussion itself does create problems to validity in that participants may have been reluctant to reveal stages they did not like for fear of how they would be viewed by the other participants. This was not felt to be a significant problem, over the course of the work on the project the participants had bonded and had become comfortable in discussing tasks they could not do with each other and expressing personal opinions confidently. The final risk of the credibility of this session is that participants would be uncomfortable in relating to the author any aspects they had not enjoyed of the process. Through the process the author had become regarded as a member of the group and the participants may have been wary of offending her. In order to minimise this risk the author attempted to portray the analysis of the process in the same way that the analysis of the prototypes and software had been conducted. Participants were informed that by revealing negative aspects they would be assisting in making the process better, had there been any major negative aspects of the study the participants by this stage would probably have had the confidence to reveal this to the author.

Following the debrief with the participants a separate interview was held with the staff liaison, she also reflected positively on the experience of being involved and offered further insight into changes observed by staff on the participants' confidence and behaviour since the start of the study.

Robert had always been the most confidant and outspoken of the group and had the least change noted. The only difference that the staff had observed was that he no longer sat alone at lunch because Doug now always insisted on eating with him.

Danni had also previously shown some confidence but this had been noticed as increasing. The other major difference seen was that she was now expressing an interest in doing more projects and getting into possible employment.

The person who had shown the greatest change was Doug who previously had been very isolated and reluctant to enter into conversations with people outside of his immediate social circle. Since taking part in the project he had joined the drama group, was using his AAC device more and was voicing his opinions with staff when they did something he was unhappy with.

These observations would appear to suggest that not only did participants give a large contribution to the design and functionality of the software, the experience was positive for the participants as well.

6.17 Credibility of study

The main danger to the credibility of this portion of the study was the analysis of whether adults with SSPI could take an active role in user centred design. There was a risk that the author would unintentionally make the decisions on the software design or requirements but attribute them to the participants, or provide the answer to questions she had asked and then only ask participants if they agreed. This would make it appear that the participants were able to provide requirements and make decisions on design when they actually had not done this.

The author was conscious of this and did her best to minimise this. At regular intervals the author checked with participants that they meant what the interpretation was. In addition the researcher was careful not to provide suggestions immediately, only if the participants could not think of any solution. All of the sessions with participants were video recorded and these videos were viewed by a second researcher who agreed that this had not occurred.

The sample size was small, under Nielsen's usability guidelines (1993) there were enough participants to evaluate the usability of the study. However, to investigate if people with SSPI can conduct User Centred Design a larger sample size is required and in different centres. Restricting the study to one centre opens it up to the risk that the staff in the centre or the type of participants who chose to use that particular centre could be confounding the result.

There was the risk that the group of participants did not constitute a true reflection of the SSPI population. As far as possible representative sampling was conducted, the final four participants covered the age range of 25-65 (mean=43.75, median = 42.5). Unfortunately it was not possible to get a balanced male to female ratio (3:1). This was due to the fact there were fewer females at the centre who used AAC devices and only the one (Danni) who was available to take part in the sessions and who was able to provide informed consent (the two other females from Section 6.5 were no longer available as participants when the requirements stage began). Despite this it was felt that the sample was representative. The participants had varied home environments: living with family, living in supported accommodation and living in care centres. They also had a variety of educational experience and used a variety of communication aids.

6.18 Limitations to Study

It would have been useful if more methods could have been trialled but it is not clear if this would have uncovered more information from users. It would not be ethical to 'waste' their time using more methods to get the same information back. It is also not clear if this would have been a useful way to evaluate the potential for methods as after the first method in a process the participants could lose motivation or not provide information as they would feel it to be the same as the previous method.

The study could have been improved by conducting two concurrent software development projects with two groups of participants to compare the differences between the groups and to ensure that the one group of participants were not an isolated event. This would have proved difficult due to the extra time this would have required and the extra resources needed.

Given the ways the study could have been improved it was felt that a second study was needed to address the concerns that the results may have been due to the participants and to explore different methods for the stages of the User Centred Design process. This second study would be conducted with a new group of participants from a different centre to ensure that the participants were not a special case and that more people with SSPI could take part in User Centred Design. It was also decided to use a new developer to ensure that this was not a confounding variable.

A set of guidelines were produced (see Chapter Seven) following the CHAMPION study, these were then provided to a second developer in a new software project with adults with SSPI (see Chapter Eight).

6.19 Summary

The CHAMPION study demonstrated that adults with SSPI were able to be actively involved in the development of software. The author learnt many lessons in this project and it was felt that these lessons could be of benefit to others. The lessons were developed into guidelines and these are discussed in Chapter Seven.

Chapter 7. Development of Guidelines

"This is the way you should do it".

Robert, a participant in the study with SSPI

7.1 Introduction

Chapter Six demonstrated the ability of adults with Severe Speech and Physical Impairments (SSPI) to take part in software development tasks. Given the lack of guidance available for developers wishing to work with this group in design, the author made the decision to produce a set of guidelines based upon lessons learnt in Chapter Six.

The way in which the guidelines were collated along with an overview of the guidelines are presented in this chapter.

7.2 Development of Guidelines

Following the completion of the CHAMPION development an analysis was conducted to investigate what lessons had been learnt and if these could be of benefit to other developers. This analysis involved a triangulation of data from video tapes, transcripts and an examination of features in the software.

7.2.1. Video Tape Analysis

All of the sessions with participants had been video recorded, with the exception of the debrief at the end of the project. These videotapes were reviewed by the author, the tapes were analysed for moments in which one of the challenges in the literature had been addressed. Every time a challenge was addressed it was noted, at the end of the analysis all of the notes were examined together to identify what challenges had been addressed and how they had been addressed. The problems that occurred in the different sessions were noted during the watching of the videos and the author analysed why these problems occurred and what if anything could have been done to prevent them.

An independent researcher was asked to watch two hours of randomly selected videos to provide quality control. Ideally all of the video tapes would have been validated by a second researcher but due to time constraints this was not possible.

7.2.2. Transcript Analysis

Transcripts were prepared for every session and these were analysed for contributions to the software and for themes using thematic coding. This analysis was focused on both development in the participant's ability to take part in User Centred Design and in the contributions provided to features and requirements in the software. These were again verified by an independent researcher.

7.2.3. Feature Analysis

The final piece of triangulation analysis was the investigation into the origins of features and requirements. The features and requirements in the software which had been provided by participants were traced back to the session in which they originated. The reasons for the success in achieving this information were then analysed.

7.3 **Preparation of Guidelines**

Using this triangulation of data, the key areas were brought together and used to inform a new set of guidelines. Due to the popularity of the ISO guidelines in User Centred Design the guidelines for working with adults with Severe Speech and Physical Impairments were based upon these. The new guidelines are not intended to be used in isolation for mainstream development and should be used alongside current guidelines on User Centred Design. The guidelines are presented below but are also available as an independent document in Appendix L of this thesis.

7.4 Guidelines for Working With Adults with SSPI in User Centred Development

7.4.1. Introduction

User Centred Design is an approach of interactive system development that focuses specifically on making systems usable. It is a multi-disciplinary activity which incorporates human factors and ergonomics knowledge and techniques. The application of human-factors and ergonomics to interactive systems design enhances effectiveness and efficiency, improves human working conditions, and counteracts possible adverse effects of use on human health, safety and performance. Applying ergonomics to the design of systems involves taking account of human capabilities, skills limitations and needs.

A person with Severe Speech and Physical Impairments (SSPI) will have different problems than a person with a Communication impairment which describes a disability affecting speech and language. A person with a communication impairment may have difficulties including articulation problems, fluency problems, aphasia and delays in speech (Medline Plus, 2009). SSPI encompass more than a communication disorder and include a range of physical, sensory and cognitive impairments (Balandin, 2002).

7.4.2. Working with adults with Severe Speech and Physical Impairments in User Centred Design

This standard provides guidance on conducting User Centred Design with users with Severe Speech and Physical Impairments throughout the lifecycle of the development of a technology. This technology is not limited to assistive technology and could be a piece of technology for mainstream consumption. This standard is concerned with both hardware and software components of interactive systems.

This standard addresses the planning and management of User Centred Design with users with SSPI. It is based upon the guidance in ISO-9241-210 regarding conducting User Centred Design and as such does not address the specifics of the stage of the lifecycle specified in ISO-9241-210 but instead provides guidance on how these stages can be conducted with users with SSPI.

The main users of this standard will be those developers or researchers who will be working directly with users with SSPI. However it is beneficial for the entire development team to be aware of the issues in working with these users and how the project is addressing the issues.

7.4.3. Terms and definitions

For the purposes of this standard, the following terms and definitions apply.

a) Prototype

Representation of all or part of a product or system that, although limited in some way, can be used for evaluation [ISO 9241-210:2010, definition 2.2]

b) Usability

Extent to which a product can be used by specified users to achieve specified goals with effectiveness, efficiency and satisfaction in a specified context of use. [ISO 9241-210:2010, definition 2.3]

c) Effectiveness

Accuracy and completeness with which users achieve specified goals. [ISO 9241-11:1998, definition 3.2]

d) Efficiency

Resources expended in relation to the accuracy and completeness with which users achieve goals. [ISO 9241-11:1998, definition 3.3]

e) Satisfaction

Freedom from discomfort, and positive attitudes to the use of the project [ISO 9241-11:1998, definition 3.4]

f) User

Individual interacting with the system.[ISO 9241-10:1996, definition 2.2]

g) SSPI

Severe Speech and Physical Impairments encompass more than a communication disorder and include a range of physical, sensory and cognitive impairments (Balandin, 2002).

f) AAC

The general term for the methods used to aid communication by those for whom the more usual forms of communication are not sufficient. AAC can be divided into two different categories of unaided and aided. Most people who use AAC will use a combination of the two (American Speech-Language-Hearing Association, 2002).

g) Support Worker

This refers to any person who assists the user in their day to day lives or in using the system. This may be a person employed to assist the user or a family member.

h) Developer

This refers to the HCI practitioner, the software developer or the researcher who is involved in the project.

7.4.4. Rationale for working with users with SSPI

The number of computer users with cognitive and developmental disabilities is increasing. Examples of this include: day and residential units holding computer courses as part of the program for service users (Parsons et al., 2006) and schools for children with disabilities now have ICT lessons in their basic curriculum (Judge, 2001). In addition, as adults with developmental disabilities become more integrated into the workplace, the demand for computer software which is accessible by a wider spectrum of users is likely to increase.

However, the main use of technology by this user group remains in assistive technology and AAC devices. As with other user groups, a lack of User Centred Design may contribute to the poor adoption of technology (e.g., the rate of abandonment of AAC devices is reported to be as high as 53.3% (Riemer-Reiss and Wacker, 2000)). The main reasons given for abandonment

or rejection of assistive technology are a lack of training for the end user and a lack of flexibility in adapting the system for the individual. Other reasons reported include a lack of confidence on the part of the user in their ability to use the system and the prohibitive costs in learning how to use the system. It is suggested that the inclusion of end users in the design process may

reduce this abandonment rate (Waller et al., 2005b).

7.4.5. Structure of this standard

- Clause 5 outlines the problems that can occur with Severe Speech and Physical Impairments.
- Clause 6 gives an overview of the technology that are used to assist people with SSPI.
- Clause 7 discusses the problems that are noted when doing User Centred Design with users with SSPI.
- Clause 8 gives guidance on how to deal with these problems.
- Clause 9 highlights methods known to work at each stage of ISO 9241-210 with users with SSPI.

Clause 5 - Problems for people with SSPI

People with SSPI may have cognitive impairment which can range from mild to severe and may have difficulty in understanding all that is being said in the conversation and the developer's questions.

People who use an AAC device for communication can produce words up to 25 times slower than those with normal verbal speech (Higginbotham et al., 2008b). The time therefore for a participant to reply, for example "Yes, that has happened to me" could take up to 2 minutes for a participant relying on
AAC (Augmentative and Alternative Communication at the University of Washington, 2009). A person with dsyarthric speech may be able to produce words at the same rate as a participant without a SSPI, however the need for clarification of the response can slow down the rate at which the response is understood by the group (Hustad, 2006a).

People with SSPI are likely to become tired more quickly than the general population (Jahnsen et al., 2003a). This is attributed to "post-impairment syndrome", which means that those with SSPI will use between 3 and 5 times as much energy as those without the condition (Wood et al., 2008). Fatigue is also one of the most common and debilitating symptoms associated with Traumatic Brain Injury.

Clause 6 - Technology to assist people with SSPI

Augmentative and Alternative Communication (AAC) is the general term for the methods used to aid communication by those for whom the usual form of communication through speech is not sufficient. AAC can be divided into the two different categories of unaided and aided. Most people who use AAC will use a combination of both (American Speech-Language-Hearing Association, 2002).

In unaided AAC systems additional equipment is not used, for example sign language, Makaton (Communication Matters, 2008) and facial expressions (Communication Matters, 2003). These systems have the benefit that they can be used anywhere and the user does not need to carry additional equipment around. However unaided AAC has the disadvantage that the communication partner needs to understand how the system works. Aided AAC systems refer to a communication methods that involve a device – this may be electronic or paper based (e.g. a word board) – of some form which is external to the user. The device can be used to transmit or to receive messages and generally display symbols which the user selects to convey messages to listeners, for example a symbol board (Beukelman and Mirenda, 1992). A user can select a symbol, picture or in the case of literate users a selection of letters to build up a message. The main advantage of an aided device is its ability to offer potential for communication to people with SSPI for whom unaided systems require refined motor skills or cognitive load not within the user's capability (Hampson, 2006). Electronic devices also offer the potential for synthesised speech output in the case of speech generating devices.

Other assistive technology for this group may include electric scooters or wheelchairs to assist with mobility, personal organisers or reminders to assist with cognitive difficulties, screen readers, magnifiers and hearing aids to help with sensory problems and environmental controls to help them to live as independently as possible.

Clause 7 - Problems that can occur when doing User Centred Design with users with SSPI

- a) Can a group of 'representative users' be gathered?
 - gathering a representative sample of different disabilities and needs is more challenging
- b) How can different accessibility needs be met?

- c) Can informed consent be gathered from participants, and if not is it possible to gain assent from the participant and informed consent from a legal guardian.
 - It is important to ensure that participants are aware of what they are consenting to
 - Some participants may have a guardian appointed who is required legally to consent to research.
- d) How will you encourage participants to give feedback?
 - Participants are often uncomfortable in giving negative feedback to those who have designed the software
- e) How will feedback be recorded?
 - Usual methods such as notes and audio recordings may not be suitable
 - Audio recordings may miss information if communication is not through a voice output machine
 - It is difficult to take notes if participants are communicating via eyegaze and word board
- f) Is it necessary to provide incentives to participants?

Payments can interfere with benefit rules

- g) What measures will be put in place after the study to allow participants to continue with new skills if they wish to?
 - Participants may have developed new skills during their involvement with the research and may be keen to continue with this after the project ends.

- h) How will you report back to participants on what you used their feedback for?
 - The traditional methods for reporting back to participants is by writing a letter to participants thanking them for their help and explaining what stage the project is now at. This may not be suitable for users with SSPI.
- i) How will contact be made with participants?
 - Recruiting end users can be a challenge in any situation, unlike traditional development advertising in local press or through local universities can be inappropriate or unlikely to reach the desired participants.

Clause 8 - Guidance on issues for User Centred Design and users with SSPI

- a) Can a group of 'representative users' be gathered?
 - It is not difficult to gather a spectrum of different ages and a mix of genders among participants but gathering a representative sample of different disabilities and needs is more challenging.
 - It may be possible to use purposive or judgemental sampling to find users who cover most of main disabilities (communication impairment, mobility impairment, cognitive impairment and sensory impairment).

- Developers can encourage users to consider how others whom they know with different disabilities would interact with the system.
- Participants can be asked to consider talking to others about the system or to even ask them if they wish to take part in later evaluations.
- By allowing different groups of participants to do later evaluations it is possible to gain as wide a range as possible.
- b) How can different accessibility needs be met
 - Conflicting requirements are already part of the user centred design process, in this situation however discussions between users and designers is unlikely to help resolve the issue.
 - The designers need to carefully examine the possibilities for including as many accessibility and adaptations as possible.
 - It may be possible to switch accessibility options on and off
 - Careful planning during development can often allow different adaptations to be used through the same piece of code.
- c) Can informed consent be gathered from participants, and if not is it possible to gain assent from the participant and informed consent from a legal guardian?
 - By talking to participants after going through Participant Information Sheet and asking yes/no questions we can check that they understand what they are agreeing to.
 - It is still important to gain assent from the participants in the instance of a guardian signing. This should include questions to ensure they want to take part and know what will be happening in the process.

- d) How will you encourage participants to give feedback?
 - Participants need to be encouraged to give negative feedback.
 - It is helpful to remind participants that you really want them to find flaws in the technology.
- e) How will feedback be recorded?
 - Video can be intrusive, but careful planning can minimise this
- f) Is it necessary to provide incentives to participants?
 - Methods of incentives can be through gift tokens etc, which should not interfere with benefit rules.
 - Participants will often volunteer to work without payment due to the enjoyment of taking part.
- g) What measures will be put in place after the study to allow participants to continue with new skills if they wish to?
 - If it is possible for participants to take part in other projects then putting them in contact with the person organising these should be discussed
 - Researchers should discuss with the participants and their support staff what parts of the project they enjoyed most
- h) How will you report back to participants on what you used their feedback for?
 - With participants with SSPI it is beneficial for them to meet with the development team and be given personal feedback on how the project has developed and how their own contributions have been used.
- i) How will contact be made with participants?

- Charities who work with people with SSPI may be willing to make the initial contact on behalf of the development team and then if participants are in agreement to put the two parties in contact.
- There may be additional ethics procedures to go through with a charity and time should be allowed for this.

Clause 9 - Example methods for stages in User Centred Design

General Points

At the beginning of each session review the previous session, this will help any participants who were not present to be caught up on what has happened and will help participants with cognitive problems to recall what took place.

It is important to ensure that participants do not become overtired during the sessions, and to allow for extra time to facilitate breaks. Designers should remember that participants with SSPI are more likely to have illnesses than the general population and there may be times when sessions have to be cancelled or altered at short notice due to illness.

While it is important not to talk in an overly complicated or technical way with participants it is also vital that participants are not spoken to in a condescending tone and are made to feel as equals with the designer.

Understand and specify the context of use

Make contact with potential users and/or organisations who work with people with SSPI. It is important that participants become comfortable with the

researcher/developer prior to the work commencing. The staff or family who spend time with participants should also meet with the researcher/developer.

Examine the ethical situation carefully, in addition to any internal ethics procedures, there may be ethics applications required by the charities or organisations involved. The local health authority may also require an ethics application depending where the sessions with the participants will be conducted and how participants will be recruited.

Spend time in the environment and understand how the technology may be used.

Requirements Gathering

Requirements for how technology could be used

Adults with SSPI may at first be hesitant in giving requirements to researchers due to previous poor experience of having people in authority/people from outside of their immediate circle listening to them. As such a group discussion for requirements may be beneficial – particularly if it can be arranged for the participants to know one another before hand. During group discussions it is important for people to have time to form answers using their communication device and not to feel rushed into answering. Participants will benefit from being given feedback on what was discussed at the end of the session in addition to the traditional summary which is sent out a later date. By feeding back immediately on what the researchers have heard, participants will know that their opinions have been listened to and taken onboard.

Requirements on how technology should operate

Participants may have trouble imagining how technology will work or trouble keeping track of what part of the process in using the technology they are currently imagining. Providing storyboards or videos on problems that currently exist and/or how the technology could possibly be used may help alleviate this. It is important not to specify exactly what the system will do, but allows participants to visualise and imagine more easily. The research/designer can show on the screen an image of the position of the process that is currently being discussed. It may be helpful to prompt for consideration of other groups.

Produce Design Solutions

Lo-fidelity prototyping can involve producing sketches of how technology will look and/or in User Centred Design having the participants produce sketches for how they imagine the system looking. This is not usually a suitable method of prototyping with adults with SSPI. One technique found to be successful is to create more tangible prototypes. Examples of this include making models of the device onto which components can be attached on as the participants wish, or by creating screen designs and then cutting up the buttons etc. These components can then be attached to magnets, the participants can then position the components on a magnetic board in the position they wish. It may also be beneficial to allow participants to use existing accessibility tools and simulate how they work in this situation. It is important that when showing prototypes designs are kept flexible enough to allow instant alterations.

Evaluation

Designers should consider different evaluation tools and not simply accept that evaluations can only take place through observations. Consider different evaluation tools, it may be possible for participants to be trained to perform heuristic evaluations. If support staff can be included in the evaluations and are willing to encourage participants to take part then there can be a role for them, many support workers will be used to completing diaries or logs as part of their daily routine and so an evaluation diary may work better in this situations than it does in a business environment for example.

7.5 Summary

The lessons learnt from Chapter Six were examined for inclusion in a set of guidelines for other developers who wished to work with adults with SSPI in the development of software. These guidelines were produced based upon the format of existing ISO guidelines.

Following the development of these guidelines they were presented to a new software development in order to evaluate their usefulness and to compare the results from Chapter Six. This new software study is presented in Chapter Eight.

Chapter 8. Evaluation of Guidelines and Second Study

"I liked seeing my ideas in the program".

Robert, a participant in the study with SSPI

8.1 Introduction

Chapter Six demonstrated the ability of adults with Severe Speech and Physical Impairments (SSPI) to take part in software development tasks. Given the lack of guidance available for developers wishing to work with this group in design, the author made the decision to produce a set of guidelines based upon lessons learnt in Chapter Six. These guidelines were provided in Chapter Seven, this chapter discusses the second study with adults with SSPI and the evaluation of the guidelines.

8.2 Second Study

Work from this study contributed to a paper presented at the 2010 RaATE conference (see Appendix E).

The author co-supervised HB's project and observed HB during the development. The aim of this observation was to identify information that was not addressed in the guidelines and to monitor the use of the guidelines.

HB was to create a Talking Photoalbum for adults with Severe Speech and Physical Impairments (SSPI). The aim of the software was to allow adults with SSPI to share stories relating to photographs. Adults with SSPI often struggle to share personal narratives (Black et al., 2010b) and it was hoped that this software would help to facilitate this.

People with SSPI often rely upon a carer to enter stories into the device, but this is time consuming and results in static, monologue output as the story text tends to be stored under a single key. Another way of sharing stories is through photographs, with some use being made of "speaking photograph albums"; allowing voice recordings to be linked to individual photographs. However, non-speaking individuals still have to rely on speaking helpers and the playback still mimics a monologue.

Eight adults with SSPI and a support worker were enabled to participate in all stages of the design using innovative design methods; these included magnetic white boards, etran boards and PowerPoint designs.. Participants were initially invited to choose the focus of the project; they expressed a desire to share their stories and their photographs but indicated that they currently require assistance in doing this. The participants then helped design the software from low level paper based designs through to a fully working piece of software.

User Centred Design techniques were employed throughout the development and worked with eight participants with SSPI in all stages of the development (see Table 12). All participants used a day resource centre five days a week.

HB used a variety of methods in the development process. In the identification of the context of use she conducted a focus group with all of the participants. The requirements gathering methods included semi structured interviews and observation of participants using existing interactive photo album technology. The development used methods similar to those seen in the CHAMPION development: magnetic low fidelity prototyping, Clicker5 for mid fidelity and Visual Studio for high fidelity prototyping. A full project report from HB on the work done can be found in Appendix F.

Participant	Method of Augmentative and Alternative Communication (AAC)	Other Means of Communication			
Anna	Dynavox	Vocalisation			
Betty	Lightwriter	Gestures Vocalisation			
Carly	Palm-top communication aid with Bliss symbols broken	Gestures Vocalisation			
Danni	None	Eye gaze			
Eric	Dynavox,	Limited gestures			
Fred	Word board	Gestures, dsyarthric speech			
George	Dynavox	Eye gaze			
Harry	None	Nodding/Shaking Head			

Table 12 - Participant Demographics in HB's Study

8.3 Evolution of Guidelines

While HB was using the guidelines, it was also hoped that she would be able to contribute to the further development of the guidelines. The author recognised that all of the possible information a developer would need could not be learnt from one single development project.

HB provided weekly debriefs in meetings with the author which identified the successes and challenges that she faced on a weekly basis. These weekly meeting notes were analysed using thematic analysis and verified by an independent researcher. These challenges are categorised by the stage in which they were

identified. Table 13 identifies whether the theme presented new information or if this was already addressed in the guidelines.

8.3.1. Preparation

Before HB began her work with adults with SSPI she completed the same set of teaching materials on SSPI and AAC as the author had. In addition to completing the study materials, HB met with two adults with SSPI as part of a separate research project. HB felt that this preparation had helped her to be better prepared and less nervous prior to meeting participants.

8.3.2. Identify Context of Use

Although HB had had this preparation she was still nervous before meeting participants. The author reassured HB that she too had been nervous in the lead up to her first meeting. HB reported being reassured by this.

Following the first meeting with participants HB agreed that she would email the manager looking after the participants with possible dates for sessions. HB sent this straight after the meeting but then struggled to get hold of the manager. This had not been a problem in the pilot, possibly because there was the direct link with the resource staff member on the project. The author offered to help with the management of organising sessions and contacting the manager. This meant HB was able to concentrate on the project and contact was easier for the author who had worked previously with the manager. HB commented on this being a big help.

For the first session HB asked an assistant to act as note taker and record the conversations and interactions in the session. The note taker was able to record much of the discussion but struggled to keep up with the nonverbal interaction and how the participants responded to the discussion. HB realised that a video camera would be

required for the following sessions and it was arranged for this to be available. In the second session the note taker was present again, however after the session it was decided that the video had captured everything that the note taker had, as well as other nonverbal information. For the remaining sessions the note taker was not present with a video camera used instead. In the weekly debriefs HB discussed the benefits of the video camera with the author.

Theme	New Information	Stage Identified		
	(Yes/No)			
It is useful to learn about AAC First	No	Preparation		
It is normal to feel nervous	Yes	Identify Context of Use		
Assistance may be needed in	Yes	Identify Context of Use		
contacting participants				
It is important to determine all the	Yes	Identify Context of Use		
ways in which a participant				
communicates early on				
It can be beneficial to challenge	Yes	Requirements Gathering		
Participants				
A video camera is very useful	Yes	Requirements Gathering		
(essential in some situations)				
It is vital, even in lab notes to protect	Yes	Low Tech Prototyping		
participant anonymity				
It is possible that participants will	Yes	Low Tech Prototyping		
want to keep work from the session				
There should be consistency in design	Yes	High Tech Prototyping		

Table 13- Themes from HB's Work

At the first meeting HB was introduced to a young male participant who introduced himself using his AAC device. HB then went through the informed consent sheet. When it came to the section where participants are asked yes and no questions the participant did not appear to reply. After an uncomfortable three minutes the participant's support worker told HB that the participants did not use their AAC device to communicate yes and no. For these replies the participant used unaided AAC by looking up for 'yes' and down for 'no'. Had HB been aware of this earlier it would have saved awkwardness and confusion.

8.3.3. Requirements Gathering

Participants can be challenged in one of two ways: either as participants in the sessions or as end users using the software. For the 1st design session HB had three activities for the participants to carry out:

1) Participants were asked to select a photograph from a selection provided and discuss what extra information they would want from the photographer;

2) They were asked to use a physical talking photo album;

3) They were asked to design their own version using magnetic paper prototypes.

The author raised concern that participants might struggle with the cognitive load of the tasks and advised HB to have back up plans if participants could not cope. At the end of the sessions HB reported that the participants had enjoyed and coped with the tasks. When the author spoke to the head manager of the service that the participant were from, the manager was enthusiastic about the sessions and said that it was excellent that the participants had been challenged to achieve more.

8.3.4. Prototyping

When participants are challenged during sessions they may feel a sense of pride in their achievements and want to keep the work they have done so that they can show it to others. In some cases this is not possible for example when dealing with the actual software and Newell cautions that this may be a problem (Newell and Gregor, 2000). However, in early stages of design when work is done with paper prototypes HB was able to accommodate this by taking close up photographs of the work and allowing participants to take the originals home.

Throughout the project HB kept detailed handwritten notes in a project logbook. During a meeting the logbook was being referred to and the author noticed that the notes on participants included their real names. This was not surprising given that many of the notes had been taken during meetings with the participants. This can be dangerous as log books are easily misplaced or lost entirely, it is also in violation of the promise given in the ethical consent forms which states that all information linking participants to the study will be stored securely within the School of Computing. HB was told of the importance of using pseudonyms in all documentation which discusses work participants have done in the sessions. The only time a participant's real name should be used is in the participant consent form.

It is of course always important to protect participants' anonymity but particularly with this vulnerable group.

During an interface design review meeting the author noticed that HB had placed the text below the image on each button. The author advised HB to change this so that there was consistency between current paper symbol boards and the new technology.

146

Consistency in design is important so that users can quickly adapt between different pieces of software.

When people with SSPI use paper symbol boards the text is placed along the top of each symbol. This is so that when a communication partner is looking at the board they can read the text if they do not know what the symbol means. When the text is at the bottom of the symbol it can be covered and a partner cannot read it.

It is important to keep a consistent style throughout development and where possible to have this related to a user's existing knowledge.

8.4 Evaluation of Guidelines

HB was presented with the guidelines prior to the beginning of the project and read them in full. She was able to refer to the guidelines at different stages of the project, they were used most prior to the beginning of a new stage in the User Centred Design lifecycle. HB reported finding the guidelines to be very helpful.

8.5 Conclusion of Second Study

The second study had similar results to the first, with participants contributing the majority of the requirements and features (a full breakdown of these is available in Appendix F). Taking both of these studies it is possible to answer the research questions.

8.6 Summary

This chapter discussed the evaluation of the guidelines created in Chapter Seven. These lessons were given to aid a MSc student who was working with adults with SSPI. The lessons and results from this second study were presented in this chapter. The following chapter will consider how the results of the CHAMPION study and HB's work can be used in answering the research questions.

Chapter 9. **Discussion**

"If we can see the goal of the project its worth doing".

Robert, a participant in the study with SSPI

9.1 Introduction

This chapter begins with an overview of the work done in Chapter Six with adults with Severe Speech and Physical Impairments using User Centred Design. The outcomes of this work are then discussed in relation to the literature and the research questions posed in Chapter Four.

9.2 Summary of Research

The CHAMPION study looked at the development of an electronic multimedia patient profiling software for use by adults with Severe Speech and Physical Impairments (SSPI) (see Section 6.2).

Prior to any development the first stage of the User Centred Design lifecycle was the planning of the Human Computer Interaction (HCI) process. This stage involved considering the ethical issues that could occur in the process and how informed consent would be gained from participants with little or no verbal communication.

The first stage of the development was to identify the context of use. Observation was the first method; the author spent time at a care centre for adults with complex disabilities including SSPI. During the week the author spent at the centre, she learnt about the different staff roles and how the service users with SSPI communicated with others. Following the observations, a focus group was held with six adults with

SSPI. This focus group showed that the adults with SSPI could think about the issues surrounding the software and the different situations it would be useful in.

The next stage in the development process was to gather the requirements. Two methods were used for this, forum theatre and storyboarding. Three participants with SSPI were shown a forum theatre video and were able to identify with the characters in the video and imagine alternative endings to the stories presented. A set of seven storyboards were created. Participants were shown the storyboards and asked to think of software requirements for each of these stages.

The prototyping stage in the development of CHAMPION was divided into three phases: low fidelity, mid fidelity and high fidelity.

In the low fidelity stages, participants used magnetic paper prototyping techniques developed by the author. In this stage participants suggested two functional and two non functional requirements. The mid fidelity stages involved the use of Clicker5 software and incorporated the participants' access methods. Participants contributed three functional alterations and two design alterations. The high fidelity prototypes were produced in Visual Studio. In the first iteration of the high fidelity prototyping again the participants showed the ability to critique the software; they were able to identify whether the problem was a design or a functional problem and suggested solutions to the issues.

The short term evaluation took place over the course of three days at the care centre, four participants took part in these evaluations and had 30 minutes training on the software. During this longer term evaluation there were problems with equipment at the centre and this meant that no sessions of more than 5 minutes were attempted.

Despite the problems with the longer term evaluation, the results from the shorter term evaluation were promising and showed that the initial usability of the software was good.

Following the CHAMPION study a set of draft recommendations were created and used by a MSc student on her own software development project with adults with SSPI. This study added further information for recommendations for other developers and showed that it was possible for other developers to carry out full User Centred Design with adults with SSPI.

Taking these studies it is possible to answer the research questions identified in Chapter Four.

9.3 "Can the problems cited in the literature be overcome to allow User Centred Design to be conducted with participants with SSPI?"

The first research question looked at "Can the problems cited in the literature be overcome to allow User Centred Design to be conducted with participants with SSPI?". The problems highlighted in Chapter Four were examined alongside the results from both the author's and HB's study to see if it had been possible to overcome all of them.

9.3.1. Representative Sample

Gathering a representative sample when working with users with SSPI can be difficult. There are many different levels of physical, cognitive, sensory and communication impairments in SSPI (Newell and Gregor, 2000).

In the CHAMPION project, an attempt was made to gather a representative sample of the user group. This meant aiming for a spread of ages and ability, a good mix of male and female participants and a range of different methods of communicating. While it is unlikely that every combination of communication methods and impairments can be included in the one participant pool, it is important to get as representative a sample as possible. With careful planning with support staff it is possible to gain a representative sample.

9.3.2. Achieving Consensus

Another issue that arises from the wide variation found in people with SSPI is that it can make it difficult to achieve consensus on requirements (Newell and Gregor, 2000). This group may find it more difficult to hold discussions on their different needs and be able to discuss the pros and cons of tailoring to each disability (Waller et al., 2005a).

With support and guidance the CHAMPION project showed that participants are able to discuss the different needs and how these should be reflected in the requirements (Prior, 2010).

The other issue is that there can be a danger that a researcher will pay attention to the articulated needs of one participant (Newell and Gregor, 2000). In the study the author found that participants were very good at considering the needs of others and how they would use the technology. They were able to consider different disabilities or abilities. This reduced some of the risk highlighted here. The other help with this problem is for researchers to be aware of the problems when one participant is dominating the conversation. Researchers need to ensure that all the participants

raise their views and that they encourage more confident participants to listen to the views of the shyer participants and encourage them to share their views.

9.3.3. Communicating Thoughts

Newell (2000) suggested that a user may struggle to communication their thoughts. The author understood this concern to mean that a participant may not be able to offer suggestions for how features of a design could be improved, or discuss their views on the way a design is progressing.

It is true that it does take longer when working with these participants for them to be able to share their thoughts (Balandin et al., 2007). The author found in the CHAMPION study that they need to be supported in feeling confident to communicate what they are thinking. The researcher needs to continually encourage participants by reminding them that there are no wrong answers and that any problems they are experiencing with the technology will be useful in helping the researcher to make better designs. Communicating exactly what they are thinking does not come naturally to a wide variety of users, but this may be more true for adults with SSPI who may not have had much experience of being asked to clearly articulate what they are thinking (Waller et al., 2005a). The author found that with time and help the majority of users were able to do this.

Sometimes prompts may help the participants to communicate their thoughts more clearly. In the CHAMPION study rather than be asked directly what information they would want to share with medical staff, participants were shown a storyboard of the hospital experience and asked at each clip on the comic strip what information they think would aid the situation at this point.

9.3.4. New Technology

Augmentative and Alternative Communication (AAC) is a rapidly developing field and many of the new devices onto the market are featuring technology that would not have been imagined a few years ago (for example the Dynavox Eyemax (Mayer-Johnson, 2010) features advanced environmental controls, wheelchair control and cameras to assist vision). Participants may find it difficult to provide requirements for technology which they have not seen before (Rice et al., 2007). Again this is a challenge for all developers, but the additional problem when working with participants with SSPI is that their access to new developments in the wider technology field may have been limited (Waller et al., 2005a).

By using storyboards, scenarios and forum theatre some of these problems can be tackled. These techniques helped participants to use their imagination and visualise how new technology might be used in the future.

9.3.5. Ethical and Legal Issues

Newell (2000) suggests that some of the traditional methods used to uncover a user's thoughts or opinions on a device or software may be unethical when working with this user group, for example letting the participants make mistakes or struggle. To deal effectively with this takes time.

In the CHAMPION study this problem was dealt with by building up a relationship between the author and participants. Once the participants become comfortable with the author, the author would encourage them to find things that were broken or things that they found difficult. Once participants got to know the author they enjoyed 'breaking' the software and pointing out the author's errors. Another issue that it is suggested may be unethical is that often the measure of success of a piece of technology will be the disappointment that is displayed when the technology is removed (Newell and Gregor, 2000). This is a problem, particularly when designing technology that may provide access to communication for people with SSPI.

One solution may be to look at less expensive ways of continuing to provide the software, for example having it run on laptops.

For some people with SSPI it is not possible to gain informed consent (Balandin et al., 2006). In Scotland (and other countries), legislation exists which allows a guardian to be appointed for an adult over the age of eighteen who is deemed by a judge to be incapable of making legal and medical decisions on their own behalf (Adults with Incapacity Act (Scotland), 2000). This guardianship order is appointed through a court order and specifies the different areas in which the guardian has the ability to make decisions on behalf of the adult in question. Some guardianship orders have the provision for the guardian to consent to the adult taking part in research. Even when this order is in place however, it is good ethical practice to receive "assent" from the participant in the same manner as happens with participants under the age of sixteen (Waller et al., 2009).

9.3.6. Payment

Research projects frequently offer participants remuneration in some form in return for their assistance in research work (Molich et al., 2001). There is significant debate over the issue of payment of participants in research (Dickert and Grady, 1999). A major concern in paying participants is that participating in research may cause less economically stable participants to feel undue pressure to volunteer, this is a particular concern in medical research which can often come with important risks (Dickert and Grady, 1999).

In software engineering where the risks tend to be much lower than in medical research (Molich et al., 2001) there is less controversy over the payment of participants in software research (Gebhardt, 2003, Ross, 2010), indeed it is usually expected (Newell and Gregor, 2000). This can cause problems when working with adults with SSPI as it may interfere with benefits rules (Newell and Gregor, 2000). It is permissible to pay expenses for time spent working on the research project and this should be done.

In the CHAMPION study it was found that the participants were happy to work for no payment and got enough pleasure from feeling that they were being useful and from participating in the wider society.

9.3.7. Purchase of Software

In the past in assistive technology the end user was not involved in any stage of the purchase of Assistive Technology (Cudd (ed), 2010). Now the end user is gradually being involved in expressing some of their problems with current technology but they are still not generally the purchaser of the technology (Cudd (ed), 2010). Assistive Technology is usually bought by NHS therapists for their clients (Murray, 2008). This can now be compared to the way technology for children finds its route to market. The children may be involved in design and development but it will be their parents who make the purchase of the technology. The children will however tend to have a say in the purchase (Druin et al., 1998).

Research is also ongoing into how a new model of route to market for assistive technology can be developed as seen by the recent Association for the Advancement of Assistive Technology workshop on Assistive Technology and Technology Transfer (Cudd (ed), 2010)

The first research question identified in Chapter Four investigated if the problems highlighted in the literature could be overcome. Through analysis of the process of the author's and HB's study it has been possible to address these problems. The satisfaction that the users reported having when working on the User Centred Design is a crude measurement, however there are no other reported measures in the literature which can be used to assess the effectiveness of User Centred Design. Therefore it would appear that the problems cited in the literature can be overcome to allow User Centred Design to be conducted with participants with SSPI.

9.4 Can methods currently used in Human Computer Interaction be combined and adapted to allow people with SSPI to contribute to the design of technology?

The second research question was concerned with how current methods used in Human Computer Interaction (HCI) could be adapted for use with participants with SSPI.

A review of some of the common methods in HCI was presented in Chapter Five. It is not possible to quantify exactly how many HCI methods are currently in use, developers and experts in HCI are continually striving to produce new methods. The author chose the methods they believed to be the most commonly used and analysed these for use in the CHAMPION development.

The CHAMPION project used ten methods in total in all the stages up to the longitudinal evaluation (see Chapter Six) and participants with SSPI reported feeling

useful and that they were contributing in all of the methods, although they conceded that many were challenging. There was not a measure take of the degree of involvement the participants felt. It is possible that a likert scale could have been used to provide measures for this involvement but the author was more interested in the analysis of the results from the debriefing focus group.

None of the adapted methods had to be abandoned and participants were able to complete all of the tasks in the methods. The one stage that was not completed was the long term evaluation, this stage did not complete due to staff finding it difficult to find the time needed to support participants in using the software. The participants were able to use the software independently and were in many cases able to set up the hardware themselves. However, in the care centre the service users are not allowed to use the hardware without supervision for fear of breaking the technology. On the occasions when the author had to return to the care centre to correct a problem with the hardware, the participants were able to demonstrate to her that they knew how to fix the problem but were not allowed to. In the shorter evaluation all participants were able to complete the methods that were intended to be repeated throughout the longitudinal evaluation.

The methods were not greatly adapted with the exception of the focus group and low fidelity prototyping. For the majority of the methods they were conducted in a manner similar to that found in a general development (Sharp et al., 2007). The main difference was the way the questions were phrased and the time taken to interpret the answers given by participants. This did require considerable preparation time, in a traditional development the preparation to session time ratio is reported at 3:10 (Gebhardt, 2003). In the CHAMPION development the ratio of preparation to

session time was on average 8:1, HB's project was higher at 14:1 although this may have been due to HB having less experience than the author in User Centred Design. This extra time should be factored into the project planning.

As previously discussed (see Chapter Six), often it is not possible to cover as much detail and information in one session with adults with SSPI as would be covered when working with adults without disability. This means that a researcher must be willing to plan additional sessions to ensure all the information is covered.

In each stage of the development the participants made a larger contribution to the requirements, ideas and suggestions than the author. Table 14 shows the percentage of requirements and suggestions for features that came from the participants in each of the sessions. The lowest level of contribution was in the mid fidelity stage, the author believes that there are two possible reasons for the reduction in involvement in this stage. The first possible reason is that participants may have been struggling to make the adjustments to the more polished appearance of the software. In this stage, Clicker5 was used may have made the participants more hesitant of providing negative criticism, however their level of involvement significantly increased by 27% in the next stage. The possible other reason is that two of the participants were feeling unwell during the session. Both were keen to take part but it became clear during the session that they were not as willing as they had been previously to engage in conversation.

The calculation of the degree of participant involvement was taken by analysing the transcripts of the sessions, looking at the changes made after each session and the discussion that preceded this change. This analysis was verified by an independent

researcher. HB conducted a similar analysis and found a contribution of 68% (see Appendix F).

Observation	Focus	Forum	Story	Low	Mid	High	High	Evaluation	Total
	Group	Theatre	Boards	Fidelity	Fidelity	Fidelity	Fidelity		
				Prototype	Prototype	Prototype	Prototype		
						#1	#2		
60%	100%	100%	100%	100%	58%	83%	100%	Unable to	78%
								Complete ¹³	

Table 14 - Degree of Participant Involvement

As stated previously it is not known if the participants in either study would have been able to cope with more traditional methods or if traditional methods would have yielded an even greater response from participants. With correct planning, consideration of the risks and with participants who understand the experimental nature of the methods it may be possible to explore methods not used in this thesis. A review of the implementation of the methods used in CHAMPION were included in the guidelines discussed in Chapter Seven, guidance on how other methods could also be adapted were included in the guidelines. Updated guidance on how to carry out these methods with this group are presented in Appendix L.

Research question two was investigating if methods currently used in Human Computer Interaction (HCI) could be combined and adapted to allow people with SSPI to contribute to the design of technology. Through analysis of the features in the CHAMPION study it can be seen that participants made a significant contribution to the features of the software. This work has shown that several

¹³ As the end evaluation was not possible to complete, it was not included in this calculation

methods currently used in Human Computer Interaction can be combined and adapted to allow people with SSPI to contribute to the design of technology. Further work would be required to investigate if all methods currently used in HCI are suitable for this user group.

Having answered these research questions it is now possible to consider the implications of this work.

9.5 Summary

This thesis chapter has provided a summary of Chapters Five and Six. The results of the work in Chapter Eight were discussed and their relation to the research questions presented. The final chapter of this thesis, Chapter Ten will present the implications of this work in relation to practice, policy and research.

Chapter 10. Implications and Conclusion

"One last thing, thank you for listening".

Doug, a participant in the study with SSPI

10.1 Introduction

As stated in Chapter Nine, the work contained in this thesis and the outcomes of the research questions has implications for a variety of fields.

This chapter provides the implications of this research for the fields of academic research, policy and practice. Areas for future study are presented in these implications.

10.2 Implications for Policy

10.2.1. Healthcare

This work has shown that adults with Severe Speech and Physical Impairments (SSPI) can make decisions based on the evidence that they are presented with and the opinions and needs of others. This information needs to be made clear to healthcare professionals, often in healthcare adults with SSPI find that their relatives are being asked to make decisions on their behalf, despite the fact that they are capable of making these decisions independently (Hemsley et al., 2008a). In the studies reported in this thesis, the adults with SSPI were also able to communicate the reasons behind their opinions, this communication needs to be supported by both their support staff and the medical professionals, and additional training may be of benefit here.

10.2.2. Education

In the CHAMPION study, adults with SSPI were encouraged to voice their own opinions and make decisions independently. This did not appear to come naturally to two of the participants, both of whom had been educated in special needs schools. The participant who took to sharing his opinions the quickest had been educated in mainstream schools. By the end of the work all of the participants were actively advocating their opinions. One participant reported feeling that the work in the CHAMPION study had been a form of 'training' for advocating their opinions. It may be that more work in advocacy in special needs schools could benefit these young people as they transition from school to adult services.

10.3 Implications for Practice

10.3.1. Care Centre User Committees

Currently many care centres have service user committees (e.g. Capability Scotland, 2011) where service users can raise problems and concerns that they have. Typically the staff will then offer to produce solutions for these problems. This work suggests that service users with appropriate support and encouragement can take more responsibility for gathering the views of the entire community and bring them to committee. It may also be possible for these representatives to take more responsibility for finding solutions and taking the steps required for these solutions to be put in place. This would empower the adults with SSPI to take more responsibility for their own lives. As stated in Chapter One, adults with SSPI frequently report feeling a lack of empowerment (Edyburn, 2006) and this can lead to depression (Wressle and Samuelsson, 2004). This work has the potential to influence policy in this area.

10.3.2. Healthcare

The CHAMPION system which was developed in Chapter Six is now at the high level functioning prototype stage and is ready for the development needed to have it running widely. One way in which the software could be quickly brought out for use by care centres would be to have the software available in the centre along with the specialist computers needed for it to run on hospital wards. Service users at the centres could prepare their own multimedia profiles over time on the centre's computers. When a service user was admitted to hospital this profile could be transferred onto a specialist computer quickly and then taken to them in the hospital. A longterm goal would be to have the CHAMPION information stored with each patient's electronic medical record so that this information was available to GPs and hospitals across the country; however as discussed in Section 3.4, the electronic medical records system for the NHS is still far from completion.

10.3.3. Employment

One particular area of practice in which this work has implications is that of employment. There is evidence in the literature that adults with SSPI are often unemployed (McNaughton et al., 2002a, Holmes et al., 2010). One reason cited for this is due to a lack of confidence on the part of the adults or their families (McNaughton et al., 2002a, McNaughton et al., 2006). This work has shown that with encouragement, adults with SSPI can grow in confidence and develop social skills.

Another issue identified was the lack of usability of AAC devices (McNaughton et al., 2002a), this relates to the problems identified in Chapter Two. The devices currently do not allow the spontaneity needed in the work place (McNaughton et al., 2006). If the usability of AAC devices can be improved through User Centred Design then it may be that the devices are better suited to enabling adults with SSPI to enter employment.

The Communication Champion¹⁴ for children and young people aged 0-19 in England, Jean Gross, has made available the figures for AAC costs and provision in England. The Department for Education in England estimate that £500,000 is saved over the course of a lifetime for every person who changes from permanent unemployment to permanent employment (Gross, 2010). Gross suggests that this means that if even just one in ten children who require AAC in England were able to enter employment it would save the government £310 million over the course of their lives (Gross, 2010). The current cost of providing all children in England with appropriate AAC and support is estimated at £10 million. This means that with appropriate AAC and support, employment could be provided for these children, then the potential cost savings would be large.

Producing more usable AAC devices will require the input of Human Computer Interaction (HCI) developers. This work has important implications for HCI research.

10.4 Implications for Research

10.4.1. Health and Social Research

This work has shown that adults with SSPI can provide informed consent and communicate their thoughts clearly. This indicates there is potential for them to be more actively involved as participants than has often been the case previously.

In addition to this, there is the potential for them to act on advisory panels in medical research, as is common in other fields (Central and North West London, 2004). Other

¹⁴ The Communication Champion is responsible for improving services for children and young people who have speech and communication needs. The Champion will work with variety of stakeholders and agencies to do this (Bercow Report, 2008)
groups recommend training for service users, some of the training in advocacy used in this work could be of use here. Similar training could also be of benefit to adults with SSPI wishing to enter employment.

10.4.2. Further Development of Guidelines

It is likely that as more projects are conducted with participants with SSPI, more lessons will be learnt which could feed into these guidelines. Therefore the guidelines are currently being produced as a wiki webpage which developers can have access to and contribute back their own experiences and lessons that they learnt when conducting this work.

In order to empirically measure the effectiveness of the guidelines, a large multi centre study is likely to be necessary with control groups developing AAC software without any User Centred Design, groups conducting User Centred Design without the guidelines and groups conducting User Centred Design with the help of the guidelines.

10.4.3. Implications for HCI Research

This work has produced a set of recommendations for developers, when HB was asked for her opinion of the guidelines she stated that "the guidelines were very useful in helping me prepare for the challenges I faced.".

HB's work had additional lessons for HCI developers and these have been added to the recommendations. The recommendations will need continual modification as more work is done with adults with SSPI. The current version of the guidelines can be found in Appendix L. This work has shown that it is possible for HCI developers to work with adults with SSPI in the User Centred Design of software and this is something that all developers working in AAC software should give consideration to.

As seen in the CHAMPION development the care centre was not always the ideal environment for these sessions to occur. There were frequent interruptions and the rooms often had to be rearranged to allow for sessions to take place. For the final evaluations it is important that these take place in a user's own environment but for the earlier stages there may be an advantage in conducting the work in a specialist centre. This is something HCI developers should consider when planning a project. This is an area for future research.

The other disadvantage was that holding the sessions in a centre meant the participant pool was limited to those attending the centre. Adults with SSPI who do not attend care centres are restricted in their ability to take part in research if it is limited to these locations.

One solution to this would be to set up a dedicated User Centre within an accessible building so that adults from different centres and those who do not attend centres could meet and participate in research together. The model for this could be based on the existing Older Adults User Centre at the University of Dundee's School of Computing (Forbes et al., 2009).

A pilot study is now underway to look at the feasibility of having an AAC User Group meeting regularly at the School of Computing. The group currently meets once a fortnight. The group sessions look at developing the members' own IT skills and at including them in a wide range of research projects, including undergraduate projects. A full description of the work taking place at the centre is available in Appendix K.

The goals of the AAC User Centre are firstly to provide a means for adults with SSPI to engage with the wider community and develop their own skills. Secondly it is to provide a resource for researchers who wish to work with adults with SSPI in their projects. The third goal is to educate students on the needs of people with SSPI and allow them the opportunity to work with adults with SSPI.

The centre is currently still in its pilot stage and is currently run by members of the School of Computing AAC research team, in the future it is hoped that the centre can be expanded and a member of staff can be dedicated to the running of it.

Another area for future investigation is to look at if improvements in usability reduce abandonment. As was stated in Chapter Four the aim of this work was not to investigate if User Centred Design meant a reduction in abandonment of AAC technology. This work was the first stage in the process of investigating this. Now that it was been established that User Centred Design can be conducted with adults with SSPI the next stage would be to conduct a large randomised control trial looking at AAC developments which used User Centred Design with end users and AAC development which worked with traditional participants (e.g. Speech and Language Therapists). The different devices would then need to be used with a large set of participants who used AAC to look at the different rates of abandonment over time.

Conducting a trial of this nature would be incredibly complicated, people with SSPI are a very heterogeneous group (Glennen and Descoste, 1997) and finding enough

participants with a wide enough range of impairments would prove challenging, indeed it may be that the study would have to be international.

To find a large enough pool of AAC developers would also likely require that AAC manufacturers came onboard and provided their staff to be developers.

An alternative would be to encourage other researchers to consider using User Centred Design with end users with SSPI and if a large enough pool of case studies could be gathered, to perform a case study analysis.

10.5 Summary

This thesis has investigated the issue of abandonment in Augmentative and Alternative Communication (AAC) and how this may be linked to poor software usability. The most common method in software development for improving usability is to use User Centred Design with end participants. In the development of AAC software there has traditionally been limited work done in User Centred Design and many reasons for this are cited in the literature.

This thesis has reported on a pilot study which involved adults with SSPI throughout the development of a piece of information sharing software. This work resulted in a set of recommendations which were then provided to a MSc student in a design study. Using the recommendations the student was able to complete a second User Centred Design study involving adults with SSPI. The student found the recommendations to be highly usable. The results from these studies were used to inform the answers to the research questions.

This work has implications for medical research, employment for adults with SSPI, social care and in particular Human Computer Interaction.

To further develop the techniques in conducting User Centred Design, a specialised centre has been created and is currently in a pilot study.

References

- ADAMO-VILLANI, N. 2006. A Virtual Learning Environment for Deaf Children: Design and Evaluation. World Academy of Science, Engineering and Technology, 16, 18-24.
- ADULTS WITH INCAPACITY ACT (SCOTLAND) 2000.
- ALDRED, C., GREEN, J. & ADAMS, C. 2004. A new social communication intervention for children with autism: pilot randomised controlled treatment study suggesting effectiveness. *Journal of Child Psychology and Psychiatry*, 45, 1420-1430.
- ALLEN, J. 2005. Designing desirability in an augmentative and alternative communication device. 4, 135-145.
- ALLEN, M., MCGRENERE, J. & PURVES, B. 2007. The Design and Field Evaluation of PhotoTalk: A Digital Image Communication Application for People with Aphasia. Assets'07: Proceedings of the Ninth International Acm Sigaccess Conference on Computers and Accessibility, 187-194.
- ALM, N., DYE, R. & HARPER, G. 1995. ALADIN Advanced Language Device for Interaction. *Assistive technology a way to independent life*. Lisbon.
- AMERICAN SPEECH-LANGUAGE-HEARING ASSOCIATION 2002. Augmentative and Alternative Communication: Knowledge and Skills for Service Delivery [Knowledge and Skills]. <u>www.asha.org/policy</u> downloaded 19 July 2008.
- ARMITAGE, G., SUTER, E., OELKE, N. & ADIAR, C. 2009. Health systems integration: state of the evidence. *International Journal of Intergrated Care*, 9.
- ASHA. 2011. Augmentative and Alternative Communication (AAC) [Online]. American Speech-Language-Hearing Association. [Accessed 01-02-2011 2011].
- AUGMENTATIVE AND ALTERNATIVE COMMUNICATION AT THE UNIVERSITY OF WASHINGTON. 2009. *Rate Enhancement* [Online]. Seattle. [Accessed 20 April 2009 2009].
- BACH, P., DELINE, R. & CARROLL, J. 2009. Designers wanted: participation and the user experience in open source software development. *Proceedings of the 27th international conference on Human factors in computing systems.* Boston, MA, USA: ACM.
- BAILEY, J., BUDGEN, D., TURNER, M., KITCHENHAM, B., BRERETON, P. & LINKMAN, S. Evidence relating to Object-Oriented software design: A survey. Empirical Software Engineering and Measurement, 2007. ESEM 2007. First International Symposium on, 20-21 Sept. 2007 2007. 482-484.
- BALANDIN, S. 2002. Message from the President. The Bulletin ISAAC, 67, 2.
- BALANDIN, S., BERG, N. & WALLER, A. 2006. Assessing the loneliness of older people with cerebral palsy. *Disability & Rehabilitation*, 28, 469 479.
- BALANDIN, S., HEMSLEY, B., SIGAFOOS, J. & GREEN, V. 2007. Communicating with nurses: The experiences of 10 adults with cerebral palsy and complex communication needs. *Applied Nursing Research*, 20, 56-62.
- BARRETO, A. B., SCARGLE, S. D. & ADJOUADI, M. 1999. A real-time assistive computer interface for users with motor disabilities. ACM.
- BARTLETT, G., BLAIS, R., TAMBLYN, R., CLERMONT, R. J. & MACGIBBON, B. 2008. Impact of patient communication problems on the risk of preventable adverse events in acute care settings. *Canadian Medical Association Journal*, 178, 1555-1562.

- BATTLESON, B., BOOTH, A. & WEINTROP, J. 2001. Usability testing of an academic library Web site: a case study. *The Journal of Academic Librarianship*, 27, 188-198.
- BERCOW REPORT 2008. A Review of Services for Children and Young People (0– 19) with Speech, Language and Communication Needs. Nottingham: DCSF Publications.
- BEUKELMAN, D. & MIRENDA, P. 1992. Augmentative and Alternative Communication, USA, Paul H.Brookes.
- BEVAN, N. & SPINHOF, L. 2007. Are guidelines and standards for web usability comprehensive? Proceedings of the 12th international conference on Humancomputer interaction: interaction design and usability. Beijing, China: Springer-Verlag.
- BLACK, R., WALLER, A., REITER, E. & TURNER, R. 2010a. A new tool to support interactive narrative experience for children with communication disorders. *Biennial Conference of the International Society for Augmentative and Alternative Communication* Barcelona, Spain.
- BLACK, R., WALLER, A., REITER, E. & TURNER, R. 2010b. "A New Tool to Support Interactive Narrative Experience for Children with Communication Disorders". 14th Biennial Conference of the International Society for Augmentative and Alternative Communication. Barcelona, Spain.
- BLACKSTONE, S. W. 2009. Evidence: Using AAC to support patient-provider communication. *Augmentative Communication News* 21, 11-12.
- BLUM, R. W., RESNICK, M. D., NELSON, R. & GERMAINE, A. S. 1991. Family and Peer Issues Among Adolescents With Spina Bifida and Cerebral Palsy.
- BOAL, A. 2000. Theatre of the Oppressed, Pluto Press.
- BOEHM, B. & IN, H. Identifying quality-requirement conflicts. Requirements Engineering, 1996., Proceedings of the Second International Conference on, 15-18 Apr 1996 1996. 218.
- BOEHM, B. W., BROWN, J. R. & LIPOW, M. 1976. Quantitative evaluation of software quality. *Proceedings of the 2nd international conference on Software engineering*. San Francisco, California, United States: IEEE Computer Society Press.
- BOYD-GRABER, J., L., NIKOLOVA, S., S., MOFFATT, K., A., KIN, K., C., LEE, J., Y., MACKEY, L., W., TREMAINE, M., M. & KLAWE, M. 2006. Participatory design with proxies: developing a desktop-PDA system to support people with aphasia. *Proceedings of the SIGCHI conference on Human Factors in computing systems*. Montr\&\#233;al, Qu\&\#233;bec, Canada: ACM.
- BRAUN, J., AL MAHMUD, A. & MARTENS, J.-B. 2010. Capture Me! Helping People with Aphasia to Capture and Share Life Experiences. *BCS HCI 2010*. Dundee, Scotland.
- BRERETON, P., KITCHENHAM, B. A., BUDGEN, D., TURNER, M. & KHALIL, M. 2007. Lessons from applying the systematic literature review process within the software engineering domain. *Journal of Systems and Software*, 80, 571-583.
- BRITTEN, N. 1995. Qualitative Research: Qualitative interviews in medical research. *BMJ*, 311, 251-253.
- BROWN-SICA, M. 2010. User-centered design of learning spaces on a diverse, urban commuter campus. 2010 National Diversity in Libraries Conferences. Princeton, New Jersey.
- CAMERON, J. 2005. Focussing on the Focus Groups. *In:* HAY, I. (ed.) *Qualitative Research Methods in Human Geography.* 2nd ed. Melbourne: Oxford University Press.

- CAPABILITY SCOTLAND. 2011. Supported Living [Online]. Available: <u>http://www.capability-scotland.org.uk/supportedliving.aspx</u> [Accessed 19 March 2011].
- CARMICHAEL, A., NEWELL, A., DICKINSON, A. & MORGAN, M. 2005. Using theatre and film to represent user requirements. *Proceeding of Include Conference Royal College of Art.* London.
- CARMICHAEL, A., RICE, M., LINDSAY, S. & OLIVIER, L. 2008. iTV as a Platform for Rich Multimedia Reminders for People with Dementia. *Proceedings of the 6th European conference on Changing Television Environments*. Salzburg, Austria: Springer-Verlag.
- CARTWRIGHT, C. A. & CARTWRIGHT, P. G. 1974. *Developing Observation Skills*, McGraw-Hill.
- CATTERALL, M. & MACLARAN, P. 1997. Focus Group Data and Qualitative Analysis Programs: Coding the Moving Picture as Well as the Snapshots. *Sociological Research Online*, 2.
- CENTRAL AND NORTH WEST LONDON 2004. Service User Involvement Policy. *In:* NHS (ed.).
- CHANDRASHEKAR, S., STOCKMAN, T., FELS, D. & BENEDYK, R. 2006. Using think aloud protocol with blind users:: a case for inclusive usability evaluation methods. *Proceedings of the 8th international ACM SIGACCESS conference on Computers and accessibility.* Portland, Oregon, USA: ACM.
- CLAYTON, L. 2006. HCI and cognitive disabilities. *Magazine interactions A contradiction in terms?*, 13, 14-15.
- COMMUNICATION MATTERS. 2003. *What is AAC*? [Online]. Communication Matters. Available: <u>http://www.communicationmatters.org.uk/downloads/focuson/What_is_AAC_A</u> 4.pdf [Accessed 19 August 2008 2008].
- COMMUNICATION MATTERS. 2008. Let your Hands do the Talking [Online]. Communication Matters. Available: <u>http://www.communicationmatters.org.uk/downloads/focuson/Let_your_Hands</u> <u>do_the_Talking_A4.pdf</u> [Accessed 19 August 2008 2008].
- CRICK SOFTWARE. 2009. *About Clicker 5* [Online]. Available: <u>http://www.cricksoft.com/UK/products/clicker/</u> [Accessed 11 Sept 2009].
- CRICKSOFT. 2011. *Clicker5* [Online]. Cricksoft. Available: <u>http://www.cricksoft.com/uk/products/tools/clicker/home.aspx</u> [Accessed 19 March 2011].
- CUDD (ED), P. Proceedings of the AT Technology Transfer Workshop. AT Technology Transfer, 2010 Sheffield, UK.
- CULP, D. M., AMBROSI, D. M., BERNIGER, T. M. & MITCHELL, J. O. 1986. Augmentative communication aid use - A follow-up study. Informa Healthcare.
- CUMELLA, S. & MARTIN, D. 2000. Secondary Healthcare for People with a Learning Disability. Department of Health.
- DARAN, M. & THÉVENOD-FOSSE, P. 1996. Software error analysis: a real case study involving real faults and mutations. *SIGSOFT Softw. Eng. Notes*, 21, 158-171.
- DATTILO, J., ESTRELLA, G., ESTRELLA, L. J., LIGHT, J., MCNAUGHTON, D. & SEABURY, M. 2008. "I have chosen to live life abundantly": perceptions of leisure by adults who use Augmentative and Alternative Communication. AAC: Augmentative & Alternative Communication, 24, 16-28.
- DAVIES, R., MARCELLA, S., MCGRENERE, J. & PURVES, B. 2004. The ethnographically informed participatory design of a PD application to support

communication. *Proceedings of the 6th international ACM SIGACCESS conference on Computers and accessibility.* Atlanta, GA, USA: ACM.

- DE JONGE, D. M. & RODGER, S. A. 2006. Consumer-identified barriers and strategies for optimizing technology use in the workplace. Informa Healthcare.
- DEMCHAK, M., RICKARD, C. & ELQUIST, M. 2002. Using Cues to Enhance Receptive Communication. *In:* PROJECT, N. D. S. I. (ed.). Nevada: University of Nevada.
- DEPARTMENT OF HEALTH 2003. Care Homes for Adults (18-65). National Minimum Standards Care Homes Regulations. *In:* HEALTH, D. O. (ed.). London: The Stationery Office.
- DERUYTER, F., MCNAUGHTON, D., CAVES, K., BRYEN, D. N. & WILLIAMS, M. B. 2007. Enhancing AAC connections with the world. *Augmentative and Alternative Communication*, 23, 258 270.
- DICKERT, N. & GRADY, C. 1999. What's the Price of a Research Subject? Approaches to Payment for Research Participation. *New England Journal of Medicine*, 341, 198-203.
- DICKINSON, A., ARNOTT, J. & PRIOR, S. 2007. Methods for human-computer interaction research with older people. *Behaviour and Information Technology*, 26, 343-352.
- DOURISH, P., LAMPING, J. & RODDEN, T. 1999. Building bridges: customisation and mutual intelligibility in shared category management. *Proceedings of the international ACM SIGGROUP conference on Supporting group work.* Phoenix, Arizona, United States: ACM.
- DRUIN, A., BEDERSON, B., BOLTMAN, A., MIURA, A., KNOTTS-CALLAHAN, D. & PLATT, M. 1998. Children as our technology design partners. *The design of children's technology*. Morgan Kaufmann Publishers Inc.
- DUNLOP, H., CUNNINGHAM, S. J. & JONES, M. 2002. A digital library of conversational expressions: helping profoundly disabled users communicate. *Proceedings of the 2nd ACM/IEEE-CS joint conference on Digital libraries*. Portland, Oregon, USA: ACM.
- EASTON, R. D. 1992. Inherent Problems of Attempts to Apply Sonar and Vibrotactile Sensory Aid Technology to the Perceptual Needs of the Blind.
- EDMOND, I. & WILLIAM, M. 2004. Human Factors Risk Management as a Way to Improve Medical Device Safety: A Case Study of the Therac 25 Radiation Therapy System. *Joint Commission Journal on Quality and Patient Safety*, 30, 689-695.
- EDYBURN, D. 2006. Assistive Technology and Mild Disabilities. *Special Education Technology Practice*, 8, 18-28.
- EISMA, R., DICKINSON, A., GOODMAN, J., SYME, A., TIWARI, L. & NEWELL, A. F. 2004. Early user involvement in the development of information technology-related products for older people. *Universal Access in the Information Society*, 3, 131-140.
- ENDERBY, P. & PHILIPP, R. 1986. Speech and language handicap: Towards knowing the size of the problem. *British Journal of Disorders of Communication*, 27, 159-173.
- ENDERS, A. & HALL, M. 1990. Assistive Technology Sourcebook, Washington, Resna Press.
- ENGELBERG, D. & SEFFA, A. 2002. A Framework for Rapid Mid-Fidelity Prototyping of Web Sites. Proceedings of the IFIP 17th World Computer Congress - TC13 Stream on Usability: Gaining a Competitive Edge. Kluwer, B.V.

- FINNIE, N. R. & BAVIN, J. 1997. *Handling the Young Child with Cerebral Palsy at Home*, Elsevier Health Services.
- FORBES, F., GIBSON, L., HANSON, V., GREGOR, P. & NEWELL, A. 2009. Dundee user centre: a space where older people and technology meet. Proceedings of the 11th international ACM SIGACCESS conference on Computers and accessibility. Pittsburgh, Pennsylvania, USA: ACM.
- FOREMAN, P. & CREWS, G. 1998. Using augmentative communication with infants and young children with Down syndrome. *Downs Syndr Res Pract*, 5, 16-25.
- GEBHARDT, A. 2003. *Rapid Prototyping*, Cincinnati, Ohio, Hanser Gardner Publications.
- GIBBS, A. 1997. Focus Groups. Social Research Update, 1997.
- GLANVILLE, J. 2006. Literature Searching To Inform Systematic Reviews. York.
- GLENNEN, S. & DESCOSTE, D. 1997. *The Handbook of Augmentative and Alternative Communication*, Thomson Delmar Learning.
- GOLD, R. L. 1958. Roles in Sociological Field Observations. Social Forces, 36, 217-223.
- GOODMAN, R. 1997. The Strengths and Difficulties Questionnaire: A research note. *Journal of Child Psychology and Psychiatry*, 38, 581-586.
- GOULD, J., D. & LEWIS, C. 1983. Designing for usability\—key principles and what designers think. *Proceedings of the SIGCHI conference on Human Factors in Computing Systems*. Boston, Massachusetts, United States: ACM.
- GREGOR, P., NEWELL, A., F. & ZAJICEK, M. 2002. Designing for dynamic diversity: interfaces for older people. *Proceedings of the fifth international ACM conference on Assistive technologies*. Edinburgh, Scotland: ACM.
- GRIFFITH, L. E., COOK, D. J., GUYATT, G. H. & CHARLES, C. A. 1999. Comparison of Open and Closed Questionnaire Formats in Obtaining Demographic Information From Canadian General Internists. *Journal of Clinical Epidemiology*, 52, 997-1005.
- GROSS, J. 2010. AAC The Future. Annual Communication Matters Symposium. Leicester, UK.
- GROSSMAN, T., FITZMAURICE, G. & ATTAR, R. 2009. A survey of software learnability: metrics, methodologies and guidelines. *Proceedings of the 27th international conference on Human factors in computing systems.* Boston, MA, USA: ACM.
- GROVE, N. 2003. Multi Media Profiling: a person centred toold for people with profound disabilities. *PMLD Link*, 15, 5-7.
- HAKIM, J. & SPITZER, T. 2000. Effective prototyping for usability. *Proceedings of IEEE professional communication society international professional communication conference and Proceedings of the 18th annual ACM international conference on Computer documentation: technology* & *teamwork.* Cambridge, Massachusetts: IEEE Educational Activities Department.
- HAMM, R. J., O'DELL, D. M., PIKE, B. R. & LYETH, B. G. 1993. Cognitive impairment following traumatic brain injury: the effect of pre- and post-injury administration of scopolamine and MK-801. *Cognitive Brain Research*, 1, 223-226.
- HAMMERSLEY, M. & ATKINSON, P. 1994. *Ethnography: Principles in Practice,* London, Routledge.
- HAMPSON, J. 2006. Partner Assisted Communication Systems: Let me show you how I communicate. *CM2005 National Symposium*. University of Leicester: Communication Matters.
- HEMSLEY, B., BALANDIN, P. S. & TOGHER, L. 2008a. 'We need to be the centrepiece': Adults with cerebral palsy and complex communication needs

discuss the roles and needs of family carers in hospital. *Disability & Rehabilitation*, Epub ahead of print.

- HEMSLEY, B. & BALANDIN, S. 2004. Without AAC: The Stories of Unpaid Carers of Adults with Cerebral Palsy and Complex Communication Needs in Hospital. *Augmentative and Alternative Communication*, 20, 243 258.
- HEMSLEY, B., BALANDIN, S. & TOGHER, L. 2007. Older Unpaid Carers' Experiences Supporting Adults with Cerebral Palsy and Complex Communication Needs in Hospital. *Journal of Developmental and Physical Disabilities*, 19, 115-124.
- HEMSLEY, B., BALANDIN, S. & TOGHER, L. 2008b. 'I've got something to say': interaction in a focus group of adults with cerebral palsy and complex communication needs. *AAC: Augmentative & Alternative Communication*, 24, 110-122.
- HEMSLEY, B., BALANDIN, S. & TOGHER, L. 2008c. Professionals' views on the roles and needs of family carers of adults with cerebral palsy and complex communication needs in hospital. *Journal of Intellectual & Developmental Disability*, 33, 127 - 136.
- HEMSLEY, B., SIGAFOOS, J., BALANDIN, S., FORBES, C., TAYLOR, V., GREEN, A. & PARMENTER, T. 2001. Nursing the patient with severe communication impairment. *Journal of Advanced Nursing*, 35, 827-835.
- HENGEVELD, B., VOORT, R., HUMMELS, C., DE, M., JAN, VAN, B., HANS, OVERBEEKE, K. & VAN DER, H., AADJAN 2008a. The development of LinguaBytes: an interactive tangible play and learning system to stimulate the language development of toddlers with multiple disabilities. Hindawi Publishing Corp.
- HENGEVELD, B., VOORT, R., HUMMELS, C., OVERBEEKE, K., DE, M., JAN & VAN, B., HANS 2008b. LinguaBytes. *Proceedings of the 7th international conference on Interaction design and children*. Chicago, Illinois: ACM.
- HIGGINBOTHAM, D. J., BISANTZ, A. M., SUNM, M., ADAMS, K. & YIK, F. 2008a. The effect of context priming and task type on augmentative communication performance. *Augmentative and Alternative Communication*, 25, 19-31.
- HIGGINBOTHAM, D. J., BISANTZ, A. M., SUNM, M., ADAMS, K. & YIK, F. 2008b. The effect of context priming and task type on augmentative communication performance. *Augmentative and Alternative Communication*, 99999, 1 - 13.
- HOLMES, K., JUDGE, S. & MURRAY, J. Communication Matters Research Matters: an AAC Evidence Base. Communication Matters, 2010 Leicester.
- HOPPESTAD, B. S. 2007. Inadequacies in computer access using assistive technology devices in profoundly disabled individuals: An overview of the current literature. Informa Healthcare.
- HORNOF, A. 2008. Working with children with severe motor impairments as design partners. *Proceedings of the 7th international conference on Interaction design and children*. Chicago, Illinois: ACM.
- HUGHES, J., KING, V., RODDEN, T. & ANDERSEN, H. 1994. Moving out from the control room: ethnography in system design. Lancaster: Computing Department, Lancaster University.
- HUSTAD, K. C. 2006a. A Closer Look at Transcription Intelligibility for Speakers With Dysarthria: Evaluation of Scoring Paradigms and Linguistic Errors Made by Listeners.

- HUSTAD, K. C. 2006b. A Closer Look at Transcription Intelligibility for Speakers With Dysarthria: Evaluation of Scoring Paradigms and Linguistic Errors Made by Listeners. *American Journal of Speech-Language Pathology*, 15, 268-277.
- IRAS. 2011. Integrated Research Application System. [Online]. NHS. Available: https://www.myresearchproject.org.uk [Accessed 17 Feb 2011 2011].
- ISENSEE, S. & RUDD, J. 1996. The art of rapid prototyping.
- ISO 1999. 13407 Human-centred design processes for interactive systems.: International Organization for Standardization,.
- ISO 2008. ISO 92410171:2008. Ergonomics of human-system interaction -- Part 171:Guidance on Software Accessibility. International Standards Organisation.
- ISO 2010. 9241-210 Human-centred design processes for interactive systems.: International Organization for Standardization,.
- ISO/TS 2003. 16071 Ergonomics of human-system interaction -- Guidance on accessibility for human-computer interfaces. International Organization for Standardization,.
- JAHNSEN, R., VILLIEN, L., STANGHELLE, J. K. & HOLM, I. 2003a. Fatigue in adults with cerebral palsy in Norway compared with the general population.
- JAHNSEN, R., VILLIEN, L., STANGHELLE, J. K. & HOLM, I. 2003b. Fatigue in adults with cerebral palsy in Norway compared with the general population. *Developmental Medicine & Child Neurology*, 45, 296-303.
- JEFFRIES, R., ALTHEA, T., POLSON, P. & ATWOOD, M. 1981. The Process Involved in Designing Software. *In:* ANDERSON, J. (ed.) *Cognitive Skills and Their Acquisition.* Hillsdale, New Jersey USA.
- JOHNSON, J. M., INGLEBRET, E., JONES, C. & RAY, J. 2006. Perspectives of speech language pathologists regarding success versus abandonment of AAC. *AAC: Augmentative & Alternative Communication*, 22, 85-99.
- JOKELA, T., IIVARI, N., MATERO, J. & KARUKKA, M. 2003. The standard of usercentered design and the standard definition of usability: analyzing ISO 13407 against ISO 9241-11. Proceedings of the Latin American conference on Humancomputer interaction. Rio de Janeiro, Brazil: ACM.
- JUDGE, S. L. 2001. Computer Applications in Programs for Young Children with Disabilities: Current Status and Future Directions. *Journal of Special Education Technology*, 16, 29-40.
- KENT-WALSH, J. & MCNAUGHTON, D. 2005. Communication Partner Instruction in AAC: Present Practices and Future Directions. *Augmentative and Alternative Communication*, 21, 195 - 204.
- KIENTZ, J., PATEL, S., TYEBKHAN, A., GANE, B., WILEY, J. & ABOWD, G. 2006. Where's my stuff?: design and evaluation of a mobile system for locating lost items for the visually impaired. *Proceedings of the 8th international ACM SIGACCESS conference on Computers and accessibility*. Portland, Oregon, USA: ACM.
- KINCHIN, I. M., HAY, D. B. & ADAMS, A. 2000. How a qualitative approach to concept map analysis can be used to aid learning by illustrating patterns of conceptual development. *Educational Research*, 42, 43-57.
- KITZINGER, J. 1994. The methodology of Focus Groups: the importance of interaction between research participants. *Sociology of Health & Illness*, 16, 103-121.
- KITZINGER, J. 1995. Qualitative Research: Introducing focus groups. *BMJ*, 311, 299-302.
- KRUEGAR, R. A. & CASEY, M. A. 2009. Focus Groups A Practical Guide for Applied Research, United States of America, SAGE Publications.
- KRUEGER, R. A. & CASEY, M. A. 2009. Focus Groups A Practical Guide for Applied Research, California, United States of America, SAGE Publications.

- KURNIAWAN, S., MAHMUD, M. & NUGROHO, Y. 2006. A study of the use of mobile phones by older persons. *CHI '06 extended abstracts on Human factors in computing systems*. Montreal, Quebec, Canada: ACM.
- LARRAZ, C. & ESCOIN (EDS), J. ISAAC Barcelona 2010. ISAAC, 2010 Barcelona, Spain.
- LARSSON, M., SANDBERG, A. & SMITH, M. 2009. Reading and Spelling in Children with Severe Speech and Physical Impairments. *Journal of Developmental and Physical Disabilities*, 21, 369-392.
- LAZAR, J., FENG, J. H. & HOCHHEISER, H. 2010. Research Methods in Human-Computer Interaction, United Kingdom, Wiley.
- LEVESON, N. 1995. Software: System Safety and Computers, Washington, Addison Wesley.
- LEVESON, N. G. & TURNER, C. S. 1993. An investigation of the Therac-25 accidents. *Computer*, 26, 18-41.
- LICHTNER, V., KOUNKOU, A. P., DOTAN, A., JOS\, \#233, KOOKEN, P. & MAIDEN, N. A. M. 2009. An online forum as a user diary for remote workplace evaluation of a work-integrated learning system. *Proceedings of the 27th international conference extended abstracts on Human factors in computing systems.* Boston, MA, USA: ACM.
- LIGHT, J. & DRAGER, K. 2007. AAC technologies for young children with complex communication needs: State of the science and future research directions. *Augmentative and Alternative Communication*, 23, 204 216.
- LIGHT, J., PAGE, R., CURRAN, J. & PITKIN, L. 2007. Children's ideas for the design of AAC Assistive technologies for young children with complex communication needs. *Augmentative and Alternative Communication*, 23, 274-287.
- LIGHT, J., STOLTZ, B. & MCNAUGHTON, D. 1996. Community-based employment: experiences of adults who use AAC.
- LUNDALV, M., BENGTSSON, M., BLOM, S., GUSTAFSSON, A. & WIKBERG, M. 1998. Experiences of user involvement in development and services concerning IT based AAC aids. *Improving the Quality of Life for the European Citizen*, 4, 143-146.
- LUNDALV, M., LYSLEY, A., HEAD, P. & HEKSTRA, D. 1999. ComLink, an open and component based development environment for communication aids. *Assistive Technology on the Threshold of the New Millennium*, 6, 174-179.
- MA, M.-Y., WU, F.-G. & CHANG, R.-H. 2007. A new design approach of usercentered design on a personal assistive bathing device for hemiplegia. Informa Healthcare.
- MACDONALD, B. K., COCKERELL, O. C., SANDER, J. W. A. S. & SHORVON, S. D. 2000. The incidence and lifetime prevalence of neurological disorders in a prospective community-based study in the UK. *Brain*, 123, 665-676.
- MACTAVISH, J., LUTFIYYA, Z. & MAHON, M. J. 1998. "I can speak for myself": Involving individuals with a mental disability as research participants. *Journal* of Mental Retardation, 38, 216-227.
- MAGUIRE, M. 2001. Context of use within usability activities. Academic Press, Inc.
- MAO, J.-Y., VREDENBURG, K., SMITH, P., W. & CAREY, T. 2005. The state of user-centered design practice. ACM.
- MATTHEWS, T., FONG, J. & MANKOFF, J. 2005. Visualizing non-speech sounds for the deaf. *Proceedings of the 7th international ACM SIGACCESS conference on Computers and accessibility.* Baltimore, MD, USA: ACM.
- MAUTHNER, M., BIRCH, M., JESSOP, J. & MILLER, T. 2002. *Ethics in qualitative research*, London, UK, SAGE.

- MAYER-JOHNSON, D. 2010. DynaVox Eyemax. Available: http://uk.dynavoxtech.com/download.ashx?FileId=147&DocId=2acae0f9-1677-4f7c-bd8e-a1e596796d22 [Accessed 3 February 2011].
- MCCALL, F., MARKOVÁ, I., MURPHY, J., MOODIE, E. & COLLINS, S. 1997. Perspectives on AAC systems by the users and by their communication partners.
- MCCOY, K., F., DEMASCO, P., PENNINGTON, C., A. & BADMAN, A. L. 1997. Some interface issues in developing intelligent communication aids for people with disabilities. *Proceedings of the 2nd international conference on Intelligent user interfaces.* Orlando, Florida, United States: ACM.
- MCGRENERE, J., DAVIES, R., FINDLATER, L., GRAF, P., KLAWE, M., MOFFATT, K., PURVES, B. & YANG, S. 2003. Insights from the aphasia project: designing technology for and with people who have aphasia. *Proceedings of the 2003 conference on Universal usability*. Vancouver, British Columbia, Canada: ACM.
- MCNAUGHTON, D. & BRYEN, D. N. 2007. AAC technologies to enhance participation and access to meaningful societal roles for adolescents and adults with developmental disabilities who require AAC. *Augmentative and Alternative Communication*, 23, 217 229.
- MCNAUGHTON, D., LIGHT, J. & ARNOLD, K. B. 2002a. "Getting your wheel in the door": successful full-time employment experiences of individuals with cerebral palsy who use augmentative and alternative communication. *AAC: Augmentative & Alternative Communication*, 18, 59-76.
- MCNAUGHTON, D., LIGHT, J. & ARNOLD, K. B. 2002b. "Getting your wheel in the door": successful full-time employment experiences of individuals with cerebral palsy who use Augmentative and Alternative Communication. Augmentative and Alternative Communication, 18, 59 - 76.
- MCNAUGHTON, D., LIGHT, J. & GROSZYK, L. 2001a. "Don't give up": employment experiences of individuals with amyotrophic lateral sclerosis who use augmentative and alternative communication. AAC: Augmentative & Alternative Communication, 17, 179-195.
- MCNAUGHTON, D., LIGHT, J. & GROSZYK, L. 2001b. "Don't give up": Employment experiences of individuals with amyotrophic lateral sclerosis who use augmentative and alternative communication. *Augmentative and Alternative Communication*, 17, 179 - 195.
- MCNAUGHTON, D., RACKENSPERGER, T., BENEDEK-WOOD, E., KREZMAN, C., WILLIAMS, M. B. & LIGHT, J. 2008. "A child needs to be given a chance to succeed": parents of individuals who use AAC describe the benefits and challenges of learning AAC technologies. AAC: Augmentative & Alternative Communication, 24, 43-55.
- MCNAUGHTON, D., SYMONS, G., LIGHT, J. & PARSONS, A. 2006. "My dream was to pay taxes": the self-employment experiences of individuals who use augmentative and alternative communication. *Journal of Vocational Rehabilitation*, 25, 181-196.
- MEDLINE PLUS. 2009. Speech and Communication Disorders [Online]. National Library of Medicine,. [Accessed 08-12-09 2009].
- MIESENBERGER, K., KLAUS, J., ZAGLER, W., WU, T.-F., MENG, L.-F., WANG, H.-P., WU, W.-T. & LI, T.-Y. 2002. Computer Access Assessment for Persons with Physical Disabilities: A Guide to Assistive Technology Interventions. *Computers Helping People with Special Needs*. Springer Berlin / Heidelberg.
- MILLAR, S. 1997. Personal Communication Passports. SENSE Conference. University of Dundee.

- MILLAR, S. & KERR, J. 1998. Augmentative Communication and Literacy the SAIL Kit Approach. In: WILSON, A. (ed.) Augmentative Communication in Practice: an Introduction. University of Edinburgh: Call Centre.
- MILLIKIN, C. 1997. Symbol Systems and Vocabulary Selection Strategies. *In:* GLENNEN, S. & DECOSTE, D. (eds.) *Handbook of Augmentative and Alternative Communication*. California, USA: Singular Publishing Group.
- MISERICORDIA UNIVERSITY. 2008. Augmentative and Alternative Communication (AAC) [Online]. Available: <u>http://www.misericordia.edu/misericordia_pg_sub.cfm?sub_page_id=3427&sub</u> cat_id=108&page_id=321 [Accessed 31st January 2011 2011].
- MOLICH, R., LAUREL, B., SNYDER, C., QUESENBERY, W. & WILSON, C. E. 2001. Ethics in HCI. *CHI '01 extended abstracts on Human factors in computing systems*. Seattle, Washington: ACM.
- MORGAN, D. L. & KRUEGER, R. A. 1993. *Successful Focus Groups*, California, United States of America, SAGE.
- MULROW, C. D. 1994. Systematic Reviews: Rationale for systematic reviews.
- MUNDY, P., KASARI, C., SIGMAN, M. & RUSKIN, E. 1995. Nonverbal Communication and Early Language Acquisition in Children With Down Syndrome and in Normally Developing Children. J Speech Hear Res, 38, 157-167.
- MUNSON, R. & MUNSON, M. 2000. Intervention and reflection: basic issues in medical ethics, California, United States of America, Wadsworth Publishing Co Inc.
- MURPHY, J. 2006. Perceptions of communication between people with communication disability and general practice staff. *Health Expectations*, 9, 49-59.
- MURPHY, J., MARKOVA, I., COLLINS, S. & MOODIE, E. 1996. AAC systems*: obstacles to effective use. *International Journal of Language & Communication Disorders*, 31, 31 44.
- MURPHY, K. P., MOLNAR, G. E. & LANKASKY, K. 2000. Employment and social issues in adults with cerebral palsy. *Archives of Physical Medicine and Rehabilitation*, 81, 807-811.
- MURRAY, R. 2008. Assitive Technology programme team. Cambourne: NHS.
- NEWELL, CARMICHAEL, A., MORGAN, M. & DICKINSON, A. 2006a. The use of theatre in requirements gathering and usability studies. *Interactions in Computing.*, 18, 996-1011.
- NEWELL, A. & GREGOR, P. 1999. Extra-Ordinary Human–Machine Interaction: What can be Learned from People with Disabilities? *Cognition, Technology & Work,* 1, 78-85.
- NEWELL, A. & GREGOR, P. 2000. "User sensitive inclusive design"; in search of a new paradigm. *Proceedings on the 2000 conference on Universal Usability*. Arlington, Virginia, United States: ACM.
- NEWELL, A., MORGAN, M. & GREGOR, P. 2006b. The use of theatre in AAC research. 12th Biennial Conference of the International Society for Augmentative and Alternative Communication. Germany.
- NEWELL, A. F., CARMICHAEL, A., MORGAN, M. & DICKINSON, A. 2006c. The use of theatre in requirements gathering and usability studies. *Interactions in Computing.*, 18, 996-1011.
- NIELSEN, J. 1992. The usability engineering life cycle. Computer, 25, 12-22.
- NIELSEN, J. 1993. Usability Engineering (Interactive Technologies), San Francisco, CA, Morgan Kaufmann.
- NIELSEN, J., CLEMMENSEN, T. & YSSING, C. 2002. Getting access to what goes on in people's heads?: reflections on the think-aloud technique. *Proceedings of*

the second Nordic conference on Human-computer interaction. Aarhus, Denmark: ACM.

- NIELSEN, J. & MOLICH, R. 1990. Heuristic evaluation of user interfaces. *Proceedings* of the SIGCHI conference on Human factors in computing systems: Empowering people. Seattle, Washington, United States: ACM.
- NØRGAARD, M. & HORNBÆK, K. 2006. What do usability evaluators do in practice?: an explorative study of think-aloud testing. *Proceedings of the 6th conference on Designing Interactive systems*. University Park, PA, USA: ACM.
- NORMAN, D., A. 2010. Living with Complexity, Cambridge, USA, The MIT Press.
- NUSEIBEH, B. & EASTERBROOK, S. 2000. Requirements engineering: a roadmap. *Proceedings of the Conference on The Future of Software Engineering*. Limerick, Ireland: ACM.
- O'CONNOR, C., FITZPATRICK, G., BUCHANNAN-DICK, M. & MCKEOWN, J. 2006. Exploratory prototypes for video: interpreting PD for a complexly disabled participant. *Proceedings of the 4th Nordic conference on Human-computer interaction: changing roles.* Oslo, Norway: ACM.
- O'KEEFE, B. M., KOZAK, N. B. & SCHULLER, R. 2007. Research priorities in augmentative and alternative communication as identified by people who use AAC and their facilitators. *AAC: Augmentative & Alternative Communication*, 23, 89-96.
- OFFICE FOR NATIONAL STATISTICS 2011. The 2011 Census. In: GOV, D. (ed.). London.
- OLIVER, M. 2002. Emancipatory Research: A Vehicle for Social Transformation or Policy Development. *1st Annual Disability Research Seminar*. The National Disability Authority and The Centre for Disability Studies, University College Dublin.
- OLSSON, E. 2004. What active users and designers contribute in the design process. *Interacting with Computers*, 16, 377-401.
- OSBERGER, M. J., MASO, M. & SAM, L. K. 1993. Speech Intelligibility of Children With Cochlear Implants, Tactile Aids, or Hearing Aids.
- PAGULAYAN, R. J., KEEKER, K., WIXON, D., ROMERO, R. L. & FULLER, T. 2003. User-centered design in games. *In:* JULIE, A. J. & ANDREW, S. (eds.) *The human-computer interaction handbook.* L. Erlbaum Associates Inc.
- PANEK, P., ZAGLER, W. L. & BECK, C. 1999. Development and demonstration of training and support for ECS and AAC systems using telematic links. Assistive Technology on the Threshold of the New Millennium, 6, 62-67.
- PARETTE, H. & ANGELO, D. 1996. Augmentative and Alternative Communication Impact on Families: Trends and Future Directions.
- PARETTE, H., BROTHERSON, M. & HUER, M. 2000. Giving Families a Voice in Augmentative and Alternative Communication Decision-Making. *Education and Training in Mental Retardation and Developmental Disabilities*, 25, 177-190.
- PARSONS, S., DANIELS, H., PORTER, J. & ROBERTSON, C. 2006. The use of ICT by adults with learning disabilities in day and residential services. *British Journal of Educational Technology*, 37, 31-44.
- PATTON, J. 2002. Hitting the target: adding interaction design to agile software development. *OOPSLA 2002 Practitioners Reports*. Seattle, Washington: ACM.
- PAUL-BROWN, D. & DIGGS, C. 1993. Recognizing and treating speech and language disabilities. *Journal of Speech and Hearing Disorders*, 43, 140-148.
- PETRIE, H., HAMILTON, F., KING, N. & PAVAN, P. 2006. Remote usability evaluations With disabled people. *Proceedings of the SIGCHI conference on Human Factors in computing systems*. Montr\&\#233;al, Qu\&\#233;bec, Canada: ACM.

- PREECE, J. 1993. A guide to usability human factors in computing, Addison Wesley Publishing Company.
- PRIOR, S. 2010. Involving Adults with Complex Communication Needs in the Design of CHAMPION. ACM SIGCHI Conference on Human Factors in Computing Systems. Atlanta, USA.
- PRIOR, S., CUMMINS, K., WALLER, A., KROLL, T. & BALANDIN, S. 2009. Facilitating communication of people with complex communication needs in hospital: A review of the literature. NHS Scotland Event 2009. Glasgow, Scotland.
- PRIOR, S., WALLER, A. & KROLL, T. Accepted. Focus Groups as a Requirements Gathering Method with Adults with Severe Speech and Physical Impairments. *Behaviour & Information Technology*.
- PRIOR, S., WALLER, A. & KROLL, T. In Prep. "Thank you for listening" Methods for including participants with Complex Communication Needs in Focus Group Requirements Gathering.
- PRIOR, S., WALLER, A., KROLL, T., MORGAN, M. & CUMMINGS, K. 2010. Including Adults with Complex Communication Needs as Actors in Forum Theatre. *ISAAC*. Barcelona, Spain.
- QUESENBERY, W. 2000. UPA 99 Workshop Report Crossing the Chasm-Promoting Usability in the Software Development Community. UPA Common Ground, 10.
- RACKENSPERGER, T., KREZMAN, C., MCNAUGHTON, D., WILLIAMS, M. B. & D'SILVA, K. 2005. "When I first got it, I wanted to throw it off a cliff": the challenges and benefits of learning AAC technologies as described by adults who use AAC. AAC: Augmentative & Alternative Communication, 21, 165-186.
- RATCLIFFE, A., CRESS, C. & SOTO, G. 1997. Communication isn't just talking. What's new in AAC. ASHA, 39, 30-6.
- RATE ENHANCEMENT. 2009. Augmentative and Alternative Communication at the University of Washington [Online]. Seattle. Available: <u>http://depts.washington.edu/augcomm/02_features/04d_rateenhance.htm</u> [Accessed 20 April 2009 2009].
- REDMOND, S. M. & JOHNSTON, S. S. 2001. Evaluating the Morphological Competence of Children With Severe Speech and Physical Impairments.
- RENNEKAMP, R. A. & NALL, M. A. 2000. Using Focus Groups in Program Development and Evaluation. Lexington: University of Kentucky.
- RESNA. Rehabilitation Engineering and Assistive Technology Society of North America. RESNA 2011, 2011 Toronto, Canada.
- RETTIG, M. 1994. Prototyping for tiny fingers. ACM.
- RICE, M., NEWELL, A. & MORGAN, M. 2007. Forum Theatre as a requirements gathering methodology in the design of a home telecommunication system for older adults. *Behaviour & Information Technology*, 26, 323 331.
- RIEMER-REISS, M. L. & WACKER, R. 2000. Factors Associated with Assistive Technology Discontinuance among Individuals with Disabilities. *The Journal of Rehabilitation*, 66.
- RORY, C. 2009. Using information technology to assist people with disabilities. Proceedings of the 11th international ACM SIGACCESS conference on Computers and accessibility. Pittsburgh, Pennsylvania, USA: ACM.
- ROSENBAUM, S., COCKTON, G., COYNE, K., MULLER, M. & RAUCH, T. 2002. Focus groups in HCI: wealth of information or waste of resources? *CHI '02 extended abstracts on Human factors in computing systems*. Minneapolis, Minnesota, USA: ACM.

- ROSS, J. 2010. *Recruiting Better Research Participants* [Online]. Available: <u>http://uxmatters.com/mt/archives/2010/07/recruiting-better-research-participants.php</u> [Accessed 8th March 2011 2011].
- ROY, D. P., MARILYN, HARWIN, W. & FAWCUS, R. 1993. The Enhancement of Interaction for People with Severe Speech and Physical Impairment through the Computer Recognition of Gesture and Manipulation. 1993 Virtual Reality Conference. California, USA.
- RUDD, J., STERN, K. & ISENSEE, S. 1996. Low vs. high-fidelity prototyping debate. *interactions*, 3, 76-85.
- RYOUNG, K., SONA, S. K., YOUN-KYUNG, L. & MYUNG-SUK, K. 2009. Focus group interview for designing a growing robot. *Proceedings of the 4th ACM/IEEE international conference on Human robot interaction*. La Jolla, California, USA: ACM.
- SCHERER, M. 2001. Assistive Technology: Matching Device and Consumer for Successful Rehabilitation Washington, USA.
- SCHLOSSER, R. 1999. An introduction to consumer perspectives in AAC research: Approaches, techniques and their theorectical underpinnings. *In:* LONCKE, F., CLIBBENS, H., ARVIDSON, H. & LLOYD, L. (eds.) Augmentative and alternative communication: New directions in research and practice. London: Whurr Publishers.
- SCOTT, J. 1998. Low Tech Methods of Augmentative Communication. *In:* CENTRE, C. (ed.) *Augmentative Communication in Practice: Scotland.*
- SHARP, H., ROGERS, Y. & PREECE, J. 2007. Interaction Design: beyond humancomputer interaction, Chichester, John Whiley & Sons Ltd.
- SHNEIDERMAN, B. 1998. Designing the user interface, Boston, USA, Addison-Wesley.
- SIGAFOOS, J. & IACONO, T. 1993. Selecting augmentative communication devices for persons with severe disabilities: Some factors for educational teams to consider.
- SIGCHI, A. Proceedings of the 28th international conference on Human factors in computing systems. *In:* MYNATT, E., ed., 2010 Atlanta, Georgia USA. 2644.
- SMALL, J., SCHALLAU, P., BROWN, K. & APPLEYARD, R. 2005. Web accessibility for people with cognitive disabilities. *CHI '05 extended abstracts on Human factors in computing systems*. Portland, OR, USA: ACM.
- SMITH-LEWIS, M. R. & FORD, A. 1987. A user's perspective on augmentative communication. *Augmentative and Alternative Communication*, 3, 12 17.
- SREEMANI, T. & ATLEE, J. M. Feasibility of model checking software requirements: a case study. Computer Assurance, 1996. COMPASS '96, 'Systems Integrity. Software Safety. Process Security'. Proceedings of the Eleventh Annual Conference on, 17-21 Jun 1996 1996. 77-88.
- STERIADIS, C., E. & CONSTANTINOU, P. 2003. Designing human-computer interfaces for quadriplegic people. ACM.
- SUTHERLAND, D. E., GILLON, G. G. & YODER, D. E. 2005. AAC use and service provision: A survey of New Zealand speech-language therapists. *Augmentative and Alternative Communication*, 21, 295 307.
- TAYLOR, C. 1983. Operation Elvis. Live Theatre. New Hampshire: Methuen.
- TAYLOR, T., VANDYK, G., FUNK, L. W., HUTCHEON, R. M. & SCHRIBER, S. O. 1983. Therac 25: A New Medical Accelerator Concept. *Nuclear Science, IEEE Transactions on*, 30, 1768-1771.
- TEE, K., MOFFATT, K., FINDLATER, L., MACGREGOR, E., MCGRENERE, J., PURVES, B. & FELS, S., S 2005. A visual recipe book for persons with

language impairments. *Proceedings of the SIGCHI conference on Human factors in computing systems*. Portland, Oregon, USA: ACM.

- VAN DE SANDT-KOENDERMAN, M. 2004. High-tech AAC and aphasia: Widening horizons? *Aphasiology*, 18, 245-263.
- VAUGHN, S., SCHUMM, J. S. & SINAGUB, J. M. 1996. Focus group interviews in education and psychology, California, United States of America, SAGE.
- VIRZI, R. 1992. Refining the Test Phase of Usability Evaluation: How Many Subjects Is Enough? *Human Factors: The Journal of the Human Factors and Ergonomics Society*, 34, 457-468.
- VISSER, N., ALANT, E. & HARTY, M. 2008. Which Graphic Symbols do 4-Year-Old Children Choose to Represent Each of the Four Basic Emotions? *Augmentative and Alternative Communication*, 24, 302 - 312.
- VREDENBURG, K., MAO, J.-Y., SMITH, P. W. & CAREY, T. 2002. A survey of user-centered design practice. *Proceedings of the SIGCHI conference on Human factors in computing systems: Changing our world, changing ourselves.* Minneapolis, Minnesota, USA: ACM.
- WALLER, A., BALANDIN, S., O'MARA, D. & JUDSON, A. 2005a. Training AAC Users in User-Centred Design. *Accessible Design in the Digital World*. Dundee, Scotland.
- WALLER, A., BLACK, R., O'MARA, D., PAIN, H., RITCHIE & MANURUNG 2009. Evaluating the STANDUP Pun Generating Software with Children with Cerebral Palsy. ACM.
- WALLER, A., O'MARA, D., MANURUNG, R., PAIN, H. & RITCHIE, G. 2005b. Facilitating user feedback in the design of a novel joke generation system for people with severe communication impairment. *HCII*. Las Vegas, USA.
- WESTERLUND, B., LINDQVIST, S., MACKAY, W. & SUNDBLAD, Y. 2003. Codesign methods for designing with and for families.
- WILLSON, P., BARBOUR, R., GRAHAM, C., LYNCH, M., MINNIS, H. & PUCKERING, C. 2005. Health Visitors and the Parent/Child Relationship: A Focus Group Study. *Journal of Epidemiology & Community Health*, 59, A6-A7.
- WING, L. 1981. Language, Social, and Cognitive Impairments in Autism and Severe Mental Retardation. *Journal of Autism and Developmental Disorders*, 11, 31-44.
- WOOD, D. L., KANTOR, D., EDWARDS, L. & JAMES, H. 2008. Health Care Transition for Youth with Cerebral Palsy. Northeast Florida Medicine, 59, 44-48.
- WOODS, D., PATTERSON, E., CORBAN, J. & WATTS, J. 1996. Bridging The Gap Between User-Centered Intentions And Actual Design Practice. *Human Factors* and Ergonomics Society Annual Meeting Proceedings, 40, 967-971.
- WRESSLE, E. & SAMUELSSON, K. 2004. User Satisfaction with Mobility Assistive Devices.
- YOUNG, N., L., GILBERT, T., K. , MCCORMICK, A., AYLING-CAMPOS, A., BOYDELL, K., LAW, M., FEHLINGS, D., L., MUKHERJEE, S., WEDGE, J., H. & WILLIAMS, J., I. 2007. Youth and Young Adults With Cerebral Palsy: Their Use of Physician and Hospital Services. *Archives of physical medicine and rehabilitation*, 88, 696-702.
- ZINN, C. 1995. 14000 preventable deaths in Australian hospitals. BMJ, 310, 1487-.



Presented at ACM CHI Conference 2010

HCI Methods for Including Adults with Disabilities in the Design of CHAMPION

Suzanne Prior

School of Computing, University of Dundee Queen Mother Building, Dundee, UK, DD1 4HN sprior@computing.dundee.ac.uk Abstract

The demand for software, suitable for users with complex communication needs and other disabilities, is increasing. However, traditional HCI design methods are not always suitable for these users. To address this, the CHAMPION project is piloting adapted methods in the development of a patient hospital profile for this user group. Initial results show that users with cognitive and communication disabilities can be involved in participatory design. The challenge is now to develop meaningful evaluation methods for this group.

Keywords

Disability, usability, inclusive design, methodology General Terms Design, Experimentation

ACM Classification Keywords

D.2 SOFTWARE ENGINEERING D.2.10 Design: Methodologies

Copyright is held by the author/owner(s). *CHI 2010*, April 10–15, 2010, Atlanta, Georgia, USA. ACM 978-1-60558-930-5/10/04.

Introduction

Adults with Complex Communication Needs (CCN) have been overlooked as research participants in the past. The CHAMPION project is investigating the potential to include participants with Complex Communication Needs and cognitive impairment in the design of a new health software program.

Working with these adults in participatory design is challenging; traditional HCI methods are not always suitable due to communication problems and the cognitive load involved for the participants using them. Using an AAC device to communicate is time consuming and requires a great deal of concentration by a communication partner. In addition to this, participants who are unused to taking part in any form of research need to be encouraged in providing feedback on systems.

Have you read and understool the Participant Information Sheet? Have you beengivenan opportunity to ask que tions and further discuss this study? Have you necessical safactory assesses to all of your questions? Have you received enough information ab out this study?	Yes () Yes ()No () Yes () Yes ()	No 🗆 No 🗆 No 🗆
Who have you spoken to? Dr/Mr:Mriss		
Do you understand that you zimovbeenent in the project is entrely voluntary? Do you understand that you are free to withdraw from this study at any time? Do you understand that you do not have to give a reason for withdrawing? Do you understand sturithdrawing with directly your present of thus embdical case? Do you agree to take partin this study? If <u>any</u> of the anovers are 'no' oryon don't want to take part, don't continue with the for	Yes Yes Yes Yes Yes m	No 🗆 No 🗆 No 🗆 No 🗆

Figure 1 Adapted Informed Consent Form. Participants were asked a series of multiple choice questions to which they could answer yes or no, such as "Can you withdraw at any point?", to ensure they had understood their rights in the study Within the area of Augmentative and Alternative Communication (AAC) and Complex Communication Needs (CCN), there is has been some use of User Centred Design (UCD). Alm et al., (Alm et al., 1995) included adults with CCN in focus groups which provided ideas which were fed into the software design process. Other projects have used adults with CCN to evaluate early stage paper prototypes (Waller et al., 2005b). But, to the best of our knowledge, little UCD work has been done with adults with CCN who may also have cognitive impairments.

Importance of Work

The number of computer users with cognitive and developmental disabilities is increasing e.g. day and residential units are now including computer courses as part of the program for service users (Parsons et al., 2006). As adults with developmental disabilities become more integrated into the workplace, the demand for computer software which is accessible by a wider spectrum of users is likely to increase.

However, the main use of technology by this user group remains in assistive technology and AAC devices. As with other user groups, a lack of UCD may contribute to the poor adoption of technology. The rate of abandonment of AAC devices is reported to be as high as 53.3% (Riemer-Reiss and Wacker, 2000). It is suggested that the inclusion of end users in the design process may reduce this abandonment rate (Waller et al., 2005b). While there is literature on developing UCD methods for older and disabled users together, there is little research into developing UCD methodologies for adults with CCN who may also have cognitive impairment (Waller et al., 2005a).

Adapted HCI Methods

Traditional HCI methods such as focus groups, design sessions and rapid prototyping were adapted and piloted with a group of adults with CCN and learning disabilities from a local disability day centre.

Informed Consent: As the adults with moderate to profound communicate impairment were unlikely to be able to consent through traditional channels, a modified consent process was followed (see Figure 1).

Focus Group: 6 adults with CCN (3 males and 3 females) took part in a focus group looking at the issues surrounding hospitalization and the information they would like to share with hospital staff. The focus group was held in the day centre,



Figure2 – Layout of Focus Group. As all of the participants used motorized wheelchairs a semi circle was formed with facilitators placed next to the participants who required the most help with their communication. The moderator sat at the head of the circle with the video recorder placed to the left of the moderator to ensure all the participants were captured. One participant had a hearing difficulty in

in a room used for activity sessions (Figure 2). Participants were provided with questions in advance to allow those who rely upon AAC devices to prepare longer answers in advance.

Requirements Gathering:

3 participants from the first focus group formed the design group along with a support worker. This group had never been involved in a design project before or in giving requirements for a system. They had limited computer experience beyond the assistive technology they used on a day to day basis. The first meeting of the design group involved a discussion using a storyboard showing how the CHAMPION patient profile system might be used in hospital. Once the participants were clear on its use, the author stepped through how the use of the system would be used to store information. At each stage of the discussion the participants were asked for their requirements at this point.

Design: A week later a second design meeting was held with paper mock ups of the user interface. Using magnets on the back of foam shapes representing buttons, text boxes and labels, the screen layouts were displayed on a large magnetic board. Using the magnetic board meant that participants in wheelchairs could have the 'screen' placed directly in front of them.

The design process then moved onto a medium level prototype. Clicker5 (Crick Software, 2009), usually used as a way of creating writing tools for children with disabilities, was used for the prototype. Its interactivity options were harnessed to create prototypes in a similar manner to PowerPoint. The main advantages of Clicker5 over PowerPoint for this group are its ability to be controlled by assistive technology peripherals that use scanning; and its voice output of the items on the screen. During the evaluation of the Clicker5, prototype participants chose 3 different peripheral options: scanning; mouse and keyboard; and touch screen. Participants were seen individually during this session to allow better observation of how each used the peripheral device.

Participants' responses to adapted methods

During the focus group participants engaged in discussion with each other around issues prompted by 3 present questions: 1.) The good experiences of hospital, 2.) The bad experiences of hospital and 3.) What would have made the bad experiences better? The moderator took care to ensure that the



Figure 3 Paper prototype of questionnaire screen

First of all I will ask you some questions about the kind of information you want hospital staff to know about you.
Do you have noises or gestures you use to communicate?
Yes No

Figure 4 Clicker5 development of screen in Figure 3 participants did not interrupt each other. The participants were patient in waiting for each other to either prepare an answer on an AAC device or to have their reply spoken by a facilitator.

the requirements session, the 3 In participants were able to envisage themselves using the system and were able to discuss the different needs they had for the system. When participants disagreed about the amount of support they would require when using an aspect of the system they were able to hold a discussion with each other and attempted to reach compromise on how best to meet one another's need. This is despite the fact that these participants are rarely involved in group discussions in their typical routine.

When working with the paper prototypes, participants were able to adapt quickly to imagining the results that could occur from selecting a button or pressing a text box (Figure). As they progressed through the paper prototypes participants provided useful suggestions on better ways of providing instructions or how things should be laid out. Participants agreed that working with paper designs initially reduced their stress about making mistakes and helped them to feel more confident in making suggestions on alterations.

When using Clicker5 for evaluations participants responded well to seeing their ideas progress. Although the designs were based on discussion from earlier sessions there were differences which caused participants to be unsure at stages. Despite the more technical appearance, participants still appeared comfortable in pointing out features they did not like and providing useful suggestions for improvement.

Next Stages

The remainder of this thesis will focus on moving the designs into computer software. The challenge is to ensure continued participation from adults with complex disabilities as the project moves onto more technical stages. Unlike other software projects where one main peripheral is used for development, with others added as the project nears completion, the CHAMPION project will have to adapt the software for many different forms of assistive technology throughout its development if it is to be evaluated by participants with varieties of disability.

Another challenge for CHAMPION will be how to conduct meaningful evaluations with participants over a period of time. It is likely that participants will initially be keen to create a profile and store their stories and information on the system; however it is not yet clear how willing they and their support staff will be to maintain their personal records. It may be that additional motivational tools will have to be built into the system to maintain users' interest.

Expected Contributions

To summarize, the expected contributions of these adapted methods are:

(1). Understanding how users with complex communication needs and additional cognitive impairments can contribute meaningfully to the development of new and innovative software.

(2). Adaptation of HCI methodologies, better suited to the cognitive and physical needs of a group of users previously overlooked in design

(3). Increased awareness of the role adults with complex disabilities can play in the design of not only assistive technology but in technology designed for the general population.

The number of computer users with physical, cognitive and communication disabilities is

increasing. As more adults with disabilities gain employment in roles using computers the demand for software to be accessible to a broader range of people will increase. It is likely software designers will have to begin take on board the impact of users with disabilities when designing software. Many traditional HCI methodologies are not suitable for adults with limited experience of taking part in research or design groups and who have a range of physical and cognitive disabilities. By addressing these issues adaptations can be made to traditional usability methods to ensure they are suitable for this group.

Acknowledgements

The author acknowledges her supervisors Dr Annalu Waller and Dr Thilo Kroll. This PhD study is funded by Capability Scotland (Scottish Charity No. SC011330).

Appendix B. Rationale for CHAMPION

Presented at 2009 NHS Scotland Conference

Young adults with SSPI, in particular those with cerebral palsy, are 10.6 times more likely to be admitted into hospital as an inpatient and 2.2 times more likely to attend an outpatient clinic than their peers with no disability (Young et al., 2007). There is also evidence of adults lifelong disabilities suffering from more health problems into adulthood (Hemsley, Balandin, & Togher, 2007). In addition to this there is increasing evidence that people with SSPI are at a higher risk of having illnesses, which go undiagnosed or untreated, in particular those with lifelong disabilities (Cumella & Martin, 2000). As adults with cerebral palsy and other lifelong conditions age, their needs and medical care required alter too, often increasing dramatically with age (Bakheit et al., 2001; Strauss, Ojdana, Shavelle, & Rosenbloom, 2004).

Unlike paediatric care where a patient is likely to have one primary care doctor at a hospital who takes responsibility for their care, adults with cerebral palsy are likely to have different doctors for each of their medical problems (Fitton, 1994). Communication is vital for a successful medical professional/patient relationship (Iezzoni, O'Day, Killeen, & Harker, 2004) with poor communication being associated with an elevated risk of suffering an avoidable error in care in hospital (Bartlett, Blais, Tamblyn, Clermont, & MacGibbon, 2008) or even death (Zinn, 1995). The ability to answer at a minimum yes or no to questions is recognised in medical settings (Uruma et al., 2007) to allow for basic communication, however, for an estimated 600,000 adults in the United Kingdom their SSPI is so profound that

they cannot do this without some form of Assisted or Augmentative Communication device (AAC) (Aldous, 2008).

Various supports are currently used in hospitals by either the medical staff or the patient and their support worker to help to alleviate the communication difficulties such as the constant presence of regular care givers (Hemsley & Balandin, 2004; Hemsley et al., 2007) communication passports (Gates, 2006; Great Ormond Street, 2008; Millar, 1997) and more effective training for nurses (Hardy, 2001). However their effectiveness has not been widely assessed.

As the number of adults with SSPI grows the need for their stays in hospital to be made more satisfactory with improved understanding of their needs and their methods of communication becomes more urgent (Hemsley et al., 2007).

A care book can provide medical staff with up to date information on how to treat the patient, and the information can be tailored to that particular patient's needs and methods of communication, unlike generic training which would only provide an overview of the treatment and communication methods of adults with SSPI. Research into the use of care books in residential care setting has shown promise (Millar, 1997). However these situations have the benefit of the reader having the time to find and read the information. Nurses complain that often too much information is contained within the files and that accessing the pertinent information takes too long (B. Hemsley, S. Balandin, & L. Togher, 2008b). The other issue is that frequently when these profiles have been created on paper they are stored in the adults' home. However when these patients have been admitted as emergencies they find problems in communicating to the paramedics the importance of taking this profile with them (Hemsley et al., 2008a).

Of the three main current methods of supports (Finke, Light, & Kitko, 2008), the use of a care book would appear to have the most easily addressable disadvantages. Electronic medical records (EMR) and electronic health records (EHR) (collectively known as Electronic Patient Records (EPR)) are being increasingly used in general practice surgeries and in hospitals, to the extent that a House of Common Health Committee report was commissioned to investigate their impact and best practice for their use (Great Britain Parliament House of Commons Health Committee, 2007). The report from the House of Commons Health Committee gathered feedback from various interest groups on the impact they foresaw the introduction of EPRs having on the groups they represent.

There is now growing evidence of the impact of these records both for patients and staff, as well as the problems and benefits being brought by these systems. The use of EMRs has been much debated by researchers over the past decade (Larum, Ellingsen, & Faxvaag, 2001; Morrison, Jones, Blackwell, & Vuylsteke, 2008; Pollak & Lorch, 2007; Willison, Keshavjee, Nair, Goldsmith, & Holbrook, 2003) however studies have focused primarily on the use of general EMRs which provide information on the medical records currently stored as paper documents.

Current EPRs contain information on allergies, drug reactions, current medication, basic medical history, key operations, physiological details and basic lifestyle details (Harrison, 2007) this information is largely created from the GP's records and is automatically formatted and inserted into a EPR. When doctors have had access to EMR, reviews have shown that their main use of it has been for reading patient records (Larum et al., 2001).

The advantages of EPRs come from the reduction in medical errors they lead to. Information written by hand can be difficult to read in emergency situations and the legibility of a person's handwriting can have a significant effect on the information obtained from a record (Evans, Nichol, & Perlin, 2006). EHRs have also been linked to an increase in patient education, reduction in the number of procedures which have to be carried out twice due to lost information, higher levels of quality control and an improvement in the efficiency of daily ward rounds, these improvements have been most significant when the development of the systems have involved front line medical professionals (Department of Veterans Affairs, 2006).

One of the major concerns of EPR is their security, medical staff and patients have asked questions and raised concerns over the security and safety of the data being stored (Harrison, 2007). While, in general, hospitals have still not reached the levels of computerisation that GP's practices have with some wards still operating entirely on paper records, clinical networks for specific patient groups across the country have seen the development of more advanced system which can share information between different hospitals (Harrison, 2007). The main criticisms of EMR are in regards to patient control over their data that is stored in the system (BBC, 2006), a review by the Government's Health Department into the use of EMR cited patient control over data as one of the major problems facing the implementation of EMRs (Harrison, 2007). In France this problem has been overcome through giving patients legal ownership of their EMR – in France known as Dossier Médicale Personnel (DMP) – and allowing them to have their own say in who can access the information - in addition a medical professional cannot use the record without the patient's explicit consent and in their presence (Harrison, 2007). Another criticism of the use

of the system has been the time taken to read the data electronically – reading on a screen is up to 40% slower than reading on paper (Larum et al., 2001) and the screens can be very confusing with a lot of information shown at once. This problem could be eased through more use of multimedia information.

Research has been done with EMRs displayed in 19inch monitors which were next to the patient's bedside (Morrison et al., 2008), these monitors could be adjusted and their height altered to suit the viewer. We are now at a point with technology where hospital networks could cope with storing large files including video and voice, and electronic records could allow the use of both text and audio files (Walsh, 2004).

In recent investigation by MENCAP the potential for multimedia patient profiles (MMP) to be created by service users and their care attendants and to be used for planning, introducing new staff to the users and enhancing existing staff's knowledge was explored (Cavet & Grove, 2005). The study used an initial group of 6 service users, although one later retired from the project. There were some concerns raised by the staff who assisted service users in creating their profiles such as time required, access to equipment and how they could ensure the service users were providing informed consent. However, the positives of the project were generally seen to outweigh the negatives: staff noted the service users enjoyed creating the profiles and that it had helped them to have a better understanding of how the service user communicates and their needs (Cavet & Grove, 2005). Staff reported only positive outcomes for the service users with no adverse effects. All participants benefited from improved staff knowledge of their needs and preferences, the majority of participants also benefited from improved communication with staff and from the additional activity that creating the profile allowed them.

In the MENCAP project, videos were made of sporadic events that took place with staff estimating that they spent on average 30mins a week profiling (Cavet & Grove, 2005). These videos provided new perspectives for those who are making decisions on behalf of the service users and gave the service user more self autonomy, allowing them to have a say in decisions affecting them. The success in this initial study has contributed towards Project Apple, which helps young adults withSevere Speech and Physical Impairments and other disabilities to create personal profiles incorporating multimedia, which are then used as a form of self advocacy when decisions regarding their futures are being made.

Facilitating communication of people with complex communication needs in hospital: A review of the literature

Suzanne Prior BSc., Kathleen Cummins MA, Annalu Waller PhD, Thilo Kroll PhD, Sue Balandin PhD

Did You Know?

- There are between 1.5 and 2.5 million people in the United Kingdom with a Complex Communication Need (CCN) [1] Þ
- 600,000 people in the UK are unable to communicate basic yes/no choices. [1] Þ
- Young adults with CCN are 10.6 times more likely to be admitted into hospital than their peers [2] Þ
- Adults with lifelong disabilities have more health problems [3] Þ
- People with CCN are at a higher risk of having illnesses which go undiagnosed or untreated [4].

Problems for patients with CCN in Hospital



Solutions From The Literature

	Information Interventions	Equipment Interventions	Training Interventions
Current Interventions	Detailed shift handover [7]		Persevering with communication [7, 9] Getting assistance from family [7]
Suggested Interventions	To identify patients with CCN on admission [10] To improve the flow of information [9, 10]	Inclusion of AAC devices on the ward [6]	Training for nurses [7, 9]

Current Research Projects at The University of Dundee

CHAMPION

The CHAMPION Project aims to use existing work done in multimedia [11,12] as a means of advocacy for people with CCN and to look at how best to incorporate these multimedia information profiles in a patient's existing Electronic Medical Record. This project is user centred, involving patients, their care workers and medical staff as stakeholders in the

design and evaluation of the system.

TEAACH

The TEAACH project aims to study the barriers and solutions to successful communication between patients with little or no functional speech and the nurses who care for them, and to develop a training module to overcome these communication barriers. This project is user centred, involving patients, their care workers and medical staff as stakeholders in the design and evaluation of the module.

What worked for us. The Trime. A Balandin S, Toghar L. Cider Unpaid Carwa' Experiences Supporting Ad L. Gilbert T, K. McCormick A, Ayling-Campos A, Boydel K, Law M, et al. iay and Complex Communication Needs in Hospital Journal of Developmental and Physical Daubilities. 2007;19(2):115-04. Julis With Censioni Palay: Their Use of Physician and Hospital Services. Archives of physical medicine and renabilitation. 2007;05(0):100

Hensiev R, Belandh S, Togher L. The got something to say: Interaction in a tocus group of ad-Hensiev R, Belandh PS, Togher L. We need to be the certageloor' Acuts with centeral pails, Mensiev R, Belandh PS, Togher L. We need to be the certageloor' Acuts with centeral pails.

- NAMES AND A DESCRIPTION OF A DESCRIPTION n impairment, Journal of Advanced Numing, 2001;35(6):82 atth Expectations: An international Journal Of Public Partic risk of preventable adverse events in ac surgery: a literature review. Cancer Num 15(2):5-7 artings Ca



Appendix C.Focus Groups as a Requirements Gathering Method withAdults with Severe Speech and Physical Impairments

Accepted for publication in the Journal of Behavioural and Information Technology

Focus Groups as a Requirements Gathering Method with Adults with Severe Speech and Physical Impairments

Technological supports have the potential to greatly improve the quality of life and independence of adults with Severe Speech and Physical Impairments (SSPI). In particular, Augmentative and Alternative Communication (AAC) devices can enable people with little or no speech to communicate with others. However, the rate of rejection of AAC devices is estimated to be as high as 53.3%. It is suggested that a major reason for this rejection is a lack of user centred design in the development of these devices. As part of a wider study looking at involving adults with SSPI in all stages of user centred design, this paper looks at the use of focus groups in requirements gathering with this user group.

Keywords: Disability, usability, inclusive design, accessibility, focus groups

INTRODUCTION

People with disabilities have frequently been excluded from both quantitative and qualitative research studies. Awareness is growing of how this quite heterogeneous group can be involved in and contribute to research (Oliver, 2002). One method used in HCI to gather opinions from potential users of a piece of technology is through focus groups. Focus groups can be used at any stage of the development process, to gather

requirements for a piece of technology, to comment on design ideas or to discuss the product following evaluations.

Focus Groups

Focus groups have been used as a method of market research since the 1920s; they have been used in wider research since the 1950s when 'focused interviews' were developed to uncover people's feelings and reactions to the propaganda used in the previous decade as part of the war effort (cited in Kitzinger, 1994). Focus groups are now used as a means of evaluating the experiences of people with health and social services and in action research projects (Kitzinger, 1994, Willson et al., 2005). Focus groups are also commonly used in requirements gathering to identify the requirements from different groups of people and discuss any conflicting wishes in the design. In Joint Application Development, focus groups are routinely used, and are very structured and prescribed (Sharp et al., 2007). Focus groups can be used at any stage of the development cycle; Kurniawan (2006) used them before the design and development had taken place or even been planned in order to uncover why older adults were reluctant to use existing mobile phone technology, whereas Kim (Ryoung et al., 2009, Murphy et al., 2000) used them at the design stage. The same advantages that apply to the use of focus groups in social science can apply when used in HCI.

Focus groups are useful in gathering many different views on a topic and insights into people's shared and different opinions (Gibbs, 1997). In a focus group the discussion between participants and sharing of views may lead to a greater disclosure and volunteering of information; it also allows a researcher (or designer) to discover why a participant holds a certain view (Gibbs, 1997). Participants may be encouraged or reminded of something they wish to share as they piggyback on each other's statements (Rennekamp and Nall, 2000).

A session will typically last 2 hours with a facilitator asking questions and encouraging people in their group. Traditional focus groups are composed of 4-8 participants who might have different opinions on an issue and will discuss these opinions during the course of the session; the other important feature of a traditional focus group is the moderator who will ensure that the group maintains the focus on the topic. Participants will usually sit in a circle or around a table.

Kitzinger (1994) states that focus groups should take their results from analysis of the interaction between participants and how they respond to the statements made by one another. This interaction allows the participants to explore other points of view and it is these differing points of view which provide a researcher with an overview of the issues and feelings surrounding the aspect under investigation (Cameron, 2005).

Focus groups are now becoming more common in HCI as can be seen by recent panels and workshops at conferences such as ACM CHI (Rosenbaum et al., 2002).

Severe speech and physical impairments(SSPI)

Communication impairment can be congenital (e.g. due to cerebral palsy) or acquired (as the result a brain or spinal injury from a car crash or stroke). Individuals may experience a physical inability to speak or be unable formulate the words needed for communication. Additional sensory, intellectual and/or language impairment, e.g. hearing impairment, aphasia or developmental delay, may also result in difficulties in cognitive processing and receptive communication.
Augmentative and Alternative Communication (AAC) is the general term for the methods used to aid communication by those for whom the usual form of communication through speech is not sufficient. AAC can be divided into the two different categories of unaided and aided. Most people who use AAC will use a combination of both (American Speech-Language-Hearing Association, 2002).

In unaided AAC systems, e.g., sign language, Makaton (Communication Matters, 2008) and facial expressions (Communication Matters, 2003), additional equipment is not used. These systems have the benefit that they can be used anywhere and the user does not need to carry additional equipment around. However unaided AAC has the disadvantage that the communication partner needs to understand how the system works.

Aided AAC systems refer to communication methods that involve a device – this may be electronic or paper based (e.g. a word board) – which is external to the user. Electronic AAC devices usually provide access to written or spoken output. Depending on the level of physical impairment, physical access to such technology may be restricted to one switch input.

Use of Computers by people with SSPI

The number of computer users with cognitive and developmental disabilities is increasing (Clayton, 2006); day and residential units provide computer courses as part of the program for service users (Parsons et al., 2006) and schools for children with disabilities have computer lessons in their basic curriculum (Judge, 2001). In addition, as adults with developmental disabilities become more integrated into the workplace, the demand for computer software which is accessible by a wider spectrum of users is likely to increase.

Although individuals with cognitive and developmental disabilities use general ICT, the main use of technology by this user group tends to be in assistive technology and AAC devices. However, there is a high rate of abandonment of AAC devices estimated at 33% (Johnson et al., 2006)The main reasons given for abandonment or rejection of assistive technology are a lack of training for the end user and a lack of flexibility in adapting the system for the individual. Other reasons reported include a lack of confidence on the part of the user in their ability to use the system and the prohibitive costs in learning how to use the system.

It has also been suggested that a lack of user centred design may contribute to the poor adoption of technology and that the inclusion of end users in the design process may reduce this abandonment rate (Waller et al., 2005a).

Initiatives, such as Inclusive Design (Newell and Gregor, 2000) (the development of items to be as usable as possible for as wide a group as possible), promote consideration of these groups in design. This move presents significant challenges to researchers including difficulties in obtaining informed consent, meeting the needs of different user groups and finding ways of communicating with the users in order to gain their requirements (Newell and Gregor, 2000). However, while there is literature on developing user centred design methods for older users who may have age-related disabilities, there is little research into developing user centred design methodologies for adults with SSPI with their varied physical needs and possible cognitive impairment (Waller et al., 2005a).

The aim of the CHAMPION project is to investigate ways of conducting user centred design with adults with SSPI. In order to do this, existing methods in user centred design were examined for their suitability to be used with this user group and where necessary

the methods were adapted to make them more suitable. One such method which is used by designers and developers in user centred design is focus groups.

Focus Groups with SSPI

Given that adults with SSPI have traditionally been excluded from software design and have limited experience of technology and of designing technology (Waller et al., 2005b), it would seem sensible to conduct focus groups with them when gathering requirements or evaluating design solutions. Focus groups could potentially empower these adults to offer critical feedback on requirements and design.

However a structured review of the literature using thesaurus terms for communication impairment and speech disability resulted in no mention of conducting focus groups with adults with SSPI within the software engineering process. A structured approach to searching the electronic databases, Medline, CINAHL, PsycInfo and Web of Science was used, involving Boolean combinations of free text word, indexed keyword and thesaurus terms for focus group methodology and communication impairment and speech disability. Following this an author search on recovered papers was conducted.

Inclusion criteria for this structured review were: (a) published in a peer reviewed journal between 1990-2009; (b) published in English; (c) included focus group used as the primary or one of the principal research methods; and (d) involved participants with SSPI. Excluded from the review were articles which incorporated focus groups as part of a larger process but did not report on the focus groups themselves.

Twelve (12) papers discuss holding focus groups with adults with SSPI, but these studies are restricted to the medical and social sciences fields.

The literature review shows that the use of focus group methodology with participants with SSPI has increased since 1999, prior to which there were no reports of any focus groups being conducted with participants with SSPI (Schlosser, 1999). At present there is no best practice for focus groups with participants with SSPI, with researchers altering traditional focus group methodology as per their individual research requirements.

The review also suggests that current focus group methodologies are not suitable for studies which aim to gather the views of a wide variety of participants with SSPI. Adaptations have been used in face to face focus groups through Talking Mats (Murphy, 2006) where participants are asked to place images related to topics under discussion in one of three groups to show whether they are happy, unhappy or indifferent to the topic. The impact this adaptation makes on the amount of conversation and the number of questions asked by participants is not reported.

Other adaptations that could be made to traditional focus groups to promote discussion between participants with SSPI might include providing questions ahead of time in order that participants with AAC devices could prepare longer answers in advance, and showing questions on a screen in the room in a symbol format that participants understand. This may help to include more participants with SSPI and intellectual disabilities; to date many focus groups with participants with SSPI have focused on participants at the upper end of the intellectual spectrum (e.g. Rackensperger et al., 2005, McNaughton et al., 2002a, McNaughton et al., 2006).

Online focus groups were used in 5 (42%) of the papers in the review. Online focus groups are able to solve many of the problems faced in holding traditional focus groups with participants with SSPI, such as the stress placed on participants to produce replies quickly with their AAC device and the ability to include participants from a wide

geographical area. This is particularly important when the pool of potential participants is small (McNaughton et al., 2002a, McNaughton et al., 2001a, McNaughton et al., 2006). However online focus groups are not without their own problems, holding the focus group online requires that participants are literate, able to produce messages in written form and comfortable in using the internet. Typically in online focus groups participants must also be willing to take part in the focus group over an extended period of time.

To date no focus groups have included a moderator with SSPI in face to face focus groups. The papers in this literature review made no reference to any training or preparation work done with moderators in preparing them for conducting a focus group with adults with SSPI. This may be due to the fact that the vast majority of moderators had experience as either speech pathologists in the past or had done extensive research previously with adults with SSPI. If a training program was available for all researchers in holding focus groups for adults with SSPI, it is possible that more researchers may feel confident in conducting these groups.

The analysis of focus groups, should not focus merely on the information provided by the participants to a set of questions, but should also concentrate on the interaction between participants and non verbal expressions and cues given from them (Catterall and Maclaran, 1997). All of the papers in this review used content themes (examination of transcript for recurring themes) as a means of analysis, but only 4 (33%) of the papers looked at the interaction between participants. Given the limited opportunities adults with SSPI may have for interaction with others the analysis of these aspects could provide useful insights to researchers into the participants' views and opinions.

Only one paper involved anyone with SSPI in the reporting of the focus group (Rackensperger et al., 2005) when they were included within the research group. With

conferences such as International Society for Augmentative and Alternative Communication (Larraz and Escoin (eds), 2010) now not only being accessible to adults with SSPI and other disabilities, but proactively encouraging their participation, there is now the opportunity for adults with SSPI to take an active role in the dissemination of findings.

Extra Considerations When Planning Focus Groups with Participants with SSPI

After an examination of the current state of the field for focus groups with adults with SSPI and in light of the lack of guidelines available, an examination of the general additional considerations when working with adults with SSPI was conducted.

Communication Considerations

People who use an AAC device for communication may produce words up to 25 times slower than those with normal verbal speech (Higginbotham et al., 2008a). The time therefore for a participant to reply, for example "Yes, that has happened to me" could take up to 2 minutes for a participant relying on AAC (Rate Enhancement, 2009). Another problem may exist when participants have a low tech AAC aid for which a facilitator is required to follow the participant's finger pointing and speak the message on their behalf. A participant with dysarthric speech may be able to produce words at the same rate as a participant without a SSPI; however the need for clarification of the response can slow down the rate at which the response is understood by the group (Hustad, 2006b).

Cognitive Considerations

It is a common misconception that people with SSPI will also automatically have a learning disability. This is not the case, while it is true that some will have a learning

disability which can range from mild to severe. However, it is important to consider the effect a participant's learning disability may have on their ability to participate in a focus group.

Literacy Levels

One issue concerning a participant's learning disability or cognitive impairment is the process of informed consent. Terms in consent forms and participant information sheets may contain technical and/or experimental terms or jargon which is unfamiliar to those without experience of participating in research projects (Dickinson et al., 2007), and confusing to those with a lower rate of literacy than those who traditionally participate in research projects, such as students.

Participants may be asked to take their own notes on certain discussions or be given 'prompts' to help with the discussion. It may be necessary to consider what if any prompts can be used with participants with SSPI.

Concentration Span

The other area of focus groups in which a participant's learning disability may have an effect is their ability to understand all that is being said in the conversation and the moderator's questions. A focus group typically will have a steady flow of conversation, argument and topic change (Krueger and Casey, 2009); this type of conversation may be difficult to follow for those with SSPI.

Physical Considerations

Focus Groups have traditionally been conducted around a table (Kruegar and Casey, 2009), with a moderator at one end of the table and possibly a moderator's assistant at the other who has the role of taking notes Participants with cerebral palsy, a common

cause of SSPI, are likely to have a range of motor impairments which may restrict their ability to sit at the table and/or may require them to use a wheelchair (Finnie and Bavin, 1997). As a focus group is primarily a study of the interaction between participants (Morgan and Krueger, 1993), the layout of a group requires careful consideration to ensure that participants are positioned in a way that encourages conversation and debate.

People with cerebral palsy are likely to become tired more quickly than the general population (Jahnsen et al., 2003b). This is attributed to "post-impairment syndrome", which means that those with cerebral palsy will use between 3 and 5 times as much energy as those without the condition (Wood et al., 2008). Fatigue is also one of the most common and debilitating symptoms associated with acquired brain injury. The average length of a focus group is between 90mins to 2 hours (Kruegar and Casey, 2009, Kitzinger, 1995); this is likely to be too long for participants with SSPI.

These difficulties should not, however be seen as a reason to exclude those with learning disabilities from participation in research (Mactavish et al., 1998).

Diversity

A commonly used term in focus group research is "purposive sampling". Purposive sampling can be defined as "*a procedure by which researchers select a subject or subjects based on predetermined criteria about the extent to which the selected subjects could contribute to the research study*."(Vaughn et al., 1996 p. 58). While people with SSPI will have some experiences in common which a researcher may want to delve into and discuss, people with SSPI are still a diverse group of participants. It may be that more specific requirements than simply having a SSPI are required for participants in the group, such as in studies investigating the employment experiences of adults who use AAC (McNaughton et al., 2006, McNaughton et al., 2001b, McNaughton et al., 2002b).

This can then make recruitment of participants more difficult and may require looking at a broader geographical region.

Experimental Design and Methodology

Participants

For this pilot study, AAC users who were capable of giving informed consent using a modified consent process (see Table 1) and who received services at a resource centre either as day service users or residents were potential participants. Participants were recruited through the Communication Support Worker at the resource centre.

10.5.1. Focus group composition

Six participants (3 female and 3 male) took part in the focus group. Their age ranged from 25 to 55. Five of the participants attended the resource centre as day service users; one was a resident service user.

Preparation of the Focus Group

The focus group was held in the resource centre in a room used for activity sessions. As all of the participants used motorized wheelchairs a semi-circle was formed with facilitators placed next to the participants who required the most help with their communication. The moderator sat at the top of the semi circle with the video recorder placed to the left of the moderator to ensure all the participants were in view (see Figure 1). One participant had a hearing difficulty in addition to his speech impairment so a facilitator was sat next to him at an angle to allow for better communication of the questions and to assist with signing of questions. The video camera was able to rotate and was operated by an assistant. The session was video recorded to capture the non verbal behaviour, which is often a primary means of communicating for participants with SSPI.

Obtaining Consent

As the adults with SSPI were unlikely to be able to consent through traditional channels (e.g. giving participants an information sheet to read and then asking them to sign a consent form), the protocol described in Balandin, Berg et al., (2006), was followed.

Participants were asked a series of multiple choice questions to which they could answer yes or no such as "Can you withdraw at any point?" to ensure they had understood their rights in the study, see Table 1.

Assistants in the Focus Group

In addition to the participants and the principal moderator, a moderator's assistant to assist in note taking, two facilitators to aid in communication with the participants and a further assistant who operated the video camera were present.

Length of Session

Due of the increased chances of fatigue the session only lasted 1 hour, participants were told they could take a rest or leave at any time. During the session, one participant chose to leave for 20 minutes and returned before the end of the session.

Questions

The number of questions in the session was significantly less than the usual number of questions that would be set in a focus group. A focus group with participants without disabilities could expect to devote 10 minutes to crucial questions (Kruegar and Casey,

2009), giving the potential for 6 questions an hour. In this session only 3 questions were covered.

This was for two reasons, to reduce the cognitive demands of participants and to allow participants longer to prepare their answers.

The moderator took care to ensure that the participants did not interrupt each other, and allowed one participant to complete what they were saying before starting their own reply, this intervention was only required once during the session. The participants were very patient in waiting for each other to either prepare an answer on an AAC device or to have their reply confirmed by a facilitator.

Obtaining Feedback

Frequently in focus groups, a written summary or even the full transcript will be provided to participants after the focus group session (Hemsley et al., 2008c, Hemsley et al., 2007). Given that the participants had varying levels of literacy the feedback was provided orally at the end of the session by the moderator's assistant. This feedback served two functions; firstly to ensure that the note taker had captured the main topics of conversation and secondly to act as an affirmation to the participants that their views had been listened to.

Results of Session

Requirements Gathered

At the start of the session participants were reluctant to offer many suggestions but after some initial encouragement, were keen to provide a variety of requirements regarding how they would interact with the computer system, what support they would need and how they would expect the system to respond to the interaction. Participants were also very good at considering how other users and people at the centre with different needs would use the system. The participants provided 7 significant requirements related to the functions of the system and 4 significant requirements related to the design of the interface.

Participants Response to Session

Participants were initially hesitant in answering questions and initially only gave short answers in reply to a direct question. After the first 18 minutes participants began to directly engage with each other with only minimum input from the moderator to move onto the next question or to bring the conversation back on topic.

The time spent on each question increased during the session from an initial 3 minutes on the first question to a peak of 26 minutes on the third question.

An examination of the transcript shows that 47% (n = 43) of the utterances by participants were unprompted and that all participants responded unprompted at least once. For two participants, the number of unprompted responses outnumbered their prompted responses. The other interesting result from the transcript is that the 3 participants who responded the most were the 3 with the most severe physical and speech impairments.

There were 9 significant pieces of information for the requirements of the system provided during the course of the focus group.

The feedback section of the focus group was the most animated of the hour with participants agreeing both vocally and with nodding of the head that the account was accurate. The summary section of the session finished with one participant saying "*One last thing, thank you for listening*".

Discussion

The results of this pilot study show that a focus group can be run with individuals with SSPI if the necessary planning steps are taken.

For the focus group to be successful, careful thought and consideration should be given to the planning of the session. It may be useful for the moderator to have more prior information on the needs and backgrounds of participants than is usually used in focus groups where the moderator may only know the age, gender and qualifying features of the participant to the topic under discussion. This will enable the research team to make better plans and preparation for the session.

Unlike traditional focus groups where a moderator may act alone, or with the help of one assistant, holding a focus group with adults with severe speech and physical impairments and other disabilities requires a team approach. Ideally all of those on the research team would be experienced in working with people with SSPI. Where this is not possible, at the very minimum the facilitators need to have experience in the field, ideally with those who they are facilitating in the group.

The location for the focus group needs to be visited before the session, ideally several days in advance. This visit allows those organising the focus group to check that the room is suitable and accessible for the participants. The visit can also assist the video operator to find the best position for a video camera to be placed in order to capture all the participants at once, and if this is not possible to make arrangements for a rotational stand for the camera to sit.

As with all focus groups, careful consideration of the questions is needed to help with a successful and productive focus group (Kruegar and Casey, 2009). It is important that the researcher ensures that the questions are both simple enough for the participants to understand and answer and that they do not lead participants unconsciously to any one

answer. It may also be helpful for participants to be given the questions in advance of the focus group session, this can be particularly helpful to those participants who use AAC devices and may allow those with some devices to prepare and store responses and experiences ahead of time.

Another method for increasing the production rates of conversation would be to allow participants to prepare answers simultaneously rather than waiting for one participant to complete their response before continuing the conversation. This would be likely to speed up the production of conversation but would possibly diminish the amount of interaction between participants. In an audience situation this technique could be useful where a speaker is looking for individual responses, but when using a focus group the interactions between participants and how they respond to each other's comments is valuable data. Experts on focus groups suggest that one of the defining features of a focus group which differentiates it from a group interview is the examination of the interaction between participants (Kitzinger, 1994). During the focus group session, the participants with SSPI were able to interact with each other and many of the ideas that they generated emerged as a result of discussion with and suggestions by other participants. Often one participant would share a story relating to an experience and this would prompt another to think of a similar story or offer suggestions on how a piece of technology could have helped them there.

Conclusion

There is now a growing awareness of the value in including participants with disabilities in research around technology which both may have an effect on interventions for people with that disability and also in research involving the general population. Participants with SSPI may have a range of other disabilities that have the potential to impact on their ability to participate in a focus group, however with proper planning and consideration the impact of these disabilities can be reduced.

The use of participants with SSPI in focus groups is still a new approach and as yet no best practice exists; this work builds on existing research involving participants with SSPI

as focus group participants and offers suggestions for those considering conducting focus

groups involving adults with SSPI.

References

- 1. American Speech-Language-Hearing Association, 2002. Augmentative and alternative communication: knowledge and skills for service delivery [knowledge and skills]. Available from: www.asha.org/policy [Accessed 19 July 2008].
- 2. Balandin, S., Berg, N., and Waller, A., 2006. Assessing the loneliness of older people with cerebral palsy. Disability and Rehabilitation, 28 (8), 469–479.
- 3. Cameron, J., 2005. Focussing on the focus groups. In: I. Hay, ed. Qualitative research methods in human geography. 2nd ed. Melbourne: Oxford University Press.
- 4. Catterall, M. and Maclaran, P., 1997. Focus group data and qualitative analysis programs: coding the moving picture as well as the snapshots. Sociological Research Online, 2 (1).
- Clayton, L., 2006. HCI and cognitive disabilities. Interactions A Contradiction in Terms?, 13 (3), 14– 15.
- 6. Communication Matters, 2003. What is AAC? [online] Available from:
- http://www.communicationmatters.org.uk/downloads/focuson/What_is_AAC_A4.pdf [Accessed 7. 19 August 2008].
- Communication Matters, 2008. Let your hands do the Talking [online]. Available from: http://www.communicationmatters.org.uk/downloads/focuson/Let_your_Hands_do_the_Talking_A4.p df [Accessed 19 August 2008].
- 9. Dickinson, A., Arnott, J., and Prior, S., 2007. Methods for human–computer interaction research with older people. Behaviour and Information Technology, 26 (4), 343–352.
- 10. Finnie, N.R. and Bavin, J., 1997. Handling the young child with cerebral palsy at home. Elsevier Health Services.
- 11. Gibbs, A., 1997. Focus groups. Social Research Update, 1997(19).
- 12. Hemsley, B., Balandin, S., and Togher, L., 2007. Older unpaid carers' experiences supporting adults with cerebral palsy and complex communication needs in hospital. Journal of Developmental and Physical Disabilities, 19 (2),115–124.
- Hemsley, B., Balandin, S., and Togher, L., 2008. Professionals' views on the roles and needs of family carers of adults with cerebral palsy and complex communication needs in hospital. Journal of Intellectual & Developmental Disability, 33 (2), 127–136.
- 14. Higginbotham, D.J., et al., 2008. The effect of context priming and task type on augmentative communication performance. Augmentative & Alternative Communication (AAC), 25 (1), 19–31.
- 15. Hustad, K.C., 2006. A closer look at transcription intelligibility for speakers with dysarthria: evaluation of scoring paradigms and linguistic errors made by listeners.
- 16. American Journal of Speech-Language Pathology, 15 (3), 268–277. ISAAC, 2009. ISAAC Barcelona 2010 [online]. Available from: http://www.isaac2010.org/ [Accessed 19 June 2009].
- 17. Jahnsen, R., et al., 2003. Fatigue in adults with cerebral palsy in Norway compared with the general population. Developmental Medicine & Child Neurology, 45 (5), 296–303.
- 18. Johnson, J.M., et al., 2006. Perspectives of speech language pathologists regarding success versus abandonment of AAC. Augmentative & Alternative Communication (AAC), 22 (2), 85–99.

- 19. Judge, S.L., 2001. Computer applications in programs for young children with disabilities: current status and future directions. Journal of Special Education Technology, 16 (1), 29–40.
- 20. Kitzinger, J., 1994. The methodology of focus groups: the importance of interaction between research participants. Sociology of Health & Illness, 16 (1), 103–121.
- 21. Kitzinger, J., 1995. Qualitative research: introducing focus groups. BMJ, 311 (7000), 299-302.
- 22. Krueger, R.A. and Casey, M.A., 2009. Focus groups: a practical guide for applied research. 4th ed. Thousand Oaks, CA: SAGE Publications.
- 23. Kurniawan, S., Mahmud, M., and Nugroho, Y., 2006. A study of the use of mobile phones by older persons. Paper presented at the CHI '06 extended abstracts on human factors in computing systems. 8
- 24. Mactavish, J., Lutfiyya, Z., and Mahon, M.J., 1998. "I can speak for myself": involving individuals with a mental disability as research participants. Journal of Mental Retardation, 38 (3), 216–227.
- McNaughton, D., Light, J., and Arnold, K.B., 2002. "Getting your wheel in the door": successful fulltime employment experiences of individuals with cerebral palsy who use augmentative and alternative communication. Augmentative & Alternative Communication (AAC), 18 (2), 59–76.
- McNaughton, D., Light, J., and Groszyk, L., 2001. "Don't give up": employment experiences of individuals with amyotrophic lateral sclerosis who use augmentative and alternative communication. Augmentative & Alternative Communication (AAC), 17 (3), 179–195.
- McNaughton, D., et al., 2006. "My dream was to pay taxes": the self-employment experiences of individuals who use augmentative and alternative communication. Journal of Vocational Rehabilitation, 25 (3),181–196.
- Morgan, D.L. and Kruegar, R.A., 1993. Successful focus groups (Illustrated ed.). Newbury Park, CA:SAGE.
- 29. Murphy, J., 2006. Perceptions of communication between people with communication disability and general practice staff. Health Expectations, 9 (1), 49–59.
- 30. Murphy, K.P., Molnar, G.E., and Lankasky, K., 2000. Employment and social issues in adults with cerebral palsy. Archives of Physical Medicine and Rehabilitation, 81 (6), 807–811.
- 31. Newell, A.F. and Gregor, P., 2000. "User sensitive inclusive design": in search of a new paradigm. Paper presented at the Proceedings on the 2000 conference on Universal Usability.
- 32. Oliver, M., 2002. Emancipatory research: a vehicle for social transformation or policy development. Paper presented at the 1st Annual Disability Research Seminar.
- 33. Parsons, S., et al., 2006. The use of ICT by adults with learning disabilities in day and residential services. British Journal of Educational Technology, 37 (1), 31–44.
- 34. Rackensperger, T., et al., 2005. "When I first got it, I wanted to throw it off a cliff": the challenges and benefits of learning AAC technologies as described by adults who use AAC. Augmentative & Alternative Communication (AAC), 21 (3), 165–186.
- 35. Rate Enhancement. 2009. Augmentative and alternative communication at the University of Washington [online]. Available from:
- http://depts.washington.edu/augcomm/02_features/04d_rateenhance.htm [Accessed 20 April 2009].
 36. Rennekamp, R.A. and Nall, M.A. Using focus groups in program development and evaluation. Lexington: University of Kentucky.
- Rosenbaum, S., et al., 2002. Focus groups in HCI: wealth of information or waste of resources? Paper presented at the CHI '02 extended abstracts on Human factors in computing systems.
- 38. Ryoung, K., et al., 2009. Focus group interview for designing a growing robot. Paper presented at the Proceedings of the 4th ACM/IEEE international conference on Human robot interaction.
- Schlosser, R., 1999. An introduction to consumer perspectives in AAC research: approaches, techniques and their theoretical underpinnings. In: F. Loncke, H. Clibbens, H. Arvidson and L. Lloyd, eds. Augmentative and alternative communication: new directions in research and practice. London: Whurr Publishers, 257–261.
- 40. Sharp, H., Rogers, Y., and Preece, J., 2007. Interaction design: beyond human-computer interaction. Chichester: John Wiley & Sons Ltd.
- 41. Vaughn, S., Schumm, J.S., and Sinagub, J.M., 1996. Focus group interviews in education and psychology. Thousand Oaks, CA: SAGE.
- 42. Waller, A., et al., 2005a. Training AAC users in user-centred design. Paper presented at the Accessible Design in the Digital World.
- 43. Waller, A., et al., 2005b. Facilitating user feedback in the design of a novel joke generation system for people with severe communication impairment. Paper presented at the HCII.
- 44. Willson, P., et al., 2005. Health visitors and the parent/child relationship: a focus group study. Journal of Epidemiology & Community Health, 59 (Suppl 1), A6–A7.

Appendix D. Forum Production and Script

Accepted for presentation at the International Society for Augmentative and Alternative Communication Conference, Barcelona, July 2010

Including Adults with Severe Speech and Physical Impairments as Actors in Forum

Theatre

Prior, Suzanne; Waller, Annalu; Kroll, Thilo; Morgan, Maggie; Cummins, Kathleen INTRODUCTION

Forum theatre was developed by Boal in Brazil and is described in his book "Theatre of the Oppressed" (Boal, 2000). Boal originally intended forum theatre to be used by oppressed and marginalised groups to allow them to provide their views and/or influence political change; since its beginning in the 1970s forum theatre has been adapted for different purposes (Newell et al., 2006c).

Newell et al (Newell et al., 2006b) suggested that forum theatre could be used in facilitating discussion between users of AAC devices and clinicians or designers. Given the potential for forum theatre to be used as a means of promoting interests and needs of marginalised groups, it is perhaps surprising that to date no published papers report it being used for either scenarios dealing with situations faced by adults with Severe Speech and Physical Impairments or with actors who have Severe Speech and Physical Impairments (SSPI).

As part of a wider project (Prior et al., 2009, Blackstone, 2009) looking into issues surrounding adults with SSPI and their experiences in hospital, forum theatre was used to elicit requirements for an electronic multimedia profile. This paper describes the experiences of those involved in the process including two amateur actors with SSPI.

Ethical approval for this study was obtained from the organisations involved in recruitment and from National Health Service (NHS) Tayside Ethics Committee.

PROCESS

Script writing: Two adults with SSPI from a day and residential centre were recruited to act the role of patients in two specially commissioned scenarios. In keeping with previous work which used forum theatre for requirements gathering, a professional script writer was employed to develop a story, write the script and direct the actors in rehearsals (Carmichael et al., 2005, Newell et al., 2006c). Unlike previous work where scripts have been based upon the researchers' ideas of situations where a potential piece of technology could be used, the scripts in this study were based on stories told in a focus group with adults with SSPI (Prior et al., In Prep). Participants in the focus group discussed problems they had personally faced in hospital as a result of communication breaking down between themselves and medical professionals. These stories, together with others reported in the literature (Hemsley et al., 2008a), formed the kernels of two scripts. Several iterations of the script were produced and were modified in light of feedback from experienced medical staff and researchers.

Rehearsal: A single day's rehearsal would normally suffice for the preparation of the performance of two short scripts. However, three days were spent in rehearsal as amateur actors were playing the parts of patients. The two amateur actors took turns to rehearse with the professional actors in order to allow them time to rest between sessions. To reduce fatigue on the parts of the amateur actors, rehearsals took place at the residential centre. A support

worker was also present during rehearsals to assist with communication and to provide personal care.

As with other plays with nonspeaking roles, e.g., Operation Elvis (Taylor, 1983), the script writer had used prose to provide directions for the actors. Dialogue was stored in voice output communication aids where appropriate. It soon became apparent that the actors with SSPI required spoken dialogue even if their speech was unintelligible. The scripts were rewritten for day two to include dialogue for all characters.

In forum theatre questions may be asked of the actors 'in character'. An afternoon of rehearsals was spent preparing the amateur actors for this. Background stories for each character were discussed until the actors were confident they would be able to improvise for the majority of questions asked of them.

Performance: The audience for the performance was consisted of junior doctors who were attending the session as part of a disability awareness lecture. A brief introduction to the performance was given prior to the first scenario, but no specific information on content was given. After the first scenario a trained facilitator invited comments from the audience on how they felt upon seeing the scenario and what their impressions were on how each character behaved. The audience had the opportunity to ask questions of the characters on why they acted the way they had and how they felt in the situation.

REFLECTIONS ON THE EXPERIENCE

Amateur Actors' Experience: Both amateur actors with SSPI grew in confidence during the rehearsals. They were able to voice their opinions and engage in lively discussions with the researchers and the two professional actors. Both enjoyed the experience and one is now exploring further theatre work as part of training for new support workers. The amateur actors noted the significance of what they were doing. One reflected that:

"I think it will be good for the doctors to see someone like me using a talker and being in a play"

Professional Actors' Experiences: The two professional actors were initially nervous at working with the adults with SSPI. They quickly became confident in working with the two amateur actors and other adults with SSPI in the residential centre. One actor described the experience:

"I had no idea what to expect at all. On arrival to day one of rehearsals, I was really anxious ... and thought the process would be a nightmare. It was a completely different story as we had a fantastic rehearsal atmosphere and really got on with the task of producing the two scenes."

Medical Trainer: The disability awareness session for the doctors was organised by a learning disability nurse. The nurse had not met any of the actors or researchers prior to the session but had been informed of the focus for it. The nurse reflected on the experience of watching the session:

"The performances were wonderful; they really got the message across about communicating with people. I felt that it could be used as a session about how to communicate with people, as you could have had a great discussion going"

The nurse did note that the doctors may have benefited from more information on strategies for communicating with adults with SSPI and this has been noted for future work.

DISCUSSION

The process of including adults with SSPI in acting for a forum theatre session is time consuming and requires careful planning by those wishing to undertake it. The time taken for rehearsals in this instance was three times that of usually needed when working with professional actors without disabilities. The resources required to facilitate full participation were also increased as support staff was required.

The involvement of adults with SSPI in scenarios featuring characters with SSPI has many advantages. First, it is unlikely that a professional actor could realistically portray a character with SSPI. Second, the actor with SSPI is able to provide more believable feedback to the audience when questions are posed to them in character. Finally it is in keeping with the "nothing about us without us" manifesto (Oliver, 2002) to involve adults with disabilities in research that has the potential to have an impact on their lives.

CONCLUSION

Despite the difficulties and extra work involved in the preparation of the forum theatre sessions with adults with SSPI, the researchers involved feel the process would not have been as beneficial or as insightful without the help of the two amateur actors with SSPI. We recommend the use of forum theatre for facilitating discussion between service users with SSPI and professionals who may work with them.

ACKNOWLEDGEMENTS

This work is part of a doctoral study funded by Capability Scotland. We thank the actors, support staff and colleagues who participated in the project.

<u>Scripts</u>

SCENARIO 1 - " A shame, eh?"

Scene – 'hospital ward'. A bed, made up, with female patient, Maria, in it. She could be clutching an oxygen mask (?) Her eyes are initially closed but once dialogue starts, her eyes open.

Enter consultant(female) and FY1 (male).

C – (*picking up notes*)

Right Dr Jones, would you like to present your patient.

- FY1- Yes, this is Maria Russell. Brought in yesterday. She has double pneumonia. Cerebral palsied, non-speaking, can't communicate.
- C- Sad case. What is your suggested plan?
- FY1 I was about to prescribe ceftriaxone?
- C That can have some unpleasant side effects and in a case like this....have you considered palliative care?

	(Nods to side of stage)
C -	(To Maria) We'll come back in a bit

C and FY1 move to other side of stage

- FY1 Palliative care? There's no actual morbidity? (*check terms with medics*)
- C No, but in this type of case.....you know....what's the point? It just prolongs suffering. A blessed release.

By this time MR is beginning to react with noise and movement.

- FY1 She's (*checks notes*) only thirty-five, that's quite young?
- C Yes...but I mean....what sort of life?.....
- C responds to MR's now frantic shouts and movements
- C There, there. Don't you worry. We're looking after you.

MR gets even more frantic

- FY1 Can she hear us?
- C Probably not. Even if she can, she would not understand.

FY1 looks a little doubtful and unsure.

FY1 - Ah! That's my pager!

Takes pager/mobile out of pocket and looks at it)

Would it be alright with you if I go, I've been waiting on these test results. May I catch you up later?

C - Fine.

Exit FY1 in a hurry. MR is now very distressed, breathless and possible coughing.

C -	There, there! Don't you upset yourself. You're quite safe.
	I think we'd better give you something to calm you down.
	Nurse!

PAUSE

Thinks bubble for Consultant, Maria and then FY1. (As soon as thinks bubbles over, FY1 exits to change) Go from thinks bubbles to general discussion about what assumptions are being made. Remind audience that AAC device was taken away when she arrived and she was feeling too ill to make too much of a fuss.

Scene 2 – Consultant is standing by the bed, trying to calm MR. He calls for the nurse again.

Enter Peter Russell, Maria's husband.

PR	Hi Maria! How are you?
C	And you are?
PR	I'm Maria's husband.
C	Oh.
PR	(to Maria) Where's your talking device?

Maria manages to point off

PR They've not taken it off you?!

Maria nods

- PR Oh, for God's sake!
- C Maria is very distressed. I think you need to calm down.

By now, Maria has stretched out an arm, Peter has taken her hand and sat down on the bed. Maria is much calmer!

PR (*looks at MR*) What's going on? Don't worry, sweetheart. I took Jessica to your mother's, she's quite happy. Wants to come and see you but I said the doctors had to give you your medicine first. I've rung work – so they know you won't be in for a while!

	You are looking a bit rough.
	(To consultant) You've started treatment.
С	Ernot quite yet.
PR	But the GP said she's got pneumonia. That's an antibiotic job, isn't it? You do know she's allergic to penicillin don't you?
С	Er – well – yes. But we do need to.make an individual assessment for each patient. I've just now completed my assessment, and I'm going to sort out the medication right now

PAUSE

SCENARIO 2 – Who's the Expert?

Scene – A hospital waiting room. Tony Lynley, a 45-=year-old man with CP, is seated on one of the chairs. He has on a hospital gown, covered by a dressing gown. (Perhps we can mange just with the dressing gown?) He has an AAC device with him. Possibly a nurse is sorting forms etc at a nearby table. In which case –

NURSEMr Lynley?Tony nodsNURSEWould you like me to look after that for you? While you have your scan?TONYWhen I go in.NURSEOh! That's what it's for.
OK. Give it to me then.Nurse goes back to her table.
Enter Doctor Jessop. He has a good look at Tony, goes to the desk.

DOCTOR The patient's notes?

The nurse passes him the notes, which he scans very briefly.

DOCTOR (*To Nurse*) This is ridiculous! How on earth can I put him through a scan?

(To Tony) Can you stay still?

Tony shakes his head.

DOCTOR (*To Nurse*) Then this is a waste of my time! Who on earth referred him? His GP?

Doctor checks notes again

DOCTOR The orthopaedic department? What do they think they're playing at?

(*To Tony, very loudly, as if T is deaf/stupid*) I'm sorry, you'll have to go home! We can't do the scan if you can't stay still! Do you understand me?

(To nurse) Can you sort him out, get his transport back, whatever?

Doctor puts notes back on table and begins to exit

Meantime Tony has been fiddling with his AAC, to produce a message he has produced earlier.

TONY (*via AAC*) Last time I did this, I had an epidural. That relaxed me enough time to give me the scan.

Doctor does a 'double-take'.

DOCTOR What?!

TONY plays message again. Tony is looking pretty cross.

DOCTOR (visible pause) Right! That's what we'll do!

Goes back to table, picks up notes and prepares to write in them.

DOCTOR It would have been helpful if ---- ah------ I see.

Right! We'll fix another appointment and arrange for an epidural.

Tony looks fed up at the delay.

DOCTOR We'll fit you in as soon as possible. Nurse, make sure he's put on the cancellation list, will you?

PAUSE

Appendix E. Participatory Design of a Talking Photo Album

Paper presented at the 2010 RaATE Conference, Coventry UK

<u>The Participatory Design of a Talking Photo Album for Adults with Severe Speech and</u> <u>Physical Impairments</u>

S. Prior, H.Betzler, A.Waller

Communication is an essential part of life and an important part of communication is the ability to tell stories. These stories can include every day events up to momentous life occasions. Through stories people build relationships, learn new information and can argue their point of view. However, people with Severe Speech and Physical Impairments (SSPI) cannot access their stories easily. At present voice output communication aids are not suited to sharing stories; it is difficult to create a story, store it, retrieve it and narrate it (Waller 2006). People with SSPI often rely upon a carer to enter stories into the device, but this is time consuming and results in static, monologue output as the story text tends to be stored under a single key. Another way of sharing stories is through photographs, with some use being made of "speaking photograph albums"; allowing voice recordings to be linked to individual photographs. However, non-speaking individuals still have to rely on speaking helpers and the playback still mimics a monologue.

The problems surrounding the "talking photograph album" were addressed within a participatory design project which is investigating ways to include people with SSPI in the design of technology. The project followed participatory design guidelines being developed at the School of Computing at the University of Dundee (Prior 2010).

Six adults with SSPI and a support worker were enabled to participate in all stages of the design using innovative design methods; these included magnetic white boards, etran boards and PowerPoint designs.. Participants were initially invited to choose the focus of the project; they expressed a desire to share their stories and their photographs but indicated that they currently require assistance in doing this. The participants then helped design the software from low level paper based designs through to a fully working piece of software.

Appendix F.DissertationonDevelopmentofTalkingPhotoAlbum



School of Computing

Participation of People with Complex Communication

Needs in Designing Assistive Technology Software

for Information Sharing

Author: Heike Betzler

Degree: MSc in Applied Computing, University of Dundee, UK

Supervisor: Dr Annalu Waller

Date: 28st March 2011

Executive Summary

Communication is vital - not only to express wants and needs, but also to build relationships and to share experiences. People with Complex Communication Needs (CCN) have difficulty in sharing life-experiences.

236

The aim of this master project was to develop a software application for and with people with CCN that would allow them to share information.

Eight adult participants with CCN chose photos to share experiences. Using participatory design methods, they provided requirements and helped to design The Talking Photo Album. Prototypes were evaluated at each stage of the development process. The final version was developed in C#, runs on a portable touch-screen computer and includes a database to manage the photos and their descriptions.

The resulting software will be provided to the participants and used by a learning disability centre with adults with CCN.

Declaration:

"I declare that the special study described in this dissertation has been carried out and the dissertation composed by me, and that the dissertation has not been accepted in fulfilment of the requirements of any other degree or professional qualification."

237

.....

Heike Betzler

Certification:

"I certify that Heike Betzler has satisfied the conditions of the Ordinance and

Regulations and is qualified to submit this dissertation in application for the degree

of Master of Science."

.....

Dr Annalu Waller (Supervisor)

Acknowledgements

First of all, I would like to thank my supervisor Dr Annalu Waller for her affirmatory and inspirational guidance throughout the project, which created a very positive and enjoyable work environment. Furthermore, I would like to thank her and her team for making me feel part of the gang. Particular thanks to Suzanne Prior who always lent me an ear.

239

I would like to thank all the participants for their excellent contributions to my study, I enjoyed working with you. Particular thanks to Bernadette Brophy-Arnott who put me in contact with the participants and who helped to arrange the user sessions.

Table of Contents

1	Intr	oduction	1
2	Bad	ckground	3
	2.1	Complex Communication Needs (CCN)	3
	2.2	Importance of Communication	4
	2.3	Augmentative and Alternative Communication (AAC)	4
	2.4	Abandonment of AT	6
	2.5	Including People with Communication Impairments in the Design of	
	Softwa	are Products	7
	2.6	Project Aims	8
3	Hur	man-Centred Design (HCD) Methodology and Models	9
	3.1	Discussion of Different User-Centred Lifecycle Models 1	0
	3.2	Techniques Needed to Engage in User-Centred Design 1	5
	3.3	Guidelines 1	8
4	Red	quirement Gathering 2	1
	4.1	Getting the Project Started 2	21
	4.2	Idea Finding Process and Getting to Know the User Group	23

240

	4.3	Market Research 2	25
	4.3	.1 Commercially Available Talking Photo Albums	25
	4.3	.2 User Testing and Evaluation of a Commercially Available Talking	
	Pho	oto Album 2	6
	4.4	Prototyping and Evaluation 2	29
	4.4	.1 Co-design Sessions	29
	4.4	.2 Usability Button Game	33
	4.4	.3 Keyguard	35
	4.4	.4 PowerPoint (PPT) Prototypes	36
	4.5	Requirement Specifications	39
5	Sof	ftware Design and Implementation 40	6
	5.1	Use Cases	46
	5.2	Overview of Program Architecture5	6
	5.3	Entity-Relationship (ER) Diagram 5	57
	5.4	Platform	59
	5.5	Programming Language	59
	5.6	Database6	30
	5.7	Code Overview6	51

6	6 Testing	
	6.1	Testing Techniques 64
	6.2	Summative Evaluation
7	Cor	nclusions
	7.1	Summary 69
	7.2	Critical Appraisal of the Project 70
	7.3	Recommendation for Future Work 72
8	Ref	erences
9	List	of Appendices

Figure 1: Boehm"s Spiral life cycle model (adapted from Boehm, 1988) 11
Figure 2: The Star life cycle concept (adapted from Hartson and Hix, 1989) 12
Figure 3: LUCID stage flow (adapted from Kreitzberg, 2008) 13
Figure 4: The Wheel process model (adapted from Helms et al., 2006) 14
Figure 5: Commercially available photo album
Figure 6: Users with CCN during individual co-design session
Figure 7: Results from co-design session of photo album screen with users with CCN
Figure 8: Usability Button Game (with 3 button sizes)
Figure 9: Keyguard prototype #1 (1 + 2) and #2 (3) for the Toughbook
Figure 10: PPT-prototype #1
Figure 11: PPT-prototype #2 – photo album screen
Figure 12: Use case diagram for support worker 47
Figure 13: Interface sketch of option screen 48
Figure 14: Interface sketch of upload screen 50
Figure 15: Interface sketch of setup screen 51

243

Figure 16: Use case diagram for illiterate user with CCN
Figure 17: Interface sketch of photo album screen
Figure 18: Use case diagram for user with CCN and literacy skills
Figure 19: Screen structure of The Talking Photo Album application
Figure 20: ER-Diagram of The Talking Photo Album database
Figure 21: Code excerpt from the photo album screen
Figure 22: Method to retrieve data from the database
Figure 23: The final prototype of The Talking Photo Album on the Toughbook 66
Figure 24: Final C# version of The Talking Photo Album 70
Figure 25: Presentation of the master project with participants and support workers
listening

List of Tables

Table 1: Types and sub-divisions of AAC (adapted from lacono, 2004) 5
Table 2: Prior"s guidelines – Working with adults with complex communication
needs in human centred design (adapted from Prior, 2010b) 19
Table 3: Participant profiles. 22
Table 4: Different types of conversation (adapted from Waller, 1992). 24
Table 5: Usability problems identified in heuristic evaluation. 27
Table 6: Functional requirements for The Talking Photo Album
Table 7: Non-functional requirements for The Talking Photo Album 42
Table 8: Usability requirements for The Talking Photo Album
Table 9: Use case 1 for support worker. 47
Table 10: Use case 2 for support worker. 48
Table 11: Use case 3 for support worker. 50
Table 12: Use case for illiterate user with CCN

245
1 Introduction

Assistive technology (AT) devices are anything that helps individuals with disabilities to maintain or increase their functional capabilities regardless of whether the device is commercially bought or self-made (U. S. House of Representatives, 2010). The World Health Organisation (2004) defines AT as "an umbrella term for any device or system that allows individuals to perform tasks they would otherwise be unable to do or increases the ease and safety with which tasks can be performed." In the UK, the term is understood slightly different. In contrast to the former definitions, the focus is on the independence that the individual gains though the use of AT (King's Fund, 2001).

246

One category of AT devices is communication aids. These help people with speech disabilities to communicate. At first sight, speech output devices seem to be liberating for people with Complex Communication Needs (CCN) as they allow the users to be heard and understood. However, not everyone with CCN is pleased with their communication device, whether it is because the device visibly singles them out from others, whether the learning effort required is too high or the devices available are just not very user-friendly. All might be explanations for abandonment rates of augmentative and alternative communication (AAC) devices reported as high as one third (Scherer, 2002). It has been suggested that this might result from a lack of integrating the end users with CCN in the development process of the product (Prior, 2010a).

Taking this hypothesis as a starting point, the objective of this master project was to involve people with CCN in the development of an AT device from the very

1

beginning and to an extent that it was the users who decided what the software

product would be.

2 Background

This chapter describes the setting in which the software development project has been undertaken. It provides the reader with an introduction to CCN, highlights the importance of communication and explains the term AAC. It further discusses the abandonment rate of AAC devices, presents participatory design as a possible development approach and outlines the project aims of this thesis.

2.1 Complex Communication Needs (CCN)

CCN describe the more severe end of the communication impairment severity spectrum. Difficulties that people with CCN face are speech which is difficult to understand, problems in understanding what is said to them, and/or problems in expressing themselves due to limited vocabulary and skills to create sentences (lacono, 2004).

A person may have complex communication needs for multiple reasons which can be classified as developmental or acquired disabilities. In the first case, the person is born with the disability or it develops within the few years of their life; consequently it may affect their social or cognitive development (Department of Communities Disability 2009, Examples of Services, lacono. 2004). developmental disabilities are Down syndrome, cerebral palsy and autism. Acquired disabilities resulting from an illness or injury that are associated with communication impairments are, for example, traumatic brain injuries, spinal cord injuries and strokes, but also degenerative diseases such as multiple sclerosis or motor neurone diseases (Department of Communities - Disability Services, 2009, lacono, 2004). In any case, people with CCN generally have a combination of motor, language, cognitive and sensory impairments which is why they typically

rely on multiple modes to meet their communication needs (Blackstone and Berg, 2003)

249

2.2 Importance of Communication

"Information sharing" is the intended purpose of the device to be developed within the scope of this Master project. The term is included in the meaning of the word "communication" as can be seen when considering its Latin origin: "communicare". "Communicare" has two meanings: "to inform/to communicate" and "to share/to do something together" (Martin, 2004). This highlights that beyond the pure exchange of information, communication also includes sharing of experiences and emotions. Thus, communication can be considered fundamental to all aspects of life as well as for quality of life. Successful communication makes it possible for people to express their wants and needs, but also their thoughts and personality. It allows them to build relationships, make decisions and take part in social activities (Department of Communities - Disability Services, 2009). Consequently, communication can be considered as essence of life (Waller, 2009) and has been declared a human right by The United Nations" "Universal Declaration of Human Rights" (1997).

2.3 Augmentative and Alternative Communication (AAC)

AAC is a means for non-speaking people and people with speech impairments to communicate. As the name suggests, it includes methods that enhance (augment) communication, such as facial expressions or gestures, and those that replace conventional forms of communication (alternatives), e.g. using a symbol board to

communicate. An example for the importance of AAC can be seen from the

following quote by a man with severe cerebral palsy:

"If you want to know what it is like to be unable to speak, there is a way. Go to a party and don't talk. Play mute. Use your hands if you wish but don't use paper and pencil. Paper and pencil are not always handy for a mute person. Here is what you will find: people talking; talking behind, beside, around, over, under and through, and even for you. But never with you. You are ignored until finally you feel like a piece of furniture." - Rick Creech (Musselwhite and St. Louis, 1988)

There are different types of AAC which can be sub-divided in unaided and aided

systems.

Types of AAC							
Unaided							
	Examples:						
• Informal	Facial expressions Body language Gestures Vocalisations Idiosyncratic movements						
Formal	Sign language						
Aided							
	Examples:						
 Low-tech 	Communication boards/books with words, symbols or pictures						
 High-tech 	Single message switches (e.g. Big Mack) Text-to-speech communication aides (e.g. LightWriter) Voice output devices with static or dynamic displays						

Table 1: Types and sub-divisions of AAC (adapted from lacono, 2004).

As can be seen from Table 1, unaided AAC does not involve any equipment.

Informal methods can be in particular useful for AAC users to express their

feelings. However, in order for them to be interpreted correctly the communication

partner must be familiar with the behaviour patterns of the AAC user which might be affected by physical disabilities. In contrast, aided AAC uses devices to convey messages. Depending on whether the devices are electronic or not, these systems are referred to as low-tech or high-tech AAC devices. The latter range from simple single message switches to highly sophisticated computer-based voice output devices with dynamic and multi-level displays.

2.4 Abandonment of AT

Abandonment of AT describes the discontinued use of a device by the user although professionals consider the system appropriate. On average, the percentage of devices that are left behind is one third (Scherer, 2002) which corroborates findings of an earlier study by Phillips and Zhao (1993). However, it should be noted that the exact percentage depends on the degree to which the device is required, i.e. the more a person depends on a type of device, the lower its abandonment rate (Phillips and Zhao, 1993). For AAC systems, Johnson et al. (2006) found an abandonment rate of 28.68%. Their study identified a number of factors leading to abandonment such as little support from family, therapists and assistive technology specialists; lack of motivation, i.e. the user preferring other, simpler means of communication; stigmatization; lack of ongoing adjustments and resulting poor fit of the system as well as poor usability and limited functionality. Moreover, missing user involvement in the design process of the product has been cited as a major contributor to the high rate of abandonment by Riemer-Reiss and Wacker (2000) as well as Scherer (2002).

6

2.5 Including People with Communication Impairments in the Design of Software Products

There is an increasing awareness of the need to include users at all levels of the software development process in general (Sanders, 2002) and also specifically in the development and design of AT (Williams et al., 2008). However, there is debate to what extend users with CCN can be involved in the development process. Whilst Seelman et al. (1998, p. 24) advocate involvement of individuals with disabilities "in setting the research agenda, developing research questions, participating in the research as researchers, advisors, and consultants, testing research ideas, and most importantly, evaluating the results of the research", Newell and Gregor (2000) stress that this might be possible in sociological research, but not in User-Centred Design (UCD). For them, users can be involved, but the activities mentioned above require a trained researcher. Using an UCD approach with people who use AAC is very difficult and is therefore often neglected. However, Waller et al. (2005) overcame this problem by training people with CCN, families and AAC professionals in UCD workshops. In their understanding, including all affected parties in the design process from the beginning is vital for the development of usable products. Successful examples for this approach are the development of the Portland Communication Aid prototype (Allen, 2005) and the CHAMPION project (Prior, 2010a). Designer-facilitated Participatory Design (PD) methods, such as physical models rather than sketches for requirement gathering, were used to integrate AAC users into the development process from the very beginning.

7

2.6 Project Aims

The aim of this project was to develop a software system for and with adults with severe disabilities that would allow them to share information while using Prior"s guidelines: "Working with adults with complex communication needs in human centred design" (2010b). It involved working with end users in all stages of the software development process and included identifying the project remit.

3 Human-Centred Design (HCD) Methodology and Models

HCD is one of the fundamental concepts of Human Computer Interaction (HCI). As described in the standard ISO 9241-210:2010 (International Organisation for Standardization), it is an approach to interactive systems development with the goal of making systems usable and useful, i.e. increasing their effectiveness, efficiency and user satisfaction. To achieve this, HCD focuses on the user, in particular on user needs and requirements, but also human factors/ergonomics and usability knowledge. As a result, the user is involved in every stage of the development process, evaluates alternative designs and has consequently direct influence on the development of the software.

According to Sanders and Stappers (2008), we are moving from the design of products and technology to designing for people"s purposes which centre around people"s needs or societal needs such as unmet needs of people with medical conditions.

The ISO standard 9241-210:2010 lists a number of reasons for the use of HCD. In the context of this project, the most relevant reasons are the increased accessibility for disabled people, the ease of use of the product with resulting reduction of training and support costs and the improved user experience. On the operational side, the HCD approach also aids the definition of functional requirements, reduces the risk of not meeting the stakeholders" requirements and the product being rejected by users.

Moreover, the standard emphasises that the human-centred approach is to be understood as complementing existing system development approaches. The process model of the Wheel (Helms et al., 2006) is an example for this and will be elaborated in the next section of this chapter.

3.1 Discussion of Different User-Centred Lifecycle Models

The lifecycle **Model of the Wheel** developed by Helms et al. in 2006 represents a combination of four earlier models which are in chronological order Royce"s Waterfall model (1970), Boehm"s Spiral Model (1988), Harton and Hix"s Star life cycle (1989) and Cognetics" Logical User-Centre Interaction Design (LUCID) framework which was originally developed in 1998. Before presenting the Model of the Wheel itself, some brief explanations of the four models which were of influence are given.

In contrast to the classical, strictly linear **Waterfall model** (Royce, 1970) where each stage of the development process must be completed in its entirety before the next stage can begin, Boehm"s **Spiral life cycle** model (Boehm, 1988) takes an iterative, incremental and risk driven approach.



Figure 1: Boehm's Spiral life cycle model (adapted from Boehm, 1988).

As can be seen from Figure 1, each cycle begins with the identification of the objectives of the current cycle while considering alternatives and imposed constraints such as costs or schedule. At the next stage (quadrant), risks are analysed and evaluated with the help of prototypes. The biggest risk is then dealt with first and can therefore be considered the determining factor of the development in each cycle. In a next step, results are reviewed and the next iteration is planned. Advantages of this approach are that design flaws are discovered early as software is produced early in the life cycle (Boehm, 1988). Furthermore, the high amount of risk analysis is especially beneficial for mission-

critical projects. However, it should be noted that risk analysis requires high levels of expertise which may not always be given.





The **Star life cycle** concept is a usability engineering process that was developed by Hartson and Hix in 1989. As can be seen from Figure 2, usability evaluation is the central point of the model. It interconnects the different and distinct, but otherwise unconnected, development activities at the points of the star. As these are in no particular order, there is no predefined sequence of activities which the developer has to follow in the development. After an activity is completed, it is evaluated before the next activity is started. The goal of this approach is a continual evaluation and iteration during the development process with smaller loops of iteration than the spiral model. A disadvantage of this concept might be the missing notion of progress towards the final product. The **LUCID framework** "s goal is to offer product users the "five E"s": effectiveness, efficiency, engagement, error tolerance and ease of learning (Kreitzberg, 2008). The framework consists of a series of six stages which are: envision, discovery, design foundation, design detail and build release. Each stage itself entails the elements depicted in Figure 3, where the component "Activities" is replaced with the core activity of the current stage.



Figure 3: LUCID stage flow (adapted from Kreitzberg, 2008).

From this latter framework, Helms et al. (2006) extracted the four basic activity types for interaction design, i.e. analysis, design, implementation and evaluation. They constitute the four quadrants of the wheel"s mini-spirals which can be seen in Figure 4 and which themselves have been adopted from Boehm"s Spiral model. The global usability evaluation hub is borrowed from Hartson and Hix"s Star lifecycle and the direction of evolution is inherited from the classical Waterfall model.



Figure 4: The Wheel process model (adapted from Helms et al., 2006).

The process **Model of the Wheel** can be customised to the needs of a specific project as well as it can be adjusted halfway-through if required due to changes in budget, schedule or resources. In order to produce an instance of the wheel the appropriate cycles are selected and techniques for the activities in each cycle are specified. The evaluation hub can be considered a higher-level controlling activity determining the next most appropriate cycle (Helms et al., 2006).

The model adheres to the four key principals of usability engineering (Hewett and Meadow, 1986) which are:

- Early and continual focus on users (specific techniques are explained in section 3.2)
- Integrated design (starting the project at various places at once while

recognising that usability is important)

- Early and continual user testing (using mock-ups, simulations, early prototypes or demonstrations)
- Iterative design (identification of requirement changes, ability and willingness to make changes)

To conclude, the process Model of the Wheel seems worthwhile pursuing as it manages to combine major software development processes with usability engineering and interaction design activities while putting the user at the centre of all activities.

3.2 Techniques Needed to Engage in User-Centred Design

When working with users, different techniques are available at different stages of the design process to elicit user needs and attitudes, to generate ideas, but also to receive feedback from users through testing and evaluating prototypes. A number of key methods are described in this section to give insight in how users can be involved in software design and development.

Focus groups are group discussions which are often used in the analysis phase of a development project. They aim at revealing user attitudes and expectations towards a certain product or concept (Wilhelm, 2010). The participants can be either users or experts. Usually two-hour meetings are arranged with at least six participants to discuss new concepts and to identify product issues. The sessions are led by a moderator who keeps the group"s focus and ensures that all participants get a chance to express their opinion. The strengths of focus groups are their informal character, the fact that the participants are in a peer-to-peer

situation and the spontaneously generated ideas. On the other hand, several focus groups are required to get a representative result and participants may have difficulties understanding and formulating their needs (Nielsen, 1997, Wilhelm, 2010).

Another qualitative research method is the **semi-structured interview**. In preparation for the interview the aim of the interview should be determined and a representative sample selected. A characteristic of the semi-structured interview is that the sequence of questions is determined by an interview-guide whereas the questions are open-ended to make the interview flexible (Hove and Anda, 2005). Following the flow of conversation, new extending or echoing questions can be added as sort of probes. The benefits of this method are its flexibility, interactivity and that high qualitative data can be generated. On the downside, it is expensive to do and time consuming. Furthermore, there is the risk of interviewer effects and response bias.

In early user-centred design approaches users were passive subjects. The **co-design** approach in contrast, considers end-users the experts because of their experience. In this emerging discipline formerly referred to as participatory design, designers and people not trained in design collectively work together. As Sanders and Stappers (2008) point out, the expert role of the user is important for learning about the user as well as the context of use, for idea generation and the development of concepts. The task of the researcher/designer is to support the experts by providing them with appropriate tools that allow them to express their ideas, for example, through paper prototypes. On a critical note, this approach assumes that all people are creative which is obviously a broad generalisation.

A method that can be used as soon as a first prototype is developed is user observation. This is a very unobtrusive way to learn about the interaction between the end-user and the product. If the user has given consent, the interaction can also be video recorded for later analysis. Attention should be paid to extraneous factors such as environment, interruptions and distractions. However, a clear limitation is that the observer can see what is happening but without knowing why. In addition, intentions, feelings and satisfaction cannot be measured (Holzinger, 2005).

A **cognitive walkthrough** is a task-specific usability inspection method which is usually performed by a group of designers and developers. The activity aims at identifying usability issues that the targeted end-user group may encounter by going through each step of specific tasks. The focus is on exploratory learning as well as on how easily future users can complete each task. The method can be applied early in the design/development process (before coding commences) and is, therefore, an inexpensive way to gain usability insight. However, it should be noted that it is based on assumptions of the designers/developers and that there is a risk of paying too much attention to details at the beginning of a project (Holzinger, 2005).

Heuristic evaluation is the most common informal usability inspection method and can take a holistic view of a running system. Nonetheless, it can also be applied earlier in the development process, e.g. in the specification stage. Following a recognised list of usability principles (e.g. Nilsen^s Usability Heuristics (Nielsen, 2010c)), usability or domain experts steps through the interface of the application and judge each interactive element. The evaluations are carried out individually by each evaluator to ensure independence and eliminate potential

bias. Nielsen and Molich (1990) recommend that heuristic evaluations should never rely on a single evaluator as an aggregation of independently captured opinions from a number of different people will most likely always yield more usability issues. Nielson and Molich showed that the ideal number of evaluators in terms of the cost-benefit ratio is 3 – 5 people. The strengths of heuristic evaluations are that they are quick to perform, relatively inexpensive, intuitive to do and that they can be conducted early in the development process. However, specialised products require expert evaluators who may be difficult to find. Another disadvantage is that problems may be identified, but usually no suggestions for possible solutions are elicited.

However, using the techniques summarised above with severely disabled people is problematic. Traditionally, people who use AAC have only been used in final evaluations or at most as informants (Olsson, 2004). Therapists have instead been used in the design of systems. As mentioned earlier, in order to involve adults with severe disabilities in the project, guidelines developed by Prior (2010b) were followed.

3.3 Guidelines

Prior"s rationale for the guidelines "Working with adults with complex communication needs in human centred design" (2010b) was the increasing number of computer users with cognitive and developmental disabilities and the high abandonment rate of AT. Prior infers that it may be possible to reduce the abandonment rate through the inclusion of end users in the design process. The guidelines (Prior, 2010b) are based on the standard ISO-13407 (Human-centred design processes for interactive systems) and provide guidance on how design

18

processes can be conducted when involving users with CCN. The guidelines,

summarised in Table 2, are intended for developers or researchers who will be

working with users with CCN.

Table 2: Prior's guidelines – Working with adults with complex communication needs in human centred design (adapted from Prior, 2010b).

Summary of Prior's Guidelines							
Preparation phase							
 Before starting any work with people with CCN, carefully examine the ethical situation, special procedures are most likely required when working with disabled people. As recruiting a representative sample of disabilities and needs is rather difficult, purposive or judgemental sampling may be necessary to cover most impairments. Charities who work with people with CCN may be able to initiate contact to potential participants. If incentives are to be given to participants, make sure these do not interfere with benefit rules. However, often participants are happy to volunteer. Short notice cancellations due to illness are more likely to occur and should therefore be accounted for in the project plan. Before starting the actual sessions it is suggested to meet the participants and their support workers to make the participant feel comfortable. After going through the participant information sheet, informed consent can be obtained by asking yes/no questions to confirm that potential participants understand and agree to the study. In particular questions to make sure they want to participate and know how they will contribute to the study may not be missed. 							
During the sessions							
 At the beginning of each user session the previous session should be reviewed to help the participants recall the steps undertaken so far. During the sessions, attention should be paid that the work is not too tiring and extra time for breaks should be factored in. The style of speech should be neither too complicated or technical nor condescending, but participants should feel as equals with the designers. In order to overcome hesitations of the participants when giving requirements to researchers, group discussions might be a solution. Allow plenty of time for participants to answer questions. Provide participants with storyboards or videos on current problems to help them imagine possible solutions. Also encourage users to imagine how other potential users would interact with the system. Create tangible prototypes with participants rather than sketches as these are impossible to draw with reduced dexterity. Explicitly encourage participants to give negative feedback explaining them that 							

finding flaws in the technology during the development will improve the quality of the finished product.

- The intrusiveness of video recording for the purpose of gathering feedback can be minimised if planned carefully.
- Evaluations are not limited to observations as participants may be trained to perform heuristic evaluations. Also encourage support workers to take part in the evaluations.
- Designers should try to include as many accessibility modes and adaptations as
 possible with the possibility of switching them on and off.
- Provide participants with a session summary and feedback at the end of each session.
- session.

After the session

- Encourage participants to "promote" the participation in the research study/design project among other potential end-users.
- In order to avoid a feeling of abandonment for the participant at the end of a study, possibilities of participation in other projects should be discussed.
- At the end of the study, personal feedback on the project and their contribution should be given to participants rather than a thank-you letter.

4 Requirement Gathering

This chapter begins with a description of the initial and particular steps required due to working with disabled people before the actual start of the software development project. This led to the development of the actual project plan which can be found in

266

Appendix **1.** After an initial idea finding process with the participants, market research was conducted and prototypes developed in order to specify the requirements.

4.1 Getting the Project Started

As the project"s participants would be disabled people, the author was required to apply for a **Scotland Disclosure**. This certificate issued by Disclosure Scotland, an Executive Agency of the Scottish Government, is designed to enhance public safety. It ensures that vulnerable groups in society, e.g. disabled people, are protected by checking that people wanting to work with them do not have a criminal history.

In addition, **ethical approval** needed to be sought from the School of Computing as the project would involve participants. The author informed the Ethics Committee that the project would be considered part of Suzanne Prior"s PhD thesis and was given positive response to pursue the project.

The **recruitment of participants** was facilitated by a speech and language therapist from the NHS (National Health Service). Having worked with the School of Computing before, she gave the author the opportunity to pitch the project idea to her clients with CCN in one of their fortnightly meetings allowing the author purposive sampling of her participants. As it could not be assumed that the potential participants of the project were literate, the author created an A3 poster outlining the project idea in Boardmaker symbols (see Appendix 2). After a successful presentation as such, the author found it difficult to say whether the potential participants liked the idea of the project or not. Although all the support workers were very much in favour of the project, the potential participants" facial expressions and body language did not give much away for the author.

Several weeks passed and further explaining to parents and other support workers, who were not present during the presentation, was necessary (see project summary in Appendix 3). In the end, seven participants were successfully recruited (see Table 3). Participant B joined the group of participants at the end of the project to perform a heuristic evaluation of the final prototype.

Parti- cipant	Age	Gender	Condition	Mode of Communication
A	24	female	Cerebral palsy	Dynavox AAC device, paper QWERTY keyboard, pointing at white board with a couple of options, severe dysarthric speech
В	30	female	Cerebral palsy	Lightwriter text-to-speech device, nodding and shaking head, severe dysarthric speech
с	46	female	Cerebral palsy	Palm-top communication aid with Bliss symbols broken> nodding and shaking head, severe dysarthric speech
D	50	female	Cerebral palsy	Lyo gazo
E	24	male	Traumatic injury due to car accident	Dynavox AAC device, thumb up and down
F	40	male	Cerebral palsy	Word map, nodding and shaking head, gestures, severe dysarthric speech
G	52	male	Cerebral palsy	Head switch scanning device, eye gaze
Н	71	male	Cerebral palsy	Dysarthric speech

 Table 3: Participant profiles.

4.2 Idea Finding Process and Getting to Know the User Group

The author used the first user session with each of the seven participants to introduce herself and to get to know the users – in particular, what they liked to do, how they communicated, if there was something on the communication side of things that frustrated them and if they could think of anything they wished existed. As an ice-breaker, the author showed the participants a PPT-slideshow with a few pictures of herself at different places in Scotland. The pictures were chosen with the intention that they should spark conversation, which they did.

With the initial questions leading to the topic of "information sharing software", the author learned and experienced that all of the participants had their ways of expressing their wants and needs whether by indicating answers to yes/no-questions with their head, their eyes or by thump up and down. Utterances were made, word mats utilised, high-tech AAC-devices used (operated either by hand or by head switch) as well as a paper QWERTY-keyboard.

All participants had in common that they liked to socialise - some in the traditional form of a night in a pub while others preferred Facebook (a social networking website). Another observation made by the author made that the participants seemed to enjoy it when their support workers or parents reminded them of past experiences. This became apparent to the author when seeing the participants moving idiosyncratically, i.e. rocking vigorously when excited.





Table 4 shows the four types of conversation according to Waller (1992) and gives examples for each classification. The participants and the author agreed that the first two types, i.e. formulaic conversation and basic need/wants, are managed to a satisfactory degree by all participants with their existing modes of communication. Summing up, the author and the participants came to the conclusion that there was room for improvement regarding "reusable conversation" telling personal stories. Literature confirms that telling stories is important in life as it gives meaning to life and helps to create one"s self-concept (Polkinghorne, 1995, Waller, 2009). This in mind, the idea of developing a talking photo album was born. The four main findings leading to the idea were:

- All participants enjoyed looking at the author"s photos during their first session.
- One of the participants turned out to be a keen photographer (having a camera that can be mounted onto her wheelchair).
- All participants liked to share past experiences.
- One participant wished to have a text-to-speech generator/screen-reader as her literacy was very limited.

4.3 Market Research

This section describes the market research undertaken to investigate what kind of talking photo albums for people with CCN already exist. An evaluation of a commercially existing photo album follows and lays open several design flaws of a talking photo album advertised as communication aid.

270

4.3.1 Commercially Available Talking Photo Albums

An extensive internet research has shown that there is a considerable number of talking photos albums on the market, the vast majority highly resembling each other and identical in their functioning (Quality Enabling Devices Ltd, 2010, Augmentative Communication Inc., 2010, Needapresent.com, 2010, Talking Products Ltd, 2010, TERS ELECTRONIC LTD, 2010). Pictures are held in individual transparent pockets with a squeezable button to activate a recorded message. To record a message, the record button and the activation button need to be pressed simultaneously while the user has to speak into the photo album"s build-in microphone to record a message.

Only one digital photo album was found (Photos-Speak Company, 2010). In this album an audio file is attached to each picture.

A different approach to a talking photo album is the so called V Pen (Voice Pen). It comes with a voice symbol software that makes it possible to print sound onto paper by adding a nearly invisible bar code on the print which can then read by the V-Pen (Ability World Ltd, 2010).

It is interesting to note that talking photo albums are sold in the context of communication aids for people with communication impairments (Augmentative Communication Inc., 2010, Inclusive Technology Ltd, 2010, Quality Enabling Devices Ltd, 2010, Ability World Ltd, 2010) as well as in that of photo gifts in (Needapresent.com, general 2010, Talking Products Ltd, 2010. TERS ELECTRONIC LTD, 2010). In both application areas the concept of talking photo albums highly appealing customer reviews seems users as 5-star to (Amazon.com, 2010) and feedback from participants of this study demonstrate.

4.3.2 User Testing and Evaluation of a Commercially Available Talking

Photo Album

Apart from the recently launched V Pen which follows a different approach, it is surprising that there are no adjustments made to the photo albums to make them more user-friendly for their target user group when advertised as a communication aid. In order to evaluate its usability for people with CCN, one of the highly resembling talking photo albums was bought (see Figure 5).





In individual sessions, the user groups, each consisting of a person with CCN and their support worker, were asked to test the setup of the photo album and the actual use by the AAC user. Subsequently, the groups were given the task to perform a heuristic evaluation in order to find usability problems in the design. The evaluators (user group 1 - 4) scrutinized the photo album with regard to (1) physical appearance, (2) appropriateness for users with CCN, (3) fulfilment of purpose, (4) ease of use, and identified a number of problems (see Table 5).

272

Table 5: Usability problems identified in heuristic evaluation.

	p4	р3	p2	p1
Usability Problems	rou	rou	rou	rou
	er	er	er	er
		Js g	Us ^g	Us ^g
y robust	No			
ings can be easily deleted by	Re			

Recordings can be easily deleted by accident Nearly impossible to handle with reduced dexterity Many steps required to create an album No interactive conversation Setup requires speech Cannot be created independently by AAC user Buttons are very small Recording time too short

Each of the blue cells in Table 5 indicates the finding of a usability problem by one of the user groups. The table shows that on the one hand, there is an overlap between the usability problems identified by the different user groups. On the other hand, some problems have only been found by one or two user groups. This demonstrates that it is worthwhile having several independent evaluators which Nielsen (2010b) confirms in his analysis of the number of evaluators required for an evaluation. For this reason, the author has limited the number of evaluators to four.

273

The evaluated talking photo album was perceived as *not very robust* by one of the evaluators, a characteristic considered important if the book was targeted at people with CCN as spastic movements might destroy the album involuntarily.

Recordings can be easily deleted by accident – the reason for this problem lies in the fact that when the play button on a photo page is pressed during the recording process, buttons on pages underneath the current page might be pressed as well and so other photo descriptions might potentially be overwritten. To overcome this potential risk, the photo page that is meant to be described needs to be lifted up to ensure that only one play button is pressed at a time. This, however, is very unlikely to be feasible for people with CCN.

The fact that the album is *nearly impossible to handle with reduced dexterity* refers partly to the previously described problem but also to the complicated procedure of recording a message which involves two small buttons being pressed simultaneously as described in section 4.3.1.

One user group pointed out that *many steps are required to create an album.* The picture needs to be taken, it needs to be developed, potentially picked up in a store, it needs to be inserted into the photo album and finally a message needs to be recorded. It was pointed out that nearly all of these steps constitute actions which would need to be carried out for and not by the person with CCN. Furthermore, it was observed that the commercially available photo album provided *no interactive conversation* as such as everything about the picture would be said in one go - missing out the concept of turn-taking completely.

274

Setup requires speech – at a first glace this problem seemed almost ironic. A communication aid for people who cannot speak that requires speech input. However, many communication aids require a setup by an able-bodied person.

It was concluded that the examined photo album **cannot be created independently by AAC users.** The reason for this lies predominantly in the fact that speech input is required for the setup, but also in the fiddly play button that needs to be pressed in order to activate the speech.

This latter point – *buttons are very small* – was explicitly identified by two of the four user groups. Everyone agreed that this was an issue that should be addressed when designing a device for people with reduced dexterity.

Finally, it was noted that the *recording time is too short.* Ten seconds do not allow describing the story of a picture in much detail.

4.4 Prototyping and Evaluation

Following the first user sessions and the definitions of the theme, several prototypes were created and evaluated.

4.4.1 Co-design Sessions

In preparation of the co-design sessions, participants were provided with a range

of photos from which they were asked to choose two. They then had to imagine

they had taken the photos themselves and were showing them to a friend telling him/her all about it. The photo descriptions were analyzed as to what questions had been answered by the descriptions of the photos. The results were the following questions: where, when, what, who, feelings.



Figure 6: Users with CCN during individual co-design session.

The next round of meetings represented the major **co-design sessions** during which each participant created a user interface prototype of The Talking Photo Album application. Paper squares with question symbols (the five questions identified in the previous session), foam buttons in different colours and shapes as well as a range of photos and a white board were provided. Building physical

prototypes rather than sketches, as it is common procedure with able-bodied people, proved to be a viable solution. The participants enjoyed this active work (see Figure 6). Depending on the degree of the participant"s disability the buttons were placed onto the board through the participants" own hands, through support workers after the participants with CCN had chosen one of many options given to them or in one case through eye gaze and with help of a transparent tray. The results from these individual sessions can be seen in Figure 7.



Figure 7: Results from co-design session of photo album screen with users with CCN

Prototypes number (1) and (2) show a clear spatial separation of the question

buttons, the three facial expression feeling buttons and the Yes/No buttons. This

differentiated arrangement of buttons corresponds to the cognitive capabilities of the participants experienced by the author during the duration of the study. Prototype (3) has a neat arrangement of buttons, but there is no distinction made between the different categories of buttons. The last three prototypes do not seem to follow any concept. The conclusion drawn from this exercise by the author was that spatial separation of buttons mattered more to the participants than attributing each button category a particular colour.

To test whether the participants would understand the concept of an interactive photo album using a touch-screen, the author conducted a **Wizard of Oz experiment**. A photo description was prepared for the session assistant (the "wizard") who was to simulate the voice of the photo album application. A third person was invited to have a conversation with the participant with CCN about the picture. The participant with CCN pressed the appropriate foam buttons on the interface prototype of the photo album, thus triggering the "voice" of the photo album and holding a successful conversation about the photo.

4.4.2 Usability Button Game

A general concern was the choice of the right button size. This was an important decision as the users with CCN and reduced dexterity should not become frustrated due to poor usability when using the photo album application. For this reason the author developed the Usability Button Game (see Figure 8) with three button sizes.

As the imaginary touch-screen in the Wizard of Oz Experiment was successfully used by the participants, the author chose to use the Panasonic Toughbook H1 Health (see Figure 9) available within the research group because of its touch-

screen functionality and its robustness.



Figure 8: Usability Button Game (with 3 button sizes).

In the Usability Button Game, the nine coloured squares in the middle of the screen ask a question when pressed which the user with CCN had to answer by touching the corresponding button around the outer edge just like in the photo album application to be developed. All three screens (Figure 8 (1-3)) were tried out with regard to ease of use. The participants with CCN unanimously preferred the medium button size. Prior concern from colleagues of the author that the participants with CCN might feel ridiculed when asked to play this game and evaluate the button size did not occur, quite the contrary happened. The

participants enjoyed to hear the voice saying "That"s correct!" each time they pressed the right button and did not mind the voice saying "Sorry, please try again." when they had chosen an incorrect button.

However, some participants had problems using the touch-screen without a keyguard which they were used to from other AT devices. A keyguard is a clear plastic board with holes corresponding to buttons on the touch-screen onto which it is placed. It supports the user"s hand and prevents that they press buttons by mistake.

4.4.3 Keyguard

In order to make the application accessible to a wider audience, the author created a keyguard out of Perspex with the help of a laser cutter. The first keyguard prototype can be seen in Figure 9 (1 + 2). It is propped up with toothpicks to avoid the Perspex touching the screen which would lead to permanent triggering of the touch-screen mechanism. The second keyguard prototype (Figure 9 (3)) is slightly larger than the screen of the Toughbook and is, therefore, resting on the slightly higher brim of the screen. Furthermore, edges were re-adjusted to the shapes on the screen and made round to make them smoother to touch.

35



Figure 9: Keyguard prototype #1 (1 + 2) and #2 (3) for the Toughbook.

4.4.4 PowerPoint (PPT) Prototypes

An interactive Microsoft Office PowerPoint slideshow with a text-to-speech generator was chosen as next prototype medium as it was considered relatively simple to develop, but very effective in its use.

Figure 10 (2) shows the text description entry screen, (3) shows the feeling radio buttons and (4) shows the actual photo album screen with Yes/No buttons and an exit button at the top of the screen, the photo in the middle with Previous/Next buttons on either side, the feeling button at the right side and the remaining Where/When/Who/What buttons below the photo. Participants were able to press the buttons on the photo album screen which played back the descriptions which were entered beforehand.




While the participants tried out and tested this first PPT prototype, suggestions for improvement were made by all parties involved. The Previous/Next buttons proved to be too narrow and as such not easy to use by the participant with CCN due to their reduced dexterity. The participants with CCN pointed out that they would not require Yes/No buttons as they were able to communicate agreement or disagreement without the help of buttons. The author and support workers noticed that the communication via this prototype tended to be rather one-sided. The participants with CCN always only replied to questions they were asked by their conversation partner.

The above mentioned suggestions for improvement were taken into consideration in the second PPT prototype (see Figure 11). Improved features were the enlargement of Previous/Next buttons, the removal of Yes/No buttons and the addition of a Question button. This Question button, when pressed, turns the audio buttons for the next button push into question mode and allows users with CCN to ask their conversation partner a question. The event is announced when the Question button is pressed with "I"ve got a question for you.".



Figure 11: PPT-prototype #2 – photo album screen.

When in question mode the Where button uses the location description of the current photo and asks when pressed "Have you been to <location of current photo>?". The When button picks up the date of the current picture: "What did you do in <date of current photo>?. The following three buttons have standard questions when in question mode, i.e. "Who was with you?", "What did you do there?" and "How did you like it?".

The evaluation process was guided by a questionnaire for both PPT-prototypes and the usability button game (see Appendix 4). While the appearance of the PPTprototype #2 was rated positively by the participants, it became obvious that a more advanced facility was required to administer multiple pictures. Furthermore, the Yes/No buttons were reconsidered and finally re-introduced as they were thought to be useful in situations where the user with CCN is showing The Talking Photo Album to someone not familiar with their way of saying yes and no.

4.5 Requirement Specifications

Leading on from the market research, co-design, prototyping and evaluation activities, requirements for The Talking Photo Album application were specified.

There are two different kinds of requirements in software engineering: functional and non-functional. While the former describes tasks a system must perform, the latter consists of constraints upon a system and/or the developer/designer (Wiegers, 2003). Usability is a non-functional requirement (Lauesen and Younessi, 1998). As usability requirements were of particular interest in this project, they are listed separately.

Gathering requirements from the end-users in this project was challenging due to the communication limitations. Therefore, family and support workers were equally consulted.

Table 6: Functional requirements for The Talking Photo Album.

Functional Requirements

FR1: Login procedure

Description: The Talking Photo Album shall require a user to login using a username and password.

Rationale: Prevents unauthorised access to photos of others, e.g. in care homes.

FR2: Choice of options

<u>Description:</u> After the user is logged-in, the system shall provide the user with a choice of uploading a photo, describing a photo and viewing the photos in the album and listening to their description.

Rationale: Direct access enables users to directly jump to the task they want to carry out without running through a particular sequence of steps.

FR3: Photo upload (user perspective)

<u>Description:</u> The system shall provide the user with a possibility of uploading a picture available anywhere on their device/computer (hard drive, USB-stick, CD) into The Talking Photo Album.

Rationale: Users need to be able to insert photos into the system. Different upload sources provide flexibility.

FR4: Photo upload (system perspective)

<u>Description:</u> The system shall make a copy of the photo that is to be inserted into the system in a folder that is created with the name "The Talking Photo Album" in "My Documents".

Rationale: Saving a copy in a specially created folder for the photo album ensures that the photo is still available when the data carrier is removed from the device/computer or the location where the photo was stored is changed.

FR5: Photo album description setup

<u>Description:</u> The system shall allow the user to save descriptions for each photo for the following questions: where?, when?, who?, feeling? Rationale: These questions were found to be the ones generally asked when looking at

Rationale: These questions were found to be the ones generally asked when looking at photos.

FR6: Auditory feedback (in description setup section)

<u>Description:</u> All symbol buttons in description setup section shall give auditory feedback. <u>Rationale:</u> This makes the photo album accessible to a wider range of users with CNN who might be not familiar with the symbol system used.

FR7: Photo loop functionality in the description setup

Description: The system shall allow the user to go forward and backward through the photos in an endless loop to choose for which photo they want to edit the description. Rationale: Creating a loop allows the user to get to photos at the end of the photo album quicker by going backwards than moving through the whole album to get to the end.

FR8: Photo loop functionality in the album

Description: The system shall allow the user to look at the photos going forwards and backwards in an endless loop.

Rationale: Creating a loop allows the user to get to photos at the end of the photo album quicker by going backwards than moving through the whole album to get to the end.

FR9: Introduction phrase

Description: The system shall allow users to initiate a conversation about the photos they want to show by saying so.

Rationale: Users with CCN should have the ability to initiate a conversation about their photos.

FR10: Transition phrase

Description: The system shall allow users to move a conversation to the next/another photo they want to show by saying so.

Rationale: Users with CCN should be in control to say when they want to move to the next/another photo.

FR11: Playback of descriptions when corresponding button is pressed

Description: The system shall speak the description entered in the description setup section when the corresponding symbol button is pressed.

Rationale: Allows users to say specific details or answer particular questions.

FR12: Use of symbols to convey information

Description: The system shall not require literacy and use symbols in addition to written labels.

Rationale: A high percentage of people with CCN has no or only limited literacy skills and often uses symbol systems instead.

FR13: Photo album with questions buttons

Description: The system shall have individual buttons for the questions: where?, when?, who?, what?, feeling?

Rationale: These split-up descriptions foster turn-taking in the conversation which is considered desirable as it is common in conversations between people without speech impairments.

FR14: Possibility for the user with CCN to ask questions

<u>Description:</u> The system shall allow its users to indicate that they have a question and to ask questions regarding one of the five photo description divisions: where?, when?, who?, what?, feeling?

Rationale: This feature makes the "photo conversation" resemble a conversation between people who don"t have a speech impairment.

FR15: Go-back functionality

<u>Description:</u> The system shall allow the user to return to the action choice overview page. <u>Rationale:</u> This feature allows the user to navigate back and forth through the application.

FR16: Logout functionality

Description: The system shall have an exit/logout possibility to "close" the photo album on each interface screen.

Rationale: The user should be able to exit the photo album at any time.

FR17: No speech input required for setup

<u>Description:</u> The system shall not require speech input to setup the descriptions for the photos.

Rationale: If users are literate, they should be able to setup the photo album on their own and without help from a third person.

FR18: Offline availability

<u>Description:</u> The Talking Photo Album shall be available offline. Rationale: Does not limit access to users with an internet connection.

FR19: Operational on different Windows operating systems

<u>Description:</u> The Talking Photo Album shall be fully functional on Windows XP, Windows Vista and Windows 7.

Rationale: These are the most common Windows operating systems in use, so makes the program accessible to the majority of users.

Table 7: Non-functional requirements for The Talking Photo Album.

Non-functional Requirements (1)

NFR1: Time frame of project Description: The project shall be presented on 1st September 2010. Rationale: Specification of the MSc Project – Submission Guidelines.

NFR2: Purpose of the software to be developed

Description: The finished software application shall give people with CCN the opportunity to share information.

Rationale: Specification of the Master thesis subject/title.

NFR3: Choice of database

Description: The database shall be build using Microsoft Office Access. Rationale: Microsoft Office is a widely-used application suite and includes MS Office Access which is a pseudo-relational database management system which does not require internet access This is of importance as internet access is often restricted in care homes where a proportion of potential end-user live.

NFR4: RAM requirement for photo album application

Description: The system should not require an excessive amount of RAM. Rationale: The photo album application should be usable on a wide variety of devices.

Table 8: Usability requirements for The Talking Photo Album.

Non-functional Requirements (2) Usability

UR1: Comprehensibility of error messages

Description: Error messages generated should be clear, informative and concise. Rationale: Gives information to the user as to how the error message occurred and how to avoid similar errors in the future.

UR2: Robustness of the system with regard to input

Description: The system shall be robust regardless of input where possible. Rationale: Prevents errors occurring due to incorrect user input. This is necessary as some characters have a special meaning in the languages used in development.

UR3: Confirmation messages

Description: Users should be informed that a task has been completed successfully through a confirmation message.

Rationale: Displaying a confirmation message of the task being completed increases the satisfaction and confidence of the users in the system.

UR4: Installation of The Talking Photo Album on robust device

<u>Description:</u> The Talking Photo Album shall be installable on a robust device that is not easy to destroy involuntarily.

Rationale: The device needs to be suitable for users with CCN who have reduced motor control and reduced dexterity.

UR5: Physical size and weight of devices for the photo album

<u>Description:</u> The Talking Photo Album shall be installable on portable devices and thus do not have excessive sizes and weights.

Rationale: The photo album should be easy to take along/to transport.

UR6: Appropriateness for users with CCN

Description: Users with CCN shall be able to operate the finished photo album on their own.

Rationale: The application is meant to foster (independent) communication of users with CCN.

UR7: Size of buttons on the actual photo album

Description: Users with CCN shall be able to comfortably select a target button on the actual photo album.

Rationale: The photo album application should be physically easy to use by users with CNN.

UR8: No limit on description length

Description: Users shall be able to enter descriptions as long as they wish without any restriction in length.

Rationale: The user should not feel restricted in any way when describing the photo.

UR9: No extensive training for the actual photo album

<u>Description:</u> Users with CCN shall be able to use the actual photo album without any extensive training.

Rationale: The steps to carry out for using the photo album should be easy to learn for users with CCN.

UR10: Colour and design of The Talking Photo Album

<u>Description:</u> The colour of the photo album shall have a plain and minimalistic design. <u>Rationale:</u> Users shall not be distracted and/or irritated by the colour or design of the photo album.

UR11: Memorability of how to use the album

<u>Description:</u> The user shall be able to remember the necessary steps to use to photo album after a period of not using the system.

Rationale: The Talking Photo Album is an application which is most likely not used on a daily basis. Therefore, users should be able to easily re-establish proficiency as otherwise the application might be abandoned.

UR12: User-comfort while using the photo album

<u>Description</u>: The user shall be relaxed and confident when using the photo album application.

Rationale: It should be pleasant for users to use The Talking Photo Album and not require too much effort as otherwise they may not want to use the system.

During the time-span of the project, the requirements continually evolved and

reassessments were made. The requirements listed above represent those that

were taken into consideration in the creation of the final prototype. Some

requirement had to be abandoned due to time constraints. In particular, these

were the photo delete function, and the username and password creation.

5 Software Design and Implementation

5.1 Use Cases

Use cases are an informal way to describe requirements in a narrative format.

More precisely:

"A use case is a description of the possible sequences of interactions [and variants thereof] between the system under discussion and its external actors, related to a particular goal." (Cockburn, 2000, p.15)

For the photo album application, three actors were identified: the support worker (see Figure 12); the illiterate user with CCN (see Figure 16); the user with CCN and literacy skills (see Figure 17); and. In the following, use case diagrams for each actor, descriptions of the use cases with their basic and alternative flow and supplementing interface sketches are presented.

Support Worker



Figure 12: Use case diagram for support worker.

Table 9: Use case 1 for support worker.



Alternative flow:		
UC1-AF1: Incorrect username and password Support worker/user with CCN has entered an incorrect username and password.		
System:	Displays error message "Sorry, you username or password is invalid!", thus informing the support worker why the login failed. Return to basic flow of events where support worker enters username and password.	



Figure 13: Interface sketch of option screen.

Table 10: Use case 2 for support worker.

Use Case 2		
Possible basic flow:		
UC2-BF: Upload of photos The use case begins when the support worker opens The Talking Photo Album application.		
System: Support Worker: System: Support Worker:	Perform The "Login" use case Selects the option to upload photos. Displays new screen with two numbered buttons: 1 - Browse for photo, 2 - Photo upload (see Figure 14). Selects button "1 - Browse for photo". (A1)	

System: Support Worker: System:	Opens a file open dialogue box. Selects photo file for upload into The Talking Photo Album. Presses button "2 – Photo upload". Displays message "Upload successful!" (A2)	
	launa.	
Possible alternative r	IOWS:	
*1 – Support worker The support worker h	r exits has the option to exit the photo album application at any time.	
Support worker: System:	Selects the exit-button. System returns to login page.	
*2 – Support worker returns to the welcome/home page The support worker has the option to return to the welcome/home page at any time.		
Support worker: System:	Selects the back-button. System returns to welcome/home page.	
UC2-AF1: Wrong order of buttons Support worker has pressed button "2 – Photo upload" prior to searching and selecting a photo with button "1 – Browsing for photo".		
System:	Displays error message "Please first search and select a photo using button 1".	
	Return to basic flow of events where support worker selects button "1 – Browse for photo".	
UC2-AF2: Upload of existing photo Support worker has selected a photo for upload that already exists in The Talking Photo Album.		
System:	Displays error message "Photo already exists in photo album!". Return to basic flow of events where support worker selects button "1 – Browse for photo".	



Figure 14: Interface sketch of upload screen.

Table 11: Use case 3 for support worker.

Use Case 3		
Possible basic flow:		
UC3-BF: Setup of photos The use case begins when the support worker opens The Talking Photo Album application.		
System: Support Worker: System: Support Worker: System:	Perform The "Login" use case Selects the option "Setup". Displays new screen (see Figure 15). Describes the photo by completing the answer to the question written underneath photo (in the text box) and selecting the save-button. (A1, A2) Commits the entry to the database and displays the next of the five questions for each picture.	
Possible alternative flows:		
* 1 – Support worker exits The support worker has the option to exit the photo album application at any time.		
Support worker: System:	Selects the exit-button. System returns to login page.	

UC3-AF1: Different picture The Support worker wants to describe a different photo than currently displayed.		
Support worker: System:	Moves to the desired photo using the previous- and next-button on either side of the photo. Interface moves to desired photo. Return to basic flow of events where support worker describes the photo by completing the answer to the question written underneath photo (in the text box) and selecting the save-button.	
UC3-AF2: Change of The Support worker w Support worker: System:	description Teants to change a description of a photo that already exists. Moves to the desired photo using the previous- and next-button on either side of the photo. Interface moves to desired photo and allows the support worker to overwrite the existing description. Return to basic flow where the support worker selects the save button.	



Figure 15: Interface sketch of setup screen.

Illiterate User with CCN



Figure 16: Use case diagram for illiterate user with CCN.

Table 12: Use case for illiterate user with CCN.

Use Case 1		
Possible basic flow:		
UC1-BF: Operating the photo album, i.e. showing and talking about photos in the album The use case begins after photos have been uploaded and described in The Talking Photo Album and the illiterate user with CCN is given the device with the actual photo album screen (see Figure 17).		
Illit. user with CCN: System: Conversation partner: Illit. user with CCN: System: Conversation partner: Illit. user with CCN System: Illit. user with CCN System:	Starts the conversation by clicking on the photo. "Would you like to see some pictures?" "Yes, let"s look at some photos." Presses where-button. "This in the Cornwall." "It looks pretty, did you go there on holiday?" Presses the yes-button. "Yes." Presses what-button. "I was visiting family."	

Conversation partner	"Who went with you?"			
Illit user with CCN.	Presses who-button			
System:	"I went there with my parents."			
Conversation partner:	"Was it this summer?"			
Illit. user with CCN:	Presses no-button.			
System:	"No."			
Illit. user with CCN:	Presses when-button.			
System:	"This was in the summer of 2008."			
Conversation partner:	"Seems to have been a lovely summer day?"			
System:	"Yes "			
Illit user with CCN.	resses feeling-button			
System:	"I really liked the warm weather."			
Possible alternative flow	'S:			
*1 - Next photo with ag	reement from conversation partner			
The illit. user with CCN I	has the option to ask their conversation partner if he/she wants to			
see the next picture at a	ny ume.			
Illit. user with CCN:Pres	ses the picture.			
System: Vouid you like	to see another picture?"			
Ulit user with CONPress	es the next-hutton			
System Displays the new	rt picture			
*2 - Next photo without	t prior announcement			
I ne IIIIt. User with CCN r	has the option to move to the next photo at any time.			
	Dresses the payt button			
System:	Displays the next picture			
	Displays the next picture.			
*3 – Illit. User with CCN	wants to ask a question			
I he illit. user with CCN I	has the option to ask their conversation partner a question at			
any time.				
Illit. user with CCN	Presses the question-button.			
System:	"I"ve got a question for you."			
Illit. user with CCN	Presses the where-button.			
System:	System goes into question mode and adds the location of the			
	current picture into the standard partial question "Have			
	you been to?". The partial questions are button specific as are			
	the words that are filled in.			
	"No, but I would roally love to!"			
Conversation partner:				
· · · · · · · · · · · · · · · · · · ·				
*4 – Illit. User with CCN wants to exit the photo album				
The illit. user with CCN has the option to exit the photo album at any time.				
Illit. user with CCN	Presses the exit-button.			
System:	System returns to login page.			



Figure 17: Interface sketch of photo album screen.

User with CCN and Literacy Skills



Figure 18: Use case diagram for user with CCN and literacy skills.

Depending on the degree of their disability, literate users with CCN will be able to perform the "Login" (use case) themselves and potentially the photo upload (use case) as the cognitive load should be manageable for them. As full literacy in people with CCN is scarce (Dahlgren Sandberg et al., 2010), users with CCN will be likely to require help/support for describing the photos (setup use case). This is in particular the case as the descriptions need to be grammatically correct sentences in order to be understood by conversation partners later on. As a result, there is no clear allocation of use cases for users with CCN and literacy skills, because the level of independence depends on the degree of disability of the individual. Generally speaking, a user with CCN and literacy skills will, therefore, be able to perform all use cases an illiterate user with CCN can perform and additionally some use cases that support workers will perform for illiterate users with CCN.

The development of these use cases allowed putting the functional requirements into a real context. Furthermore, their analysis highlighted alternative flows some of which had still to be incorporated into the design; these were in particular error messages helping the user to understand why an error occurred.

5.2 Overview of Program Architecture

Based on the prototypes, requirements and the use cases, a screen structure was defined (see Figure 19)



Figure 19: Screen structure of The Talking Photo Album application.

5.3 Entity-Relationship (ER) Diagram

When working with the early PPT-prototypes, it became clear that a system was required to upload and manage multiple images. It was decided that a database would be ideal to hold the required information the pictures as well as the standard fragments that would be used by the system.



Figure 20: ER-Diagram of The Talking Photo Album database.

The (photo) Data table contains all information that make up the descriptions of the photos which can be accessed through the Where/When/Who/What buttons in the application. Moreover, the table holds the file name of the photo and the path where it is stored. The *userID* is a foreign key which links the table to the **User table** with the users" login credentials. This provides users privacy as they can only see their own pictures.

The **setupQuestions table** contains the images of the different description buttons as well as the text that is read out when the button is pressed by people who have difficulties reading the question above the description entry field.

The **Descriptions table** holds the sentence beginnings for each description type. This was implemented to minimise the text entry required for each description.

The **Questions table** contains the questions or question fragments, which together with the description form a question. They are the questions users of the photo album can ask their conversation partners when the photo album is in question mode.

The **Picture table** holds a number of topic introductions which users of the photo album can trigger by touching the photo in order to ask their conversation partner if e.g. they would like to see some pictures or progressing to the next picture by saying e.g. "Look at this one.".

The author has chosen to create these tables even though the setupQuestions, Questions, Desciptions and Picture tables could have been equally hard coded. However, the idea behind putting this information into a database was that

58

changes to the phrasing of the standard text could easily be done without having to make changes in the code.

304

On reflection, one limitation of the current structure of the tables Data, Questions and Descriptions is that the information relating to the Where/When/Who/ What/Feeling buttons is organised in individual columns. This means that adding or changing questions requires changes to column headers throughout the code. Therefore, it could have been beneficial to introduce an additional table listing the question words and to add its primary key as foreign key into the Data, Questions and Descriptions tables.

5.4 Platform

As the Panasonic Toughbook H1 Health was successfully used by the participants in the prototype evaluating sessions, it was also used throughout the rest of the project.

As the Panasonic Toughbook H1 Health has a Windows XP operating system, it was decided to develop a Windows application. By default, Windows XP includes Microsoft Speech, an application programming interface that enables developers to integrate speech synthesis into their applications.

5.5 Programming Language

The first prototype started off with a PPT presentation which was later enhanced by adding ActiveX control buttons. The properties of the buttons were changed as required to adapt their appearance. Code in Visual Basic for Applications (VBA)

was then linked to the buttons to trigger actions such as voice output on clickevents or moving to the next page.

Although the use of PPT and VBA was very effective in the first and second PPTprototypes, it was considered insufficient for the final version created within this project. The main reasons for this being that with VBA, the program could not have been compiled to an executable file; the end user would be required to be in possession of MS Office; the photo upload would have been difficult to implement; and the source code would have been editable by the end user either on purpose or by accident. Therefore, an alternative programming language was looked for. As the only programming knowledge the author had were the two introductory modules to C++ of the MSc Applied Computing program, a related but more interface-orientated programming language was selected – C#. Using the highlevel programming language C# had the advantage that Visual Studio 2008, an integrated development environment form Microsoft, could be used. Particularly useful was the included forms designer for building graphical user interface applications.

5.6 Database

MS Office Access was chosen as database management system as it required no additional infrastructure (e.g. a database server) apart from the freely available MS Access Runtime. Furthermore, it does not require the user to be online while using The Talking Photo Album application which was considered of advantage as users can not be assumed to have internet access, e.g. in care homes.

305

60

5.7 Code Overview

The final version of the photo album created within this project consists of the screens: Login screen, Options screen, Photo Upload screen, Setup screen and Photo Album screen. In the C# code each screen is represented by a form. The following excerpt (Figure 21) is taken from the Photo Album screen. The *speak* function combines two parts of text, i.e. the standard beginning of a photo description and the individual description entered by the user. In the event of the *Question button* being pressed, the two parts of text being combined are the question beginning and the referral to the picture such as in the case of the *Where button* "Have you been to <location of picture>?".

```
using System.Windows.Forms;
using System.Data.OleDb; // required for connection to database
using SpeechLib; // required for speech output
    private void speak(DataTable part1, int columnPart1,
      DataTable part2, int columnPart2, int rowPart2)
    {
      SpVoice voice = new SpVoice();
      string text = String.Empty;
      text += part1.Rows[0].Field<string>(columnPart1) + " " +
      part2.Rows[rowPart2].Field<string>(columnPart2);
      voice.Speak(text, SpeechVoiceSpeakFlags.SVSFDefault);
    private void btnWhere Click(object sender, EventArgs e)
    {
        if (selectQuestion == true)
        {
          speak(tblQuestions, 1, tblData, 3, r);
          btnQuestion.BackgroundImage =
          global::frScratch3.Properties.Resources.Question;
          selectQuestion = false;
        }
        else
        {
          speak(tblDescriptions, 1, tblData, 3, r);
    }
```

Figure 21: Code excerpt from the photo album screen.

As the photo album application requires access to the database in a number of different situations, the class *accessDB* was created to avoid repetition of code (see Figure 22). To retrieve data from the database, objects available in Visual Studio such as *OleDbCommand, OleDbCommand* and *OleDbDataAdapter* were used.

```
public class accessDB
[...]
  public DataSet getDataSet
  {
    get
    {
      DataSet myDS = new DataSet();
      if (File.Exists(strFile))
      {
        OleDbConnection myConn = new OleDbConnection();
                                // instantiates connection object
        myConn.ConnectionString = "Provider=" +
        "Microsoft.Jet.OLEDB.4.0;Data Source="
                                 // provider for MS Access DB
        + strFile + ";";
        myConn.Open();// opens the connection to the
                                    database
        OleDbCommand myCMD = new OleDbCommand();
                                 // instantiates command object
        myCMD.CommandText = strQuery;
        myCMD.Connection = myConn;
                                    specify connection for command
                                 11
                                    to retrieve data
        OleDbDataAdapter myAdapter = new OleDbDataAdapter();
                                 // prepare DataAdapter
        myAdapter.SelectCommand = myCMD;
                                 // connects DataAdapter to command
                                    object
        myAdapter.Fill(myDS, "Data");
                                 // fills the DataSet with data from
                                    the database
        myConn.Close();// closes the connection to the
                                    database
      }
      return myDS;
    }
  }
}
```

Figure 22: Method to retrieve data from the database.

6 Testing

Testing is performed with the intention of finding errors in software and is consequently a prerequisite to their elimination, which in turn is vital when striving for software of high quality. There are different testing techniques each looking at different aspects of the software program. For The Talking Photo Album, Black Box Testing and White Box Testing have been undertaken at the different levels of the system. Tests such as unit testing, integration and system testing, usability testing and acceptance testing were completed as part of the iterative life-cycle. In the following, the testing techniques employed are described in more detail.

6.1 Testing Techniques

White Box Testing is based on the source code of a program and therefore requires a knowledgeable tester. It intends to find faults in the program structure and is therefore also referred to as Structural Testing. The tester has to ensure that every line of code is executed at least once, i.e. all possible ways through the code using the conditional statements, loops, and Boolean data types (Agarwal et al., 2010, Spillner et al., 2007). White Box Testing helped the developer to eliminate typographical and logical errors as well as redundant code. Moreover, it was employed to confirm that data was transferred correctly to and from the database.

In **Black Box Testing** the source code, as the name suggests, is treated as a "black box" and only input and output are considered. It could therefore be performed by a tester with no programming knowledge. Black Box Testing is based on the functional requirement specifications which is why it is also referred to as Functional Testing. For The Talking Photo Album, Black Box Testing was

primarily performed by the developer when testing the application before performing the cognitive walk-throughs with the participants.

310

For The Talking Photo Album **walk-throughs**, a form of group meetings with the intention to find problems but not to solve them, were also performed. The main reason for this choice was that in contrast to other methods, such as technical reviews, no preparation from the user groups was required. A walk-through was conducted for both PPT-prototypes and for the final C#-version. In individual sessions, the author presented the application to each user and their support staff while stepping through each screen of the application. The intention of this task was to communicate information about each screen to the user as well as to discover defects. Furthermore, the walk-through served as preparation for the heuristic evaluation which will be discussed at the end of this chapter.

6.2 Summative Evaluation

In a commercial software development project, acceptance testing is commonly performed by the customers when receiving the end-product from the developers. As the Talking Photo Album was developed as part of a research project, this was not possible. Instead, acceptance testing took the shape of a summative evaluation.

A person with CCN who had not previously been involved in the project conducted a heuristic evaluation according to the 10 usability principles (the "heuristics") by Nielsen (2010a). The explanation for each heuristic can be found in Appendix 5.



Figure 23: The final prototype of The Talking Photo Album on the Toughbook.

The results of this systematic inspection were as follows:

1. Visibility of system status

The evaluator pointed out that at the start of the photo album application, they felt a bit lost as there was no introduction how to use the photo album.

2. Match between system and the real world

The Talking Photo Album was considered to speak the users" language with the words and symbols that were chosen.

3. User control and freedom

The evaluator found that even though it was possible to undo/redo e.g. picture descriptions, it was not obvious for the novice user without consulting the manual (that was still to be developed at the time of the evaluation).

4. Consistency and standards

This heuristic was considered to be fully met – words/symbols and actions were the same across all pages.

5. Error prevention

According to the evaluator the exit button could be enhanced with a question of some sort that asks the user whether they are sure they want to quit the application in order to avoid unintentional closure of the program.

6. Recognition rather than recall

The system was regarded as easy to memorize, but for the sake of completeness a list of instructions were found to be helpful.

7. Flexibility and efficiency of use

This heuristic was considered as fully met.

8. Aesthetic and minimalist design

The design was found appealing containing no irrelevant information.

9. Help users recognize, diagnose, and recover from errors

Error messages were considered to be in plain language and easy to understand. It was pointed out that a help button might be helpful.

10. Help and documentation

The evaluator was not provided with a user guide for the application as it did not exist yet at the time of the evaluation. The idea of such a booklet was given favourable opinion.

7 Conclusions

7.1 Summary

In the creation of The Talking Photo Album, people with CCN and their support workers were involved from the very beginning of the project. The project title "Participation of People with Complex Communication Needs in Designing Assistive Technology Software for Information Sharing" was formulated vaguely on purpose as the participants of the study should be involved in identifying the project remit. Keeping Prior"s (2010b) guidelines in mind, the product idea was found with the help of informal semi-structured interviews. Market research was undertaken to see what was currently on the market and to evaluate one of the commercially available talking photo albums. In co-design sessions, user interface prototypes were built by the participants and the suitability of a touch-screen with audio symbol buttons evaluated through a Wizard of Oz experiment. The evaluation of two PPT-prototypes and a usability button game through the participants followed to arrive at a user-friendly interface and an application that allowed interactive conversation between the photo album user and their conversation partner. The Talking Photo Album was implemented in C# and included an MS Office Access database to upload and store the photos and manage their descriptions (see Figure 24). At the end of the study a heuristic evaluation was performed through one of the participants with CCN. To conclude, the aim to develop a software system for and with adults with severe disabilities that would allow them to share information was achieved.



Figure 24: Final C# version of The Talking Photo Album.

7.2 Critical Appraisal of the Project

This project demonstrated successfully that participatory design methods can be

used with people with CCN. It is suggested that this method should be used more

often to develop software applications for users with CCN as this may help to develop software and devices that will meet their needs more precisely and that will not be abandoned. It also became clear that there is a need to develop software and devices that enable users with CCN to share life-experiences, i.e. communication that goes beyond the exchange of basic information.

On the administrative side of the project, it would have been beneficial to request contact details from the participants right from the start to organise sessions with their support worker directly rather than through their speech and language therapist who provided the initial contact.

The programming language C#, which was used to develop The Talking Photo Album, proved to be a very versatile programming language to create user interfaces and in particular to customise buttons even though the author was not familiar with C# before starting the project.

Two features which were not implemented due to time constraints but which would clearly complete The Talking Photo Album at its current state are a delete and a user account generation facility.

The user sessions were a valuable experience for the author although getting the balance between a spontaneous flow and predetermined structure right was sometimes challenging.

Furthermore, it should be noted that the author had the impression that the participants enjoyed to be consulted and being actively involved in the study. This made it fun and rewarding for the author to work with them. Following the invitation

of the author, there was a large attendance of participants at the public demonstration of the software created during the project (Figure 25).



Figure 25: Presentation of the master project with participants and support workers listening.

7.3 Recommendation for Future Work

With an increasing amount of photos in the photo album, the need for a photo organisation facility is like to arise. A possibility could be a screen with a bookcase and different photo album which could be viewed and maintained separately. Another useful addition to the photo album might be a tag and search facility to facilitate retrieving photos to a particular key word.
Other improvements could be an alternative information input for the photo descriptions through voice recordings or a multiple photo upload.

Last but not least, a mobile version of The Talking Photo Album would enable the use on pocket-size devices like mobile phones.

8 References

ABILITY WORLD LTD (2010) Introducing the V Pen. Retrieved 08/10/2010, from http://www.ability-world.com.

AGARWAL, B. B., TAYAL, S. P. & GUPTA, M. (2010) *Software Engineering & Testing,* Jones and Bartlett Publishers.

ALLEN, J. (2005) Designing desirability in an augmentative and alternative communication device. *Universal Access in the Information Society*, 4, 135-145.

AMAZON.COM (2010) Customer Reviews - Talking Photo Album. Retrieved 21/05/2010,fromhttp://www.amazon.com/Vat19-Talking-Photo-Album/productreviews/B0010DYDFY/ref=dp_top_cm_cr_acr_txt?ie=UTF8&showViewpoin ts=1.

AUGMENTATIVE COMMUNICATION INC. (2010) Talking Photo Retrieved21/05/2010, http://www.augcominc.com/index.cfm/talking_photo_album.htm. Album. from

BLACKSTONE, S. & BERG, M. H. (2003) Social networks: A communication inventory for individuals with complex communication needs and their communication partners, Monterey, California, USA, Augmentative Communication Inc.

BOEHM, B. W. (1988) A spiral model of software development and enhancement. *IEEE Computer*, 21, 61-72.

COCKBURN, A. (2000) *Writing Effective Use Cases,* Addison-Wesley Professional.

DAHLGREN SANDBERG, A., SMITH, M. & LARSSON, M. (2010) An Analysis of Reading and Spelling Abilities of Children Using AAC: Understanding a Continuum of Competence. *Augmentative and Alternative Communication*, 26, 191-202.

DEPARTMENT OF COMMUNITIES - DISABILITY SERVICES (2009) Complex communication needs, Brisbane, The State of Queensland, Australia.

HARTSON, H. R. & HIX, D. (1989) Toward empirically derived methodologies and tools for human-computer interface development. *International Journal of Man-Machine Studies*, 31, 477-494.

HELMS, J. W., ARTHUR, J. D., HIX, D. & HARTSON, H. R. (2006) A field study of the Wheel - a usability engineering process model. *Journal of Systems and Software,* 79, 841-858.

HEWETT, T. T. & MEADOW, C. T. (1986) On designing for usability: an application of four key principles. *SIGCHI Bull.*, 17, 247-252.

HOLZINGER, A. (2005) Usability engineering methods for software developers. *Communications of the ACM - Interaction design and children,* 48, 71-74.

HOVE, S. E. & ANDA, B. (2005) Experiences from conducting semi-structured interviews in empirical software engineering research. *Software Metrics*, 2005. 11th IEEE International Symposium.

IACONO, T. (2004) Patients with disabilities and complex communication needs. *Australian Family Physician*, 33, 585-589.

INCLUSIVE TECHNOLOGY LTD (2010) A3 Talking Photo Album. Retrieved 21/05/2010,from http://www.inclusive.co.uk/catalogue/acatalog/a3 talking photo album.html

- INTERNATIONALORGANISATIONFORSTANDARDIZATION(2010) Ergonomics of human-system interaction -- Part 210: Human-centred design for interactive systems. *ISO* 9241.
- JOHNSON, J. M., INGLEBRET, E., JONES, C. & RAY, J. (2006) Perspectives of speech language pathologists regarding success versus abandonment of AAC. *Augmentative and Alternative Communication*, 22, 85-99.

KING'S FUND (2001) Definition of the term 'Assistive Technology'. User group consultation meeting.

KREITZBERG, C. B. (2008) The LUCID Framework. Retrieved 28/01/2011, from http://www.leadersintheknow.biz/Portals/0/Publications/Lucid-Paper-v2.pdf.

LAUESEN, S. & YOUNESSI, H. (1998) Six styles for usability requirements. IN DUBOIS, E., OPDAHL, A. L. & POHL, K. (Eds.) *Fourth International Workshop on Requirements Engineering: Foundations of Software Quality REFSQ'98.* Presses Universitaires de Namur.

MARTIN, A. (2004) Latein Wörterbuch. Retrieved 12/10/2010, from http://albertmartin.de/latein/?q=communicare&con=0.

MUSSELWHITE, C. R. & ST. LOUIS, K. W. (1988) Communication programming for persons with severe handicaps: Vocal and Augmentative Strategies, Austin, Tx, Pro-Ed.

NEEDAPRESENT.COM (2010) Talking Photo Album. Retrieved 21/05/2010, from http://www.needapresent.com/gifts/talking-photo-album.

NEWELL, A. F. & GREGOR, P. (2000) "User Sensitive Inclusive Design" - in search of a new paradigm. *Proceedings on the 2000 conference on Universal Usability.* Arlington, Virginia, United States, ACM.

NIELSEN, J. (1997) The use and misuse of focus groups. *Software, IEEE,* 14, 94-95.

NIELSEN, J. (2010a) Characteristics of Usability Problems Found by Heuristic Evaluation.Retrieved03/06/2010,from http://www.useit.com/papers/heuristic/usability_problems.html.

NIELSEN, J. (2010b) How to Conduct a Heuristic Evaluation. Retrieved 15/06/2010,from http://www.useit.com/papers/heuristic/heuristic_evaluation.html.

NIELSEN, J. (2010c) Ten Usability Heuristics. Retrieved 15/06/2010, from http://www.useit.com/papers/heuristic/heuristic_list.html.

NIELSEN, J. & MOLICH, R. (1990) Heuristic evaluation of user interfaces. Proceedings of the SIGCHI conference on Human factors in computing systems: Empowering people. Seattle, Washington, United States, ACM.

OLSSON, E. (2004) What active users and designers contribute in the design process. *Interacting with Computers,* 16, 377-401.

PHILLIPS, B. & ZHAO, H. (1993) Predictors of Abandonment. Assistive Technology, 5, 36-45. Assistive Technology

PHOTOS-SPEAK COMPANY (2010) Digital Talking Photo Album / Photo Organizer. Retrieved 21/05/2010, from http://photosspeak.com/index.html.

POLKINGHORNE, D. E. (1995) Narrative configuration in qualitative analysis. IN J. A. HATCH AND R. WISNIEWSKI (Ed.) *Life history and narrative.* London, Routledge.

- PRIOR, S. (2010a) HCI methods for including adults with disabilities in the design of CHAMPION. 28th International Conference on Human Factors in Computing Systems. Atlanta, Georgia, USA, ACM.
- PRIOR, S. (2010b) Working with adults with complex communication needs in human centred design. Unpublished work.

QUALITY ENABLING DEVICES LTD (2010) Talking Photo Album. Retrieved 21/05/2010,from http://www.qedonline.co.uk/catalog/product_info.php?products_id=577.

RIEMER-REISS, M. L. & WACKER, R. R. (2000) Factors Associated with Assistive Technology Discontinuance Among Individuals with Disabilities. *Journal of Rehabilitation,* 66, no page numbers.

ROYCE, W. W. (1970) Managing the development of large software systems. *IEEE Wescon.*

SANDERS, E. B.-N. (2002) From User-Centred to Participatory Design Approaches. IN FRASCARA, J. (Ed.) *Design and the Social Sciences*. London, Taylor & Francis.

SANDERS, E. B. N. & STAPPERS, P. J. (2008) Co-creation and the new landscapes of design. *CoDesign: International Journal of CoCreation in Design and the Arts,* 4, 5 - 18.

SCHERER, M. J. (2002) The Study of Assistive Technology Outcomes in the United States. *Computers Helping People with Special Needs*. Berlin / Heidelberg, Springer.

SEELMAN, K. D., PORRERO, P. & BALLABIO, E. (1998) Disability's New Paradigm: Implications for Assistive Technology and Universal Design. *European Union TIDE Conference*. Helsinki, Finland.

SPILLNER, A., LINZ, T. & SCHAEFER, H. (2007) *Software Testing Foundations,* Santa Barbara, CA, Rocky Nook Inc.

TALKING PRODUCTS LTD (2010) Talking Photo Albums. Retrieved 21/05/2010, from http://www.talkingproducts.co.uk/talking_photo_albums.htm.

TERS ELECTRONIC LTD (2010) Talking Photo Album GR524. Retrieved 21/05/2010, from http://www.talkingproductsfactory.com/Talking-Photos-china/Talking-Photos-Talking-Photo-Album-GR524-1/.

THE UNITED NATIONS (1997) International Covenant on Civil and Political Rights.Retrieved19/08/2010,from http://www.hrweb.org/legal/cpr.html#Article 19.2.

- U. S. HOUSE OF REPRESENTATIVES (2010) 29 USC Sec. 3002(4) Assistive technologydevice.Retrieved04/01/2010,from http://uscode.house.gov/uscodecgi/fastweb.exe?getdoc+uscview+t29t32+936+83++%28assistive%.
- WALLER, A. (1992) Providing narratives in an augmentative communication system. PhD thesis. *School of Computing.* Dundee, UK, University of Dundee.
- WALLER, A. (2009) Interpersonal Communication. IN STEPHANIDIS, C. (Ed.) *The Universal Access Handbook.* CRC Press Taylor and Francis.
- WALLER, A., BALANDIN, S. A., O"MARA, D. A. & JUDSON, A. D. (2005) Training AAC Users in User-Centred Design. *In Accessible Design in the Digital World Conference*. Dundee, Scotland.

WIEGERS, K. E. (2003) *Software Requirements,* Redmond, Washington, USA, Microsoft Press.

WILHELM, T. (2010) Focus groups. Retrieved 03/02/2011, from http://www.eresult.de/leistungen/methoden_verfahren/fokusgruppen_eng.ht ml.

WILLIAMS, M. B., KREZMAN, C. & MCNAUGHTON, D. (2008) "Reach for the Stars": Five Principles for the Next 25 Years of AAC. *Augmentative and Alternative Communication*, 24, 194-206.

WORLD HEALTH ORGANIZATION (2004) A GLOSSARY OF TERMS FOR COMMUNITY HEALTH CARE AND SERVICES FOR OLDER PERSONS. Retrieved04/01/2011,from http://whqlibdoc.who.int/wkc/2004/WHO_WKC_Tech.Ser._04.2.pdf.

9 List of Appendices

Appendix 1:	Project plan with risk analysis and mitigation possibilities.
Appendix 2:	A3 poster outlining the project idea in symbols.
Appendix 3:	Project summary.
Appendix 4:	Evaluation questionnaire for prototype #1 and #2 and the usability button game.
Appendix 5:	Nielsen ^w s 10 Usability Heuristics.

Participation of People with Complex Communication Needs in Designing Assistive Technology Software for Information Sharing



Appendix G. Search Strings Used

User Centred Design and Communication Impairment*

User Centred Design AND Communication Impairment*

User Centred Design AND Severe Speech and Physical Impairments

User Centred Design AND Severe Speech and Physical Impairment*

User Centred Design AND Speech Impairment*

User Centred Design AND Speech Disability*

User Centred Design AND Communication Disability*

User Centred Design AND Cerebral Palsy

Participatory Design and Communication Impairment*

Participatory Design AND Communication Impairment*

Participatory Design AND Severe Speech and Physical Impairments

Participatory Design AND Severe Speech and Physical Impairment*

Participatory Design AND Speech Impairment*

Participatory Design AND Speech Disability*

Participatory Design AND Communication Disability*

Participatory Design AND Cerebral Palsy

Co Design and Communication Impairment*

Co Design AND Communication Impairment*

Co Design AND Severe Speech and Physical Impairments

Co Design AND Severe Speech and Physical Impairment*

Co Design AND Speech Impairment*

Co Design AND Speech Disability*

Co Design AND Communication Disability*

Co Design AND Cerebral Palsy

Appendix H. Description of Studies in Literature Review

Paper	Number of	Software Being
	Participants	Developed
(Waller et al., 2009)	9 Children with	Joke generating software
	Cerebral Palsy	
(Dunlop et al., 2002)	1 Participant with	Digital Library
	Cerebral Palsy	
(O'Connor et al., 2006)	1 Participant with	Video Software
	SSPI	
(Steriadis and Constantinou, 2003)	3 quadriplegic adults	Software to control motor
		movements
(McGrenere et al., 2003)	1 Participant with	Diary Software
	Aphasia	~
(Tee et al., 2005)	9 Adults with Aphasia	Cookery Software
(Allow at al. 2007)	5 Dorticipanta with	Image Communication
(Allell et al., 2007)	5 Participants with Aphasia	Application
(Wisser et al. 2008)	Not Available	Crophic Symbols
(visser et al., 2008)	INOL AVAIIADIE	Graphic Symbols
(Light et al., 2007)	6 Children with SSPI	AAC Device
(Boyd-Graber et al., 2006)	9 Participants with	Desktop PDA
	Aphasia	
(Allen, 2005)	Not Available	Portland Communication
		Aid
(McCoy et al., 1997)	Not Available	Intelligent AAC Device
(Hornof, 2008)	2 Children with	Communication Software
	Cerebral Palsy	
(Davies et al., 2004)	1 Teenager with	PDA Communication
	Aphasia	Device
(Hengeveld et al., 2008b)	12 Children with	Educational Software
	Cerebral Palsy	

Appendix I. Requirements

Session	Name of Requirement	Functional	Design	Suggested	Suggested
Requirement	-	Requirement	Requirement	by	By
made		-	-	Participant	Researcher
				-	
Paper	Put in voice	*		*	
prototyping	synthesis				
1 51 0	5				
Paper	Use eye gaze	*		*	
prototyping					
prototyping					
Paper	Allow videos, text	*		*	
prototyping	and photographs to				
prototyping	all be stored for the				
	same information				
	object				
Denen	Change in 1		*	*	
Paper	Change wording on		*	*	
prototyping	questions				
D	TT 1 1			ala	
Paper	Have symbols on		*	*	
prototyping	buttons				
		-t-			
Clicker5	Alter voice	*		*	
prototyping	synthesis				
Clicker5	Get rid of back	*			*
prototyping	button in				
	questionnaire				
Clicker5	Just press yes/no to	*		*	
prototyping	move through				
	questions				
	1				
Clicker5	Use AAC device as	*		*	
prototyping	keyboard				
F8					
Clicker5	Change wording on		*	*	
prototyping	questions				
r	1				
Clicker5	Stop it interfering		*		*
prototyning	with other buttons				
recorpting	on screen				
Clicker5	Make buttons large		*		*

prototyping	and centred				
Clicker5	Have symbols on		*	*	
prototyping	buttons				
Visual	Swap between	*		*	
Studio	symbols				
prototyping					
Visual	Change scanning	*		*	
Studio prototyping	speed				
prototyping					
Visual	Use less "cartoon"		*	*	
Studio prototyping	like drawings				
prototyping					
Visual	Allow changes on	*		*	
Studio prototyping	starr who can see				
prototyping	linormation				
Visual	Make all buttons		*		*
Studio prototyping	the same size and				
prototyping	consistency in the				
	software				
Visual	Change voice	*		*	
Studio	synthesiser used				
prototyping	-				
Visual	Change wording on		*	*	
Studio	some buttons				
prototyping					
Visual	Allow user to	*		*	
Studio	choose whether				
prototyping	instructions should				
	be read out				

Appendix J. Evaluations

Testing Champion

Connect to Computing Network



Connect to a net	work		Double VPN Co	Click nnection	on 1
Disconnect or con	nect to another network				
Show All Dial-up and VPN	▼	+ ₇			
VPN Connec	tion VPN connection	Name: VPN Connection			
Wireless Network Cor	nection	Name. VPN Connection			

ting t	Connection		DO NOT change anything in this box. Click Connect
User name: Password:	sprior [To change the saved password, click here]		
Domain:	COMPUTING	Ιſ	Double Click on
Save this Save this Save this Connect	s user name and password for the following users: nly ne who uses this computer Cancel Properties Help	Ŀ	VPN Connection

Record the Screen





When this option appears, ensure that the "I would like the evaluate Camtasia Studio" option is selected and press Finish.





This black box will now appear, Press the red record button to begin recording the screen



A countdown box will now appear. Wait until this has disappeared and then continue. It is important **not** to press F3 during recording as this will stop the recording.





This window will appear when the program running Champion is loaded. Please press **F5** to start Champion



Champion will now start and you can use it to carry out the tasks you are wanting to do.

Ending the session



Once you have finished using Champion and have exited the program. Press F3 to stop recording and save your recording.



A preview box will now appear with a recording showing what you did during your session. You do not have to watch all of this.

	Click on Save to store this recording
Save Produce Save this recording file and close	
Studi Preview Studiew Window.	
Vo Preview	
Cantasia Accorden	Please save the recording with

- cannasia necoraci		
🕖 📕 « Do	cuments 🕨 Camtasia Studio 🕨 📼 😽 🖌 Search	
File name:	capture-1	
Save as type:	Camtasia Recording Files (*.camrec)	

Please save the recording with todays date (in the form 12January etc) and the users name Thank you for evaluating CHAMPION if there is anything you would like to tell me about using the program, please have it written down in the Champion notebook.

Any problems with the software please contact Suzanne on sprior@computing.dundee.ac.uk

Appendix K. Straight Talking User Centre

The vision for the first computing user centre for adults with Severe Speech and Physical Impairment (SSPI) was to allow adults with SSPI to come into a computing department and become co-designers in the development of assistive technology while also learning new IT skills. It was hoped that this would empower the adults with SSPI and provide them with an opportunity to have the experience of working and potentially give them to confidence to look for paid employment. As with the existing user centre for older adults, the centre members would have the opportunity to help develop better technology for the whole of the population with SSPI. It was hoped that having adults with SSPI in the department would also help 'normalise' their presence and help students to feel comfortable when meeting people with disabilities.

To investigate the feasibility for a user centre, a pilot study with 4 participants (3 female and 1 male) was conducted. All of the members had previously been involved in research projects within the department and were known to the research staff.

All participants had Severe Speech and Physical Impairments. Three participants used a wheelchair and the other used a walking aid. All of the participants had profound communication impairments and relied on gestures and aids to communicate. Two participants used LightWriter AAC devices, one used a paper board with words written on it and the other used gestures and eye gaze.

Three participants were accompanied by a support worker to the pilot study meetings. One participant lived in residential supported accommodation while the others lived in the community.

Members were invited to an initial meeting to discuss the aim of the group and to decide upon a name for the group. This discussion was in the form of an adapted focus group (Prior, Waller, & Kroll, To Appear). The members chose the name "Straight Talking" and one participant offered to produce a logo.

During the initial discussion, the group generated 3 aims of the group:

- -To train members in becoming "expert end users"
- -To raise awareness in wider community of AAC
- -To provide a social space for members to meet

The initial sessions included tasks such as designing the group's website and a logo for the group on the computer.

It is anticipated that centre members will be used as evaluators in research and student projects. Before this service could be made available to students and staff the centre members needed preparation in being evaluators.

Centre members were given a lesson on conducting heuristic evaluations. It was not the intention of the session to produce heuristic evaluators but to provide them with a better understanding of how evaluations were conducted and provided them with a structure for their own evaluations.

Forum theatre techniques (Prior et al., 2008) were used to help prepare members for some of the challenges that come with being participants in student projects. Centre staff took on the role of the actor, playing a variety of different characters. Some portrayed students as being nervous about working with people with SSPI while others played students who were domineering and did not take the time needed to allow centre members to speak their views. One centre member would act the part of the evaluator while others were watching and providing suggestions on how they could deal with the student. By using forum theatre, participants were able to discuss difficult topics in a non-threatening environment. Because the 'students' they were discussing were seen as fictional, the participants did not feel awkward discussing them and their views about them.

At the end of the session the suggestions that had been made for dealing with the students were written on a board and discussed further.

The members were trained in IT skills and were assisted in undertaking their European Computer Driving License (ECDL) if they wished. These initial members were also provided with training on assisting others to acquire IT skills and were encouraged to give presentations on IT tasks.

The centre has had one commission from outside the department. A visiting researcher from another university wanted access to expert disabled users in order to trial design techniques with adults with SSPI. The authors met with the researcher for 3 hours prior to the session and discussed the aims for the session. The author also ensured that the visiting researcher had realistic expectations for what they could achieve in the session.

Members of the centre worked with the researcher in a specially facilitated session, the author was present throughout the session to supervise and guide the session progress.

The first author, a software developer with experience in SSPI, acted as the centre manager and coordinated sessions and computer activities. The second and third author also assisted with sessions and administration tasks to ensure participants were able to attend the sessions.

It is anticipated that when the centre expands a support worker will also be hired to provide personal care and communication support when the centre is expanded to more members.

The next step for the centre will involve developing space within the Queen Mother Building at the School of Computing for use by adults with SSPI. Existing technology labs will need to be adapted for use by a larger group of adults with mobility impairments.

The group currently meets in the older adults user centre but this is not the ideal space for wheelchair users due to the amount of space, the height of desks etc.

Members for the centre will be recruited through local adult disability centres and the council social work department. The number of members admitted to the centre at any one time will unfortunately be restricted to ensure that personal care needs can be met and there are enough assistants available.

There will be a variety of types of sessions at the centre. Drop in sessions will allow members to come in and use the computers as they wish with help if they encounter difficulties. Education sessions will include lessons given to the group on topics of interest before the opportunity to do practical tasks on the topic with support. Topics for the education sessions will be suggested by centre members, for example, during a previous session at a day resource centre participants asked for assistance in using Facebook, however due to lack of IT facilities the time on Facebook had been very limited. The third type of session will be research sessions.

Once the centre is established and has a regular group of members who are happy in attending and comfortable in working with staff, students will be invited to meet with centre users and to work with them on course assignments during scheduled research sessions. When the user centre for older adults at the University of Dundee (Forbes, 2009) was established the number of student projects looking at older adults dramatically increased. It is anticipated that a similar effect will be seen with the user centre for adults with SSPI. Student

projects might focus on developing a piece of assistive technology or in developing a piece of technology for the general population but ensuring that it is also useable by people with SSPI.

In order to facilitate this work with students and to ensure that the users with SSPI do not become over used a system will be set up whereby students must liaise with the centre and specify the number of participants they require, the number of sessions needed and the duration of these.

Research projects by staff involving adults with SSPI could lead to the development of assistive technology which has a much lower abandonment rate than current AT and could improve the quality of life and independence by people with SSPI.

The pilot study has shown that the concept of a user centre is welcomed by members, as yet no one has turned down the offer of a place. The users are keen to engage in a variety of activities and they wish to bring their own skills to the group (e.g. through creating logos or assisting with the forum theatre).

There are still many challenges to be faced in the group including supporting a larger number of members, dealing with a wider range of abilities and ensuring continued funding Appendix L. Guidelines

Guidelines Version 1

Introduction

Human-centred design is an approach of interactive system development that focuses specifically on making systems usable. It is a multi-disciplinary activity which incorporates human factors and ergonomics knowledge and techniques. The application of human-factors and ergonomics to interactive systems design enhances effectiveness and efficiency, improves human working conditions, and counteracts possible adverse effects of use on human health, safety and performance. Applying ergonomics to the design of systems involves taking account of human capabilities, skills and limitations and needs.

A person with Severe Speech and Physical Impairments (SSPI) will have different problems than a person with a Communication impairment which describes a disability affecting speech and language. A person with a communication impairment may have difficulties including articulation problems, fluency problems, aphasia and delays in speech (Medline Plus, 2009). SSPI encompass more than a communication disorder and include a range of physical, sensory and cognitive impairments (Balandin, 2002).

Working with adults with Severe Speech and Physical Impairments in human centred design

Scope

This standard provides guidance on conducting human-centred design with users with Severe Speech and Physical Impairments throughout the lifecycle of the development of a technology. This technology is not limited to assistive technology and could be a piece of technology for mainstream consumption. This standard is concerned with both hardware and software components of interactive systems.

This standard addresses the planning and management of human-centred design with users with SSPI. It is based upon the guidance in ISO-13407 regarding conducting human-centred design and as such does not address the specifics of the stage of the lifecycle specified in ISO-13407 but instead provides guidance on how these stages can be conducted with users with SSPI.

The main users of this standard will be those developers or researchers who will be working directly with users with SSPI. However it is beneficial for the entire development team to be aware of the issues in working with these users and how the project is addressing the issues.

Terms and definitions

For the purposes of this standard, the following terms and definitions apply.

Prototype

Representation of all or part of a product or system that, although limited in some way, can be used for evaluation

[ISO 13407:1999, definition 2.2]

Usability

Extent to which a product can be used by specified users to achieve specified goals with effectiveness, efficiency and satisfaction in a specified context of use.

[ISO 13407:1999, definition 2.3]

Effectiveness

Accuracy and completeness with which users achieve specified goals.

[ISO 9241-11:1998, definition 3.2]

Efficiency

Resources expended in relation to the accuracy and completeness with which users achieve goals.

```
[ISO 9241-11:1998, definition 3.3]
```

Satisfaction

Freedom from discomfort, and positive attitudes to the use of the project

[ISO 9241-11:1998, definition 3.4]

Context of use

Users, tasks, equipment (hardware, software and materials), and the physical and social environments in which a product is used.

[ISO 9241-11:1998, definition 3.5]

User

Individual interacting with the system.

[ISO 9241-10:1996, definition 2.2]

SSPI

Severe Speech and Physical Impairments encompass more than a communication disorder and include a range of physical, sensory and cognitive impairments (Balandin, 2002).

AAC

The general term for the methods used to aid communication by those for whom the more usual forms of communication are not sufficient. AAC can be divided into two different categories of unaided and aided. Most people who use AAC will use a combination of the two (American Speech-Language-Hearing Association, 2002).

Support Worker

This refers to any person who assists the user in their day to day lives or in using the system. This may be a person employed to assist the user or a family member.

Developer

This refers to the HCI practitioner, the software developer or the researcher who is involved in the project.

Rationale for working with users with SSPI

The number of computer users with cognitive and developmental disabilities is increasing. Examples of this include: day and residential units holding computer courses as part of the program for service users (Parsons et al., 2006) and schools for children with disabilities now have ICT lessons in their basic curriculum (Judge, 2001). In addition, as adults with developmental disabilities become more integrated into the workplace, the demand for computer software which is accessible by a wider spectrum of users is likely to increase.

However, the main use of technology by this user group remains in assistive technology and AAC devices. As with other user groups, a lack of user centred design may contribute to the poor adoption of technology (e.g., the rate of abandonment of AAC devices is reported to be as high as 53.3% (Riemer-Reiss and Wacker, 2000)). The main reasons given for abandonment or rejection of assistive technology are a lack of training for the end user and a lack of flexibility in adapting the system for the individual. Other reasons reported include a lack of confidence on the part of the user in their ability to use the system and the prohibitive costs in learning how to use the system.

It is suggested that the inclusion of end users in the design process may reduce this abandonment rate (Waller et al., 2005b).

Structure of this standard

Clause \Box outlines the problems that can occur with Severe Speech and Physical Impairments.

Clause 0 gives an overview of the technology that are used to assist people with SSPI.

Clause 0 discusses the problems that are noted when doing HCD with users with SSPI.

Clause 0 gives guidance on how to deal with these problems.

Clause \Box highlights methods known to work at each stage of ISO 13407.

Problems for people with SSPI

People with SSPI may have cognitive impairment which can range from mild to severe and may have difficulty in understanding all that is being said in the conversation and the developer's questions.

People who use an AAC device for communication can produce words up to 25 times slower than those with normal verbal speech (Higginbotham et al., 2008b). The time therefore for a participant to reply, for example "Yes, that has happened to me" could take up to 2 minutes for a participant relying on AAC (Augmentative and Alternative Communication at the University of Washington, 2009). A person with dsyarthric speech may be able to produce words at the same rate as a participant without a SSPI, however the need for clarification of the response can slow down the rate at which the response is understood by the group (Hustad, 2006a).

People with SSPI are likely to become tired more quickly than the general population (Jahnsen et al., 2003a). This is attributed to "post-impairment syndrome", which means that those with SSPI will use between 3 and 5 times as much energy as those without the condition (Wood et al., 2008). Fatigue is also one of the most common and debilitating symptoms associated with Traumatic Brain Injury.

Technology to assist people with SSPI

Augmentative and Alternative Communication (AAC) is the general term for the methods used to aid communication by those for whom the usual form of communication through speech is not sufficient. AAC can be divided into the two different categories of unaided and aided. Most people who use AAC will use a combination of both (American Speech-Language-Hearing Association, 2002).

In unaided AAC systems additional equipment is not used, for example sign language, Makaton (Communication Matters, 2008) and facial expressions (Communication Matters, 2003). These systems have the benefit that they can be used anywhere and the user does not need to carry additional equipment around. However unaided AAC has the disadvantage that the communication partner needs to understand how the system works.

Aided AAC systems refer to a communication methods that involve a device – this may be electronic or paper based (e.g. a word board) – of some form which is external to the user. The device can be used to transmit or to receive messages and generally display symbols which the user selects to convey messages to listeners, for example a symbol board (Beukelman and Mirenda, 1992). A user can select a symbol, picture or in the case of literate users a selection of letters to build up a message. The main advantage of an aided device is its ability to offer potential for communication to people with SSPI for whom unaided systems require refined motor skills or cognitive load not within the user's capability (Hampson, 2006). Electronic devices also offer the potential for synthesised speech output in the case of speech generating devices.

Other assistive technology for this group may include electric scooters or wheelchairs to assist with mobility, personal organisers or reminders to assist with cognitive difficulties, screen readers, magnifiers and hearing aids to help with sensory problems and environmental controls to help them to live as independently as possible.

Problems that can occur when doing HCD with users with SSPI

Can a group of 'representative users' be gathered?

gathering a representative sample of different disabilities and needs is more challenging

How can different accessibility needs be met?

Can informed consent be gathered from participants, and if not is it possible to gain assent from the participant and informed consent from a legal guardian.

It is important to ensure that participants are aware of what they are consenting to

Some participants may have a guardian appointed who is required legally to consent to research.

How will you encourage participants to give feedback?

Participants are often uncomfortable in giving negative feedback to those who have designed the software

How will feedback be recorded?

Usual methods such as notes and audio recordings may not be suitable

Audio recordings may miss information if communication is not through a voice output machine

It is difficult to take notes if participants are communicating via eyegaze and word board

Is it necessary to provide incentives to participants?

Payments can interfere with benefit rules

What measures will be put in place after the study to allow participants to continue with new skills if they wish to?

Participants may have developed new skills during their involvement with the research and may be keen to continue with this after the project ends.

While organising future support is not the responsibility of researcher or developer, there is an ethical obligation to ensure that participants do not feel abandoned at the end of the study.

How will you report back to participants on what you used their feedback for?

The traditional methods for reporting back to participants is by writing a letter to participants thanking them for their help and explaining what stage the project is now at. This may not be suitable for users with SSPI.

How will contact be made with participants?

Recruiting end users can be a challenge in any situation, unlike traditional development advertising in local press or through local universities can be inappropriate or unlikely to reach the desired participants.

Guidance on issues for HCD and users with SSPI

Can a group of 'representative users' be gathered?

It is not difficult to gather a spectrum of different ages and a mix of genders among participants but gathering a representative sample of different disabilities and needs is more challenging.

It may be possible to use purposive or judgemental sampling to find users who cover most of main disabilities (communication impairment, mobility impairment, cognitive impairment and sensory impairment).

Developers can encourage users to consider how others whom they know with different disabilities would interact with the system.

Participants can be asked to consider talking to others about the system or to even ask them if they wish to take part in later evaluations.

By allowing different groups of participants to do later evaluations it is possible to gain as wide a range as possible.

How can different accessibility needs be met?

Conflicting requirements are already part of the user centred design process, in this situation however discussions between users and designers is unlikely to help resolve the issue.

The designers need to carefully examine the possibilities for including as many accessibility and adaptations as possible.

It may be possible to switch accessibility options on and off

Careful planning during development can often allow different adaptations to be used through the same piece of code.

Can informed consent be gathered from participants, and if not is it possible to gain assent from the participant and informed consent from a legal guardian?

By talking to participants after going through Participant Information Sheet and asking yes/no questions we can check that they understand what they are agreeing to.

It is still important to gain assent from the participants in the instance of a guardian signing. This should include questions to ensure they want to take part and know what will be happening in the process.

How will you encourage participants to give feedback?

Participants need to be encouraged to give negative feedback.

It is helpful to remind participants that you really want them to find flaws in the technology.

How will feedback be recorded?

Video can be intrusive, but careful planning can minimise this

Is it necessary to provide incentives to participants?

Methods of incentives can be through gift tokens etc, which should not interfere with benefit rules.

Participants will often volunteer to work without payment due to the enjoyment of taking part.

What measures will be put in place after the study to allow participants to continue with new skills if they wish to?

If it is possible for participants to take part in other projects then putting them in contact with the person organising these should be discussed

Researchers should discuss with the participants and their support staff what parts of the project they enjoyed most

How will you report back to participants on what you used their feedback for?

With participants with SSPI it is beneficial for them to meet with the development team and be given personal feedback on how the project has developed and how their own contributions have been used.

How will contact be made with participants?

Charities who work with people with SSPI may be willing to make the initial contact on behalf of the development team and then if participants are in agreement to put the two parties in contact.

There may be additional ethics procedures to go through with a charity and time should be allowed for this.

Example methods for stages in HCD

General Points

At the beginning of each session review the previous session, this will help any participants who were not present to be caught up on what has happened and will help participants with cognitive problems to recall what took place.

It is important to ensure that participants don't become overtired during the sessions, and to allow for extra time to facilitate breaks. Designers should remember that participants with SSPI are more likely to have illnesses than the general population and there may be times when sessions have to be cancelled or altered at short notice due to illness.

While it is important not to talk in an overly complicated or technical way with participants it is also vital that participants are not spoken to in a condescending tone and are made to feel as equals with the designer.

Understand and specify the context of use

Make contact with potential users and/or organisations who work with people with SSPI. It is important that participants become comfortable with the researcher/developer prior to the work commencing. The staff or family who spend time with participants should also meet with the researcher/developer.

Examine the ethical situation carefully, in addition to any internal ethics procedures, there may be ethics applications required by the charities or organisations involved. The local health authority may also require an ethics application depending where the sessions with the participants will be conducted and how participants will be recruited.

Spend time in the environment and understand how the technology may be used.

Requirements Gathering

Requirements for how technology could be used

Adults with SSPI may at first be hesitant in giving requirements to researchers due to previous poor experience of having people in authority/people from outside of their immediate circle listening to them. As such a group discussion for requirements may be beneficial – particularly if it can be arranged for the participants to know one another before hand. During group discussions it is important for people to have time to form answers using their communication device and not to feel rushed into answering. Participants will benefit from being given feedback on what was discussed at the end of the session in addition to the traditional summary which is sent out a later date. By feeding back immediately on what the researchers have heard, participants will know that their opinions have been listened to and taken onboard.

Requirements on how technology should operate
Participants may have trouble imagining how technology will work or trouble keeping track of what part of the process in using the technology they are currently imagining. Providing storyboards or videos on problems that currently exist and/or how the technology could possibly be used may help alleviate this. It is important not to specify exactly what the system will do, but allows participants to visualise and imagine more easily. The research/designer can show on the screen an image of the position of the process that is currently being discussed. It may be helpful to prompt for consideration of other groups.

Produce Design Solutions

Lo-fidelity prototyping can involve producing sketches of how technology will look and/or in user centred design having the participants produce sketches for how they imagine the system looking. This is not usually a suitable method of prototyping with adults with SSPI. One technique found to be successful is to create more tangible prototypes. Examples of this include making models of the device onto which components can be attached on as the participants wish, or by creating screen designs and then cutting up the buttons etc. These components can then be attached to magnets, the participants can then position the components on a magnetic board in the position they wish. It may also be beneficial to allow participants to use existing accessibility tools and simulate how they work in this situation. It is important that when showing prototypes designs are kept flexible enough to allow instant alterations.

Evaluation

Designers should consider different evaluation tools and not simply accept that evaluations can only take place through observations. Consider different evaluation tools, it may be possible for participants to be trained to perform heuristic evaluations. If support staff can be included in the evaluations and are willing to encourage participants to take part then there can be a role for them, many support workers will be used to completing diaries or logs as part of their daily routine and so an evaluation diary may work better in this situations than it does in a business environment for example.

Guidelines Version 2

Introduction

Human-centred design is an approach of interactive system development that focuses specifically on making systems usable. It is a multi-disciplinary activity which incorporates human factors and ergonomics knowledge and techniques. The application of human-factors and ergonomics to interactive systems design enhances effectiveness and efficiency, improves human working conditions, and counteracts possible adverse effects of use on human health, safety and performance. Applying ergonomics to the design of systems involves taking account of human capabilities, skills and limitations and needs.

A person with Severe Speech and Physical Impairments (SSPI) will have different problems than a person with a Communication impairment which describes a disability affecting speech and language. A person with a communication impairment may have difficulties including articulation problems, fluency problems, aphasia and delays in speech (Medline Plus, 2009). SSPI encompass more than a communication disorder and include a range of physical, sensory and cognitive impairments (Balandin, 2002).

Working with adults with Severe Speech and Physical Impairments in human centred design

Scope

This standard provides guidance on conducting human-centred design with users with Severe Speech and Physical Impairments throughout the lifecycle of the development of a technology. This technology is not limited to assistive technology and could be a piece of technology for mainstream consumption. This standard is concerned with both hardware and software components of interactive systems.

This standard addresses the planning and management of human-centred design with users with SSPI. It is based upon the guidance in ISO-13407 regarding conducting human-centred design and as such does not address the specifics of the stage of the lifecycle specified in ISO-13407 but instead provides guidance on how these stages can be conducted with users with SSPI.

The main users of this standard will be those developers or researchers who will be working directly with users with SSPI. However it is beneficial for the entire development team to be aware of the issues in working with these users and how the project is addressing the issues.

Terms and definitions

For the purposes of this standard, the following terms and definitions apply.

Prototype

Representation of all or part of a product or system that, although limited in some way, can be used for evaluation

[ISO 13407:1999, definition 2.2]

Usability

Extent to which a product can be used by specified users to achieve specified goals with effectiveness, efficiency and satisfaction in a specified context of use.

[ISO 13407:1999, definition 2.3]

Effectiveness

Accuracy and completeness with which users achieve specified goals.

[ISO 9241-11:1998, definition 3.2]

Efficiency

Resources expended in relation to the accuracy and completeness with which users achieve goals.

[ISO 9241-11:1998, definition 3.3]

Satisfaction

Freedom from discomfort, and positive attitudes to the use of the project

```
[ISO 9241-11:1998, definition 3.4]
```

Context of use

Users, tasks, equipment (hardware, software and materials), and the physical and social environments in which a product is used.

[ISO 9241-11:1998, definition 3.5]

User

Individual interacting with the system.

[ISO 9241-10:1996, definition 2.2]

SSPI

Severe Speech and Physical Impairments encompass more than a communication disorder and include a range of physical, sensory and cognitive impairments (Balandin, 2002).

AAC

The general term for the methods used to aid communication by those for whom the more usual forms of communication are not sufficient. AAC can be divided into two different categories of unaided and aided. Most people who use AAC will use a combination of the two (American Speech-Language-Hearing Association, 2002).

Support Worker

This refers to any person who assists the user in their day to day lives or in using the system. This may be a person employed to assist the user or a family member.

Developer

This refers to the HCI practitioner, the software developer or the researcher who is involved in the project.

Rationale for working with users with SSPI

The number of computer users with cognitive and developmental disabilities is increasing. Examples of this include: day and residential units holding computer courses as part of the program for service users (Parsons et al., 2006) and schools for children with disabilities now have ICT lessons in their basic curriculum (Judge, 2001). In addition, as adults with developmental disabilities become more integrated into the workplace, the demand for computer software which is accessible by a wider spectrum of users is likely to increase.

However, the main use of technology by this user group remains in assistive technology and AAC devices. As with other user groups, a lack of user centred design may contribute to the poor adoption of technology (e.g., the rate of abandonment of AAC devices is reported to be as high as 53.3% (Riemer-Reiss and Wacker, 2000)). The main reasons given for abandonment or rejection of assistive technology are a lack of training for the end user and a lack of flexibility in adapting the system for the individual. Other reasons reported include a lack of confidence on the part of the user in their ability to use the system and the prohibitive costs in learning how to use the system.

It is suggested that the inclusion of end users in the design process may reduce this abandonment rate (Waller et al., 2005b).

Structure of this standard

Clause \Box outlines the problems that can occur with Severe Speech and Physical Impairments.

Clause 0 gives an overview of the technology that are used to assist people with SSPI.

Clause 0 discusses the problems that are noted when doing HCD with users with SSPI.

Clause 0 gives guidance on how to deal with these problems.

Clause \Box highlights methods known to work at each stage of ISO 13407.

Problems for people with SSPI

People with SSPI may have cognitive impairment which can range from mild to severe and may have difficulty in understanding all that is being said in the conversation and the developer's questions.

People who use an AAC device for communication can produce words up to 25 times slower than those with normal verbal speech (Higginbotham et al., 2008b). The time therefore for a participant to reply, for example "Yes, that has happened to me" could take up to 2 minutes for a participant relying on AAC (Augmentative and Alternative Communication at the University of Washington, 2009). A person with dsyarthric speech may be able to produce words at the same rate as a participant without a SSPI, however the need for clarification of the response can slow down the rate at which the response is understood by the group (Hustad, 2006a).

People with SSPI are likely to become tired more quickly than the general population (Jahnsen et al., 2003a). This is attributed to "post-impairment syndrome", which means that those with SSPI will use between 3 and 5 times as much energy as those without the condition (Wood et al., 2008). Fatigue is also one of the most common and debilitating symptoms associated with Traumatic Brain Injury.

Technology to assist people with SSPI

Augmentative and Alternative Communication (AAC) is the general term for the methods used to aid communication by those for whom the usual form of communication through speech is not sufficient. AAC can be divided into the two different categories of unaided and aided. Most people who use AAC will use a combination of both (American Speech-Language-Hearing Association, 2002).

In unaided AAC systems additional equipment is not used, for example sign language, Makaton (Communication Matters, 2008) and facial expressions (Communication Matters, 2003). These systems have the benefit that they can be used anywhere and the user does not need to carry additional equipment around. However unaided AAC has the disadvantage that the communication partner needs to understand how the system works.

Aided AAC systems refer to a communication methods that involve a device – this may be electronic or paper based (e.g. a word board) – of some form which is external to the user. The device can be used to transmit or to receive messages and generally display symbols which the user selects to convey messages to listeners, for example a symbol board (Beukelman and Mirenda, 1992). A user can select a symbol, picture or in the case of literate users a selection of letters to build up a message. The main advantage of an aided device is its ability to offer potential for communication to people with SSPI for whom unaided systems require refined motor skills or cognitive load not within the user's capability (Hampson, 2006). Electronic devices also offer the potential for synthesised speech output in the case of speech generating devices.

Other assistive technology for this group may include electric scooters or wheelchairs to assist with mobility, personal organisers or reminders to assist with cognitive difficulties, screen readers, magnifiers and hearing aids to help with sensory problems and environmental controls to help them to live as independently as possible.

Problems that can occur when doing HCD with users with SSPI

Can a group of 'representative users' be gathered?

gathering a representative sample of different disabilities and needs is more challenging

How can different accessibility needs be met?

Can informed consent be gathered from participants, and if not is it possible to gain assent from the participant and informed consent from a legal guardian.

It is important to ensure that participants are aware of what they are consenting to

Some participants may have a guardian appointed who is required legally to consent to research.

How will you encourage participants to give feedback?

Participants are often uncomfortable in giving negative feedback to those who have designed the software

How will feedback be recorded?

Usual methods such as notes and audio recordings may not be suitable

Audio recordings may miss information if communication is not through a voice output machine

It is difficult to take notes if participants are communicating via eyegaze and word board

Is it necessary to provide incentives to participants?

Payments can interfere with benefit rules

What measures will be put in place after the study to allow participants to continue with new skills if they wish to?

Participants may have developed new skills during their involvement with the research and may be keen to continue with this after the project ends.

While organising future support is not the responsibility of researcher or developer, there is an ethical obligation to ensure that participants do not feel abandoned at the end of the study.

How will you report back to participants on what you used their feedback for?

The traditional methods for reporting back to participants is by writing a letter to participants thanking them for their help and explaining what stage the project is now at. This may not be suitable for users with SSPI.

How will contact be made with participants?

Recruiting end users can be a challenge in any situation, unlike traditional development advertising in local press or through local universities can be inappropriate or unlikely to reach the desired participants.

Guidance on issues for HCD and users with SSPI

Can a group of 'representative users' be gathered?

It is not difficult to gather a spectrum of different ages and a mix of genders among participants but gathering a representative sample of different disabilities and needs is more challenging.

It may be possible to use purposive or judgemental sampling to find users who cover most of main disabilities (communication impairment, mobility impairment, cognitive impairment and sensory impairment).

Developers can encourage users to consider how others whom they know with different disabilities would interact with the system.

Participants can be asked to consider talking to others about the system or to even ask them if they wish to take part in later evaluations.

By allowing different groups of participants to do later evaluations it is possible to gain as wide a range as possible.

How can different accessibility needs be met?

Conflicting requirements are already part of the user centred design process, in this situation however discussions between users and designers is unlikely to help resolve the issue.

The designers need to carefully examine the possibilities for including as many accessibility and adaptations as possible.

It may be possible to switch accessibility options on and off

Careful planning during development can often allow different adaptations to be used through the same piece of code.

Can informed consent be gathered from participants, and if not is it possible to gain assent from the participant and informed consent from a legal guardian?

By talking to participants after going through Participant Information Sheet and asking yes/no questions we can check that they understand what they are agreeing to.

It is still important to gain assent from the participants in the instance of a guardian signing. This should include questions to ensure they want to take part and know what will be happening in the process.

How will you encourage participants to give feedback?

Participants need to be encouraged to give negative feedback.

It is helpful to remind participants that you really want them to find flaws in the technology.

How will feedback be recorded?

Video can be intrusive, but careful planning can minimise this

Is it necessary to provide incentives to participants?

Methods of incentives can be through gift tokens etc, which should not interfere with benefit rules.

Participants will often volunteer to work without payment due to the enjoyment of taking part.

What measures will be put in place after the study to allow participants to continue with new skills if they wish to?

If it is possible for participants to take part in other projects then putting them in contact with the person organising these should be discussed

Researchers should discuss with the participants and their support staff what parts of the project they enjoyed most

How will you report back to participants on what you used their feedback for?

With participants with SSPI it is beneficial for them to meet with the development team and be given personal feedback on how the project has developed and how their own contributions have been used.

How will contact be made with participants?

Charities who work with people with SSPI may be willing to make the initial contact on behalf of the development team and then if participants are in agreement to put the two parties in contact.

There may be additional ethics procedures to go through with a charity and time should be allowed for this.

Example methods for stages in HCD

General Points

At the beginning of each session review the previous session, this will help any participants who were not present to be caught up on what has happened and will help participants with cognitive problems to recall what took place.

It is important to ensure that participants don't become overtired during the sessions, and to allow for extra time to facilitate breaks. Designers should remember that participants with SSPI are more likely to have illnesses than the general population and there may be times when sessions have to be cancelled or altered at short notice due to illness.

While it is important not to talk in an overly complicated or technical way with participants it is also vital that participants are not spoken to in a condescending tone and are made to feel as equals with the designer.

Understand and specify the context of use

Make contact with potential users and/or organisations who work with people with SSPI. It is important that participants become comfortable with the researcher/developer prior to the work commencing. The staff or family who spend time with participants should also meet with the researcher/developer.

Examine the ethical situation carefully, in addition to any internal ethics procedures, there may be ethics applications required by the charities or organisations involved. The local health authority may also require an ethics application depending where the sessions with the participants will be conducted and how participants will be recruited.

Spend time in the environment and understand how the technology may be used.

Requirements Gathering

Requirements for how technology could be used

Adults with SSPI may at first be hesitant in giving requirements to researchers due to previous poor experience of having people in authority/people from outside of their immediate circle listening to them. As such a group discussion for requirements may be beneficial – particularly if it can be arranged for the participants to know one another before hand. During group discussions it is important for people to have time to form answers using their communication device and not to feel rushed into answering. Participants will benefit from being given feedback on what was discussed at the end of the session in addition to the traditional summary which is sent out a later date. By feeding back immediately on what the researchers have heard, participants will know that their opinions have been listened to and taken onboard.

Requirements on how technology should operate

Participants may have trouble imagining how technology will work or trouble keeping track of what part of the process in using the technology they are currently imagining. Providing storyboards or videos on problems that currently exist and/or how the technology could possibly be used may help alleviate this. It is important not to specify exactly what the system will do, but allows participants to visualise and imagine more easily. The research/designer can show on the screen an image of the position of the process that is currently being discussed. It may be helpful to prompt for consideration of other groups.

Produce Design Solutions

Lo-fidelity prototyping can involve producing sketches of how technology will look and/or in user centred design having the participants produce sketches for how they imagine the system looking. This is not usually a suitable method of prototyping with adults with SSPI. One technique found to be successful is to create more tangible prototypes. Examples of this include making models of the device onto which components can be attached on as the participants wish, or by creating screen designs and then cutting up the buttons etc. These components can then be attached to magnets, the participants can then position the components on a magnetic board in the position they wish. It may also be beneficial to allow participants to use existing accessibility tools and simulate how they work in this situation. It is important that when showing prototypes designs are kept flexible enough to allow instant alterations.